ECONOMIC GOVERNANCE, FOREIGN DIRECT INVESTMENT AND ECONOMIC GROWTH IN SOUTH AND EAST ASIA PACIFIC REGION: EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALYSIS

SRIDEVI YERRABATI

A thesis submitted in partial fulfilment of the requirements of the University of Greenwich for the Degree of Doctor of Philosophy

September 2014

DECLARATION

I certify that this work has not been accepted in substance for any degree, and is not concurrently being submitted for any degree other than that of Doctor of Philosophy being studied at the University of Greenwich. I also declare that this work is the result of my own investigations except where otherwise identified by references and that I have not plagiarised the work of others.

Sridevi Yerrabati Date Dr. Denise Hawkes Date

ACKNOWLEDGMENTS

"Karmanye Vadhikaraste Ma Phaleshu Kadachana, Ma Karma Phala Hetur Bhurma Tey Sangostva Akarmani"

- Shrimad Bhagvad Gita

[Meaning: Do your duty and be detached from its outcome, do not be driven by the end product, enjoy the process of getting there.]

I have been extremely fortunate in my life to have a lovely and adorable Grandmother, Mother and Sister who have shown me unconditional love and support throughout my life. They have played a very important role in the development of my identity and shaping the individual that I am today. I cannot find words to express my thanks to those without whose love, blessings, understanding, inspiration, continuous support and encouragement I would have never thought of stepping into the research world. I would like to dedicate this thesis to them. This thesis would not also have been possible without the support and encouragement of my well-wisher and dearest friend Nishad Narayana Pillai. I would like to extend my deep gratitude and heartfelt thanks for his moral support and inspiration during the period of study.

During the process of writing this thesis, I have accumulated enormous intellectual and material debts to many people. Two people at Business School, University of Greenwich have played a very important role in ensuring continuity and completion of this thesis. First of all, I am highly indebted to my first supervisor and EdD programme leader at Doctoral School, Institute of Education, University of London Dr. Denise Hawkes, for her initiation, inspiration, support, sincere encouragement and constructive comments and suggestions which have helped me to overcome many struggles during my research and writing of this thesis. Without her generous understanding, support, assistance and patience I would never have completed this thesis.

Second, Prof. Stephen Thomas, Director of Research, Business School, University of Greenwich has played a very important role in encouraging my study and giving me confidence in completing it. Dr. Denise Hawkes and Prof. Stephen Thomas were always there when I needed help. I have never met such simple, understandable and down to earth people who were always approachable and understanding; they left a remarkable impact on me. My second supervisor Prof. Ozlem Onaran, has provided constructive comments and suggestions which were particularly useful in providing new direction to my thesis. A special thanks goes to her for the time and patience she has shown with my work and for giving valuable suggestions.

I am also thankful to Prof. Mehmet Ugur whose constructive comments and suggestions have provided valuable insights into developing my thesis further. Through his expertise in metaanalysis and institutional economics, Mehmet has helped in unearthing and resolving the research related issues I have faced. I particularly thank him for extra support, time and energy he has given me during the process of my research. I would also like to thank my other supervisor Dr. Larry Su for the time he has taken in attending my presentation and raising thought provoking questions.

My gratitude extends to the research support staff in Business School research office, Mrs. Gillian Haxell for her sincere help and support. Gill has been very lovely and kind in helping me out with all administrative related tasks. I would also like to thank my fellow research students for their moral support and inspiration over the duration of my studies. Dr. S.R.R.Rao (from University of West of England) and Dr. Bhaskar Patel (from Kavikulguru Institute of Technology and Science) have initiated the idea and encouraged me to apply for a research degree in the UK. I would like to express my deep gratitude to both of them for this.

Earlier versions of the three main chapters of this thesis (chapter 2, 3 and 4) have been presented at University of Greenwich Business School research students seminars, University of Greenwich poster conference 2013 & 2014, MAER Net Colloquium 2013 & 2014; Cambridge Business & Economics Conference 2014 and Association for Heterodox Economics conference 2014. This thesis has greatly benefitted from extremely useful and detailed comments and suggestions received during all these presentations.

In the event that I have failed to refer to some important studies that I have used in this thesis, I hope authors will accept my apologies. My list of acknowledgements would be incomplete without extending my thanks to Business School, University of Greenwich for funding this study under Business School Scholarship Scheme 2011 and supporting my research during the three year period of study.

ABSTRACT

Good economic governance is considered to be one of the key drivers of both inward FDI and economic growth. In spite of this wide belief, empirical estimates focusing on South and East Asia Pacific countries are less than conclusive. The aim of this thesis is to summarise the empirical findings of existing studies on the effect of governance on FDI, FDI on growth and governance on growth for South and East Asia & Pacific regions using systematic literature review and meta-regression analysis. Findings of first meta-regression analysis based on 771 estimates from 48 empirical studies suggest that, except for corruption all measures of governance have an important effect on FDI. While on one hand political stability, government effectiveness and regulation are positively related to FDI, on the other hand rule of law is negatively related to FDI. As expected, aggregate governance has positive effect FDI.

Results of second meta-regression analysis applied to 633 estimates from 37 empirical studies indicate that FDI shows growth enhancing effect in the region as a whole. While FDI showed growth enhancing effects in the case of all estimates, estimates controlling for endogeneity and South East Asia, I did not have sufficient observations in the case of South Asia and East Asia to reach firm conclusions. The findings of third meta-regression analysis using 554 estimates from 29 studies suggest that except for corruption, other measures of governance such as law and aggregate governance have positive effect on growth. Surprisingly, in case of voice and accountability, research literature has failed to provide evidence of genuine effect of it on growth. In addition to the above, this thesis highlights that effect size and statistical significance of the reported estimates depends on study, real world, author and journal related aspects. The results of these three studies have important policy implications.

TABLE OF CONTENTS

CHAPTER 1 INTRODUCTION	1
1.1 INTRODUCTION	2
1.2 MOTIVATION BEHIND THE THESIS	3
1.3 METHODOLOGY	4
1.4 SCOPE OF THE THESIS	4
1.5 STRUCTURE OF THE THESIS	5
CHAPTER 2 ECONOMIC GOVERNNACE AND INWARD	7
FOREIGN DIRECT INVESTMENT IN SOUTH AND EAST ASIA & PACIFIC REG	ION:
EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALY	/SIS
2.1 INTRODUCTION	8
2.2 MEASURES OF GOVERNANCE	8
2.3 METHODOLOGY	10
2.4 LITERATURE REVIEW	13
2.4.1 THEORETICAL VIEWS ON GOVERNANCE AND FDI RELATIONSHI	IP13
2.4.2 EMPIRICAL VIEWS ON GOVERNANCE AND FDI RELATIONSHIP	14
2.5 DISCUSSION OF RESULTS	17
2.5.1 FUNNEL PLOTS	18
2.5.2 CHRONOLOGICAL ORDER OF ESTIMATES	20
2.5.3 SIMPLE META-REGRESSION ANALYSIS	22
2.5.4 MULTIPLE META-REGRESSION ANALYSIS	23
2.5.5 MODERATOR VARIABLES ANALYSIS	28
2.6 CONCLSUIONS	37
CHAPTER 3 INWARD FOREIGN DIRECT INVESTMENT AND	41
ECONOMIC GROWTH IN SOUTH AND EAST ASIA & PACIFIC REGION:	
EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALY	SIS

3.1 INTRODUCTION	42
3.2 INWARD FDI AND ECONOMIC GROWTH IN	
SOUTH AND EAST ASIA & PACIFIC COUNTRIES	42
3.3 METHODOLOGY	45
3.4 LITERATURE REVIEW	47
3.4.1 THEORETICAL VIEWS ON FDI-GROWTH NEXUS	47
3.5 DISCUSSION OF RESULTS	49
3.5.1 FUNNEL PLOTS	50
3.5.2 CHRONOOGICAL ORDER OF ESTIMATES	51
3.5.3 SIMPLE META-REGRESSION ANALYSIS	52
3.5.4 MULTIPLE META-REGRESSION ANALYSIS	53
3.5.5 MODERATOR VARIABLES ANALYSIS	56
3.6 CONCLUSIONS	61
CHAPTER 4 ECONOMIC GOVERNANCE AND ECONOMIC GROWTH	
IN SOUTH AND EAST ASIA & PACIFIC REGION: EVIDENCE FROM	
SYSTEMATIC LITERATURE REVIEWS AND META-ANALYSIS	63
4.1 INTRODUCTION	64
4.2 METHODOLOGY	65
4.3 LITERATURE REVIEW	67
4.3.1 THEORETICAL VIEWS ON GOVERNANCE AND	
GROWTH RELATIONSHIP	67
4.3.2 EMPIRICAL VIEWS ON GOVERNANCE AND	
GROWTH RELATIONSHIP	68
4.4 DISCUSSION OF RESULTS	68
4.4.1 FUNNEL PLOTS	69
4.4.2 CHRONOLOGICAL ORDER OF ESTIMATES	71

4.4.3 SIMPLE META-REGRESSION ANALYSIS	72
4.4.4 MULTIPLE META-REGRESSION ANALYSIS	73
4.4.5 MODERATOR VARIABLES ANALYSIS	77
4.5 CONCLUSIONS	81
CHAPTER 5 CONCLUSION CHAPTER	83
5.1 INTRODUCTION	84
5.2 FINDINGS	84
5.3 CONTRIBUTION TO THEORY	86
5.4 LIMITATIONS	87
5.5 RESEARCH IMPLICATIONS	88
5.6 POLICY IMPLICATIONS	89
REFERENCES	91
APPENDIX	145

LIST OF TABLES

TABLE 2.1: SIMPLE META-REGRESSION ANALYSIS RESULTS

TABLE 2.2: MULTIPLE META-REGRESSION RESULTS

TABLE 2.3: EFFECT OF MODERATOR VARIABLES

TABLE 3.1: SIMPLE META-REGRESSION RESULTS

TABLE 3.2: MULTIPLE META-REGRESSION ANALYSIS RESULTS

TABLE 3.3: EFFECT OF MODERATOR VARIABLES

TABLE 4.1: SIMPLE META REGRESSION RESULTS

TABLE 4.2 MULTIPLE META-REGRESSION RESULTS

TABLE 4.3: EFFECT OF MODERATOR VARIABLES

LIST OF FIGURES

FIGURE 2.1: SUMMARY OF METHODOLOGY

FIGURE 2.2: FUNNEL PLOTS FOR MEASURES OF GOVERNANCE AND FDI ESTIMATES

FIGURE 2.3: CHRONOLOGICAL ORDER OF MEASURES OF GOVERNANCE AND FDI ESTIMATES

FIGURE 3.1: FOREIGN DIRECT INVESTMENT AND GROSS DOMESTIC PRODUCT PER CAPITA IN SOUTH AND EAST ASIA & PACIFIC REGION

FIGURE 3.2: SUMMARY OF METHODOLOGY

FIGURE 3.3: FUNNEL PLOT FOR FDI GROWTH ESTIMATES

FIGURE 3.4: CHRONOLOGICAL ORDER OF ESTIMATES BASED ON AVERAGE YEAR OF SAMPLE PERIOD

FIGURE 4.1: SUMMARY OF METHODOLOGY

FIGURE 4.2: FUNNEL PLOTS FOR MEASURES OF GOVERNANCE AND ECONOMIC GROWTH ESTIMATES

FIGURE 4.3: CHRONOLOGICAL ORDER OF ESTIMATES BASED ON AVERAGE YEAR OF SAMPLE PERIOD

LIST OF ABBREVIATIONS

EM	EMPIRICAL STUDY
FDI	FOREIGN DIRECT INVESTMENT
IV TECHNIQUES	INSTRUMENTAL VARIABLE TECHNIQUES
MMRA	MULTIPLE META-REGRESSION ANALYSIS
OLS	ORDINARY LEAST SQAURES METHOD
PIOS	POPULATION, INDEPENDENT VARIABLE,
	OUTCOME VARIABLE AND STUDY DESIGN
SMRA	SIMPLE META-REGRESSION ANALYSIS
Т	THEORETICAL STUDY
TEM	THEORETICAL AND EMPIRICAL STUDY

CHAPTER 1

INTRODUCTION

1.1 INTRODUCTION

Until 1980, most Asian countries viewed inward foreign direct investment (hereafter FDI) with caution. The presence of multinational companies was viewed as impinging on the national sovereignty. However, this view has changed in the past three decades as most of these countries have welcomed foreign investments. While improvements in globally integrated production and marketing systems, and an increase in the number of bilateral investment treaties can be seen as an important reason for this, improvements in developing countries' physical, human and institutional factors have also contributed to it (Brooks et al., 2003).

While South and East Asia & Pacific countries had long pursued the traditional strategy of selfreliance, FDI has become topical in South and East Asia region since the late 1980's when most of the countries in the region adopted an open door policy to welcome FDI (figure 3.1) (for example, India in 1981 and China's open door policy in 1978) (Wang, 1995). This change is seen as a result of major political decision and economic development strategy so as to uplift the economies from their economic backwardness and reach their long term goals of development.

In addition to opening doors for foreign investments, the need was felt for appropriate policies and for an institutional environment for economic growth. Countries within this region have suffered with poor quality of governance. Poor governance has not only affected the economic performance of these nations to some extent but have also acted as constraint in its speedy development. However, over a period of time, most of the countries in this region have all, to varying degrees, made changes to their governance to make investment environment conducive and to sustain future economic growth (Haggard, 2004; Lee, 2002). Needless to say, economic growth was needed to make any sustained and meaningful reduction in poverty, in reducing unemployment and improving living standards of people.

In recent times inward FDI into developing Asia has surged tremendously mainly with the liberalisation of investment policies and lowering of capital controls (ABD, 2007). Inward FDI has played a very important role in many regions of South and East Asia & Pacific countries development. While these countries have welcomed varying degrees of inward FDI into these regions, their effect on economic growth has been different based on the investment policies they have adopted. Some light is shed on FDI trends in this region from 1980 – 2012 in figure 3.1.

Figure 3.1 shows that there have been fluctuating trends in FDI into South and East Asia & Pacific countries during this period which can broadly be due to the kind of investment policies followed by these countries and also as a result of external factors such as currency appreciation. While most of the East Asian countries have initially restricted FDI into their countries to promote and

protect domestic companies, others such as such as Malaysia, Thailand and Indonesia had different policies for different industries. In Thomson's (1999) view, while investment was completely restricted in certain strategic industries, it was limited in others.

In addition to above, external factors such as currency appreciation of Yen around the 1980's has made it expensive to manufacture labour intensive goods. As a result, Japan started looking for other countries in Asia where labour costs were cheap. Yen appreciation has also created a wealth effect which led to an increase in inward investments to East Asian countries such as South Korea and Taiwan and later to China (Willem and Salike, 2013)

On the other hand, investment policies have been restrictive in South Asia until the 1990's when most of the countries in this region has opened up their doors and made it conducive for foreign investors (Sahoo, 2006). Most of the countries have also used tax incentive policies in order to attract FDI to promote employment opportunities, develop rural areas and the development of specific industries. Overall, inward FDI was regulated differently with differing degrees of efficiency by countries in this region.

Nevertheless, if the decisions made to welcome FDI and to create a conducive environment for such investments was right or not remained a big question. Even after nearly three decades of taking such decisions, there still remains a debate on this issue. Hence, the purpose of this thesis is to synthesise the research results in governance, investments and growth in the case of South and East Asia & Pacific economies using empirical studies published from 1980 - 2012. Such synthesis is aimed at aiding policy makers in making evidence based decisions in the area of governance, growth and investments.

1.2 MOTIVATION BEHIND THE THESIS

Two factors have contributed to the emergence of this thesis. Firstly, the infleuntial study of North (1990) and Dunnings OLI framework (1980) which has highlighted the growing importance of the role played by governance on investments and economic growth in general. According to Dunning (1980) an MNC will enter a host country when each of the ownership, location and organisation factors are met. In this context, economic governance can be seen as a location factor which might deter investments or serve as a helping hand for foreign investors depending on the form of investment and the industry into which these investments flow (Dunning, 1980). North (1990) in his institutional theory posits that institutions in the form of political, economic and structural interactions are human-made constraints which aim to decrease the level of uncertainty and allow for firms and individuals to interact efficiently. However, when these institutions function inefficiently it increases the transaction costs and hence discourages FDI inflows and

economic activity. In this context economic governance can be seen as an institutional factor which can either help in attracting or deterring FDI (Dahlstrom and Johnson, 2007).

Empirically the effect of economic governance on FDI and economic growth is widely debated in the case of South and East Asia & Pacific countries leaving a scope for aggregating these studies to explore the genuine and overall impact of governance on both inward FDI and economic growth. It is worth noting that this thesis only focuses on formal institutions or economic governance and does not include informal institutions such as religion, customs and values among many others.

Secondly, there is a growing surge in the inward FDI and economic growth in the case of South and East Asia & Pacific countries since 1980's. For instance FDI into South Asia increased from \$4,814m to \$3,06,660m, FDI into East Asia increased from \$1,85,173m to \$2,49,2960m (UNCTAD, 2013). However, empirical evidence on whether or not inward FDI has contributed to economic growth is debatable. This leaves scope to summarise empirical studies on the aggregate effect of inward FDI on economic growth.

1.3 METHODOLOGY

The review methodology used in this thesis i.e the methods used for searching studies, study selection, critical evaluation and data extraction is informed by three sources. Firstly, Cambell and Cochrane Collaboration guidelines on systematic reviews in healthcare and social policy; secondly, Centre for Reviews and Dissemination (CRD, 2009) of the University of York; thirdly, Evidence for Policy and Practice Information and Co-ordinating Centre (EPPI-Centre) of the Institute of Education. Data analysis is informed by Doucouliagos et al., (2010), Doucouliagos and Ulubasoglu (2008) and Stanley and Doucouliagos (2012). Reporting guidelines are informed by Stanley et al., (2013). I have also received extensive knowledge and comments on analysis at MAER-Net Colloquium 2013, Cambridge Business & Economics Conference 2014 and 16th Conference of the Association for Heterodox Economics. All search results of the three studies are uploaded to Endnote provided by the University of Greenwich.

1.4 SCOPE OF THE THESIS

This study formulated a specific eligibility criterion for each included study based on population, independent variable, outcome variable and study design to determine the kinds of study that should be included or excluded in the systematic review and meta-analysis. This is done to allow for other researchers to either replicate or extend this study in future. While the scope of this study is limited by the set criteria, these are detailed here. Firstly, geographically the scope of each study is limited to South and East Asia and Pacific countries as defined by World Bank and South Korea.

Secondly, I restricted this study to include both published and unpublished empirical studies from 01.01.1980 to 31.12.2012 written in the English language only. This leaves scope for extending this study in future by including studies published in other languages. Thirdly, the scope of the study is also limited by University of Greenwich subscriptions to various journals in Business and Economics. However, I have mitigated this limitation by hand searching and using the Google search engine. Last but not least, this thesis only looks at formal governance and includes only those studies that have defined it as an index based on a scale and has excluded the ones defining it in terms of raw numbers such as number of assassinations or number of punishments.

1.5 STRUCTURE OF THE THESIS

In order to achieve the research aims, this thesis will contain three meta-regression analysis chapters in addition to an introductory chapter and a concluding chapter. The first and the third meta-regression chapters will address the overall impact of measures of governance on inward FDI and economic growth respectively. The second meta-regression chapter will focus on the impact of inward FDI on economic growth. In addition to this, each chapter has focused on the factors causing heterogeneity in the reported results drawing from a range of variables both paper and journal specific.

The first meta-regression chapter is entitled 'Economic governnace and Inward Foreign direct investment in South and East Asia & Pacific region: Evidence from Systematic Literature Reviews and Meta-Analysis'. The aim of this chapter is to address the overall impact of economic governance on inward FDI. Results of meta-regression analysis based on 771 estimates from 48 empirical studies demonstrate that governance indeed matters for attracting foreign investments and that governance in the form of voice and accountability, political stability, regulation, government effectiveness along with aggregate governance will be major determinants of inward FDI for South and East Asia & Pacific countries.

The second meta-regression chapter is entitled 'Inward Foreign direct investment and Economic growth in South and East Asia & Pacific region: Evidence from Systematic Literature Reviews and Meta-Analysis'. This chapter investigates the role played by inward FDI on economic growth. Meta-regression analysis on 633 estimates from 37 empirical studies shows that inward FDI has a positive effect on economic growth in the case of models with all estimates. This is also true in case of estimates controlling for endogeneity. However, I show that the effect differs across regions.

The third meta-regression chapter is entitled 'Economic governance and economic growth in South and East Asia & Pacific region: Evidence from Systematic Literature Reviews and MetaAnalysis'. Using 554 estimates from 29 empirical studies, this chapter tests the genuine overall effect of governance on growth. The results of this chapter demonstrate the importance of governance for economic growth. In particular I find that voice and accountability, regulation, law and aggregate governnace have growth enahncing effects. While government effectiveness has negative effect, corruption and political stability are not important for growth.

The concluding chapter of this dissertation summarises the major findings as well as points out the main policy implications of the overall thesis. In addition to this, it also identifies the limitations of the overall thesis and addresses some possible areas of furture work.

CHAPTER 2

ECONOMIC GOVERNNACE AND INWARD FOREIGN DIRECT INVESTMENT IN SOUTH AND EAST ASIA & PACIFIC REGION: EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALYSIS.

PRESENTED AT MAER-NET COLLOQUIUM 2013.

SUBMITTED TO JOURNAL OF ECONOMIC SURVEYS.

2.1 INTRODUCTION

Given the important role inward FDI can play in accelerating economic growth and transformation, developing countries are interested in attracting it. Amongst many other benefits such as creating employment and increasing technological development, inward FDI provides a more stable source of external financing than sources such as private debt and portfolio flows (Gastanaga et al., (1998); Globerman and Shapiro (2002a); Gani (2007)). Hence, countries in South and East Asia & Pacific region have liberalised their FDI regime and have pursued policies to attract FDI. They have also addressed various governance related issues to maximise such attraction. However, whether governance in these countries has achieved the purpose or not remains debatable.

Hence, the aim of this study is to contribute to evidence based policy making and to academic research on governance FDI relationship by providing meta synthesis of empirical evidence on various measures of governance and FDI, identifying factors causing heterogeneity in results, pointing to policy implications of the results and identifying potential avenues for future research within this field of study. In order to achieve the research aim, I raised the following questions: Is there a genuine effect of measures of governance on inward FDI? What is the directionality of such effect? I answer these questions by using all available empirical evidence obtained using systematic literature review from 1980 – 2012 on effects of governance on inward FDI.

The rest of the paper is organised as follows. Section 2.2 defines and outlines various measures of governance followed by section 2.3 which outlines the methodology used in this study. Section 2.4, presents a brief overview of literature on measures of governance and FDI from 1980 - 2012. Section 2.5 discusses results followed by conclusions.

2.2 MEASURES OF GOVERNANCE

The definition of economic governance has evolved over the last few years. According to Kaufmann et al, (1999a) Governance consists of the traditions and institutions by which authority in a country is exercised. This includes the process by which governments are selected, monitored and replaced; the capacity of the government to effectively formulate and implement sound policies; and the respect of citizens and the state for the institutions that govern economic and social interactions among them. Good, transparent and efficient governance in host countries ensures the safety of investments and thus attracts foreigners to invest. While there are many international and local authorities which give both subjective and objective information on governance, literature in the field of governance and inward FDI has used four main sources. They are worldwide governance indicators provided by Kaufmann et al., 1996) under World Bank project, Freedom House measure of voice and accountability and political rights, Polity dataset

and International Country Risk Guide (ICRG). These governance measures are briefly explained here.

First, a more recent and widely applied governance measure is constructed by Kaufmann et al. (1999). Governance infrastructure measured using Kaufmann et al. (1996 – 2002) describes six aspects of governance such as the rule of law, government effectiveness, regulatory quality, control of corruption, political stability and voice and accountability. These are meta governance estimates which are estimated using 31 different qualitative measures from 13 different sources such as the World Bank, BERI, Heritage Foundation and the World Economic Forum and the Economist Intelligence Unit. These are available for a set of 212 countries from 1996 – 2012. As compared to individual measures of governance, these meta-governance indicators provide more precise measures of governance. Governance score for each country is given on a scale of + 2.5 to - 2.5, from good governance to bad governance respectively.

Secondly, Freedom House measures voice and accountability and political rights aspects of governance. Scores given to each country are based on their level of political rights and civil liberties. Most democratic countries are assigned 1 and least democratic countries are assigned 7. Thirdly, similar to Freedom House, Polity dataset provides measures on the level of voice and accountability in a country. It measures the nature of regime in a country on a 20 point scale from -10 to a fully autocratic country to +10 to a fully democratic country.

In the fourth instance, governance quality indicators are also provided by International Country Risk Guide (ICRG) through a monthly publication of Political Risk Services (PRS). Overall governance environment is measured using the following indices: Corruption, Rule of Law, Bureaucratic Quality, Risk of Expropriation and Government Repudiation of Contracts. While Corruption, Rule of Law and Bureaucratic Quality are measured on a scale of 0 to 6, Risk of Expropriation and Government Repudiation a scale of 0 to 10. In both cases, higher values indicate better quality of governance and vice versa.

These different datasets on the quality of governance raise the issue of divergence in various measures of governance measured by these institutions. In order to synthesise governance – growth effects, I delved deeper into the sub measures of each measure of governance to synthesise them based on the common sub measures. After observing the individual variables (representative sources) that have been used in measuring governance by these different data sources, I have classified governance into 7 measures based on World Wide governance measures. These seven measures are termed hereafter as voice and accountability, political stability, government effectiveness, regulation, law, corruption and aggregate governance.

2.3 METHODOLOGY

For information on sources of methodology please refer to section 1.3. I started by establishing a pre-established search criteria to identify all studies in the English language on measures of dependent variable (FDI) and independent variable (governance). This was done in two stages: the first stage involved identifying databases for published and unpublished studies. The second stage involved specifying key words, searching databases and storing results.

For published studies, databases such as EBSCO host (Business and Economics database), Web of Knowledge (Social Sciences), International Bibliography of the social sciences (Economics, Politics, Sociology, Anthropology and Economics), Science Direct (Science and Humanities), Swetswise and JSTOR (Social Sciences) were used. For unpublished studies, databases such as World Bank e-library, Harvard Kennedy e-library, Asian Development Bank e-library, National Bureau of Economic Research and IMF e-library were used. In addition to these databases, two search engines namely Google Scholar and Web of Knowledge provided by University of Greenwich were utilised. In addition to the above, manual search was performed in order to identify grey literature using two approaches – snowball approach and random search of studies in 5 journals. Under the snowball approach I have started with the reference list of studies identified through systematic review and proceeded to find new studies. These exhaustive searches were carried out to identify all possible studies on measures of governance and inward FDI.

FIGURE 2.1: SUMMARY OF METHODOLOGY



With a pre-defined list of key words for measures of governance and inward FDI (appendix 2.1), 'title', 'abstract', 'text' and 'keyword' were searched in the above databases. The time period of the study was January 1980 – December 2012. A total number of 4996 studies were retrieved which have analysed the relationship between measures of governance and inward FDI. From this, 150 and 109 duplicate studies were removed using automatic and manual duplicate searches respectively. This left a total of 4728 unique studies for further screening. Figure 2.3.2.1 summarises the methodology used in this chapter.

The relevance of each study was checked based on whether the study estimates or analyses the relationship between measures of governance and inward FDI? While the earlier study is coded as 'E', later ones are coded are 'T'. If a study estimates and analyses the relationship then it is coded as 'TE'. Studies which do not satisfy any of these criteria are not included in meta-analysis. 131 studies were selected from the initial screening stage and these were considered for the critical evaluation stage. This was done using PIOS (Population-Independent variable-Outcome variable-Study design) criteria (appendix 2.2). While 94, 62, 68 and 94 studies have satisfied population, independent variable, outcome variable and study design respectively, only 40 studies have satisfied all four criteria (appendix 2.3, 2.4). Another 8 studies were added to this number by hand searching, making a total of 48^1 studies for meta-analysis (appendix 2.5 for composition of published and unpublished studies). My exclusive search for studies on South Korea did not result in any records.

The following data were obtained from 48 studies. Firstly, bibliographical information such as name of the first author and University, year of publication of study and type of study (whether it is a published or unpublished study). Secondly, study characteristics such as kind of data used, information on dependent and independent variables such as their functional form and their data sources, and estimation methods. Thirdly, outcome related information such as estimated parameters, t values, standard errors, P value, Z value, F value for linear, non linear and squared terms was obtained.

The general form of econometric model used in the primary empirical studies with linear terms only (equation 1.1) and that with linear, non-linear and squared terms (equation 1.2) is shown below:

$$\begin{split} &Y_{it} = \alpha_0 + \alpha_1 X_{it} + \gamma F_{it} + \epsilon_{it} & \text{equation (1.1)} \\ &Y_{it} = \alpha_0 + \alpha_1 X_{it} + \alpha_2 X_{it} \cdot K_{it} + \alpha_3 X^2_{it} + \gamma F_{it} + \epsilon_{it} & \text{equation (1.2)} \\ &\text{In above equations,} \end{split}$$

¹ After including South Korea in the list of sample countries, I have repeated the above procedure to identify studies focusing on South Korea. This search did not retrieve any study that satisfied all four PIOS criteria.

Y – Inward FDI

X - Measures of governance,

- F- Vector of other variables
- i Country indices
- t-Time indices
- α_0 Constant term
- α_1 Marginal effect of governance on Y
- $X \cdot K$ Interaction term of measures of governance with K
- X² Non-linear term of measures of governance
- α_2 Measures the effect of X.K on inward FDI conditional on the value of K
- α_3 Measures the effect of X² on Y conditional on its own value
- ϵ Random error term

The effect size is measured using partial correlation to allow for meaningful comparison across different models. Various estimates of α_1 are converted into partial correlations using the formula $r = [t/\sqrt{t^2 + dof})$. Where, t stands for t –statistics of the multiple regression coefficient, dof stands for the degrees of freedom of the respective t –statistic.

Modelling simple and meta-regression analysis

The following equation is used for simple meta-regression analysis for estimating the overall effect after correcting for publication bias²:

 $r_{ij} = \beta_0 + \beta_1 \operatorname{SE}^2_{ij} + \varepsilon_{ij} \operatorname{equation} (1.3)$

The following equation is used for multiple meta-regression analysis for estimating the overall effect after correcting for publication bias:

 $r_{ij} = \beta_0 + \beta_1 SE^2_{ij} + \beta_2 X_{ij} + \epsilon_{ij}$ equation (1.4)

The following equation is used for multiple meta-regression analysis with study and journal specific moderator variables.

 $r_{ij} = \beta_0 + \beta_1 \operatorname{SE}^2_{ij} + \beta_2 X_{ij} + \beta_3 Z_j + \varepsilon_{ij} \operatorname{equation} (1.5)$ i = estimate

² This thesis focuses on estimating overall effect of governance on FDI, FDI on growth and governance on growth respectively after correcting for publication bias. Publication bias is tested using Funnel Asymmetric Test (FAT) and Precision Effect Test (PET). FAT-PET is estimated using equation $t_i = \beta 1 + \beta 0 (1/SE_i) + v_i$ (where FAT is H₀: $\beta 1 = 0$ and PET is H₀: $\beta 0 = 0$). These aspects are explored in a different study.

j = journal

r = partial correlation coefficient

SE = standard error

SE²=squared standard error

 β_0 = shows the effect of independent variable on dependent after correcting for publication bias

 β_1 = coefficient of SE²

 β_2 = coefficient of other factors such as real world

 β_3 = coefficient of study and author related factors

 $\epsilon_i = error term$

X = estimate specific covariates

Z = journal specific covariates

It is worth highlighting at this point that while some studies have defined r on a scale of 0-1 from low to high governance, others have used it as 0-1 high to low governance. In order to aggregate estimates, I have rescaled all estimates as 0-1 low to high governance³. This was done by inversing and multiplying both coefficients and standard errors of estimates defined on the opposite scale (i.e. 0-1 high - low governance) by -1.

2.4 LITERATURE REVIEW

While it is generally believed that good governance in a host country helps in attracting inward FDI, most of the empirical studies show that this is not the case. A systematic literature review of these empirical papers is presented here with a view to unearthing the issues within existing literature in terms of differences in their findings and the reasons causing such differences.

2.4.1 THEORETICAL VIEWS ON GOVERNANCE AND FDI RELATIONSHIP

Two main theoretical frameworks have been used in explaining the relationship between economic governance and inward FDI. Firstly, Dunning's OLI framework (1980) explains various reasons for which an MNC enters into a host country. According to Dunning (1980) an MNC will enter a host country when each of the ownership, location and organisation factors are met. In this context, economic governance can be seen as a location factor which might deter investments or serve as a helping hand for foreign investors depending on the form of investment and the industry into which these investments flow.

³ Low governance means less democracy, low political stability, less regulation, low levels of government effectiveness, less of rule of law, high corruption and low overall governance.

Secondly, North (1990) in his institutional theory posits that institutions in the form of political, economic and structural interactions are human-made constraints which aim to decrease the level of uncertainty and allow for firms and individuals to interact efficiently. While governance aims to facilitate investments, they effect transaction (ex: cost of protecting property rights) and transformation costs (ex: by effecting production interruptions) which in turn effects the profitability of such investments (Dahlstrom and Johnson, 2007). Both Dunning's and North's theories suggest that based on contextual factors, governance can have either positive or negative effects on FDI.

2.4.2 EMPIRICAL VIEWS ON GOVERNANCE AND FDI RELATIONSHIP

Empirical studies on the measures of governance and inward FDI for South and East Asia & Pacific region that have been identified in the search are: Gastanaga et al., (1998), Globerman and Shapiro (2002a), Globerman and Shapiro (2002b), Hsiao and Shen (2003), Anghel (2004), Globerman and Shapiro (2004), Gani (2007), Hur et al., (2007), Adeoye (2009), Brunetti and Weder (1998), Wernick et al., (2009), Ali et al., (2010), He et al., (2011), Muhammad et al. (2011), Jadhav (2012), Luca and Spatafora (2012), Habib and Zurawicki (2001), Wei (2000), Teksoz (2004), Voyer and Beamish (2004), Straub (2005), Dahlstrom and Johnson (2007), Khamfula (2007), Brouthers et al., (2008), Cole et al., (2009), Sadig (2009), Woo and Heo (2009), Qian et al., (2012) and Mathur and Singh (2013), Nigh and Schollhammer (1987), Singh and Jun (1995), Busse and Hefeker (2005), Baek and Qian (2011), Zheng (2011) and Driffield et al., (2012), Sevoum (1996), Lee and Mansfield (1996), Ahn et al., (1998), Li and Resnick (2003), Nunnenkamp and Spatz (2004), Ahlquist (2008), Mayer (2006), Elo (2007), Yackee (2008), Zhang and Fu (2008), Akisik and Pfeiffer (2009), Rai (2009), Azemar and Desbordes (2010), Binici (2010), Goodspeed et al., (2011), Arbatli (2011), Davis (2011) and Gordon et al., (2012), Cyrus et al., (2006), Fan, Morck et al., (2009), Arbatli (2011), Busse et al., (2011) and Wang et al., (2011), Harms and Ursprung (2002), Addision and Heshmati (2003), Jensen (2003), Li and Resnick (2003), Jensen & McGillivray (2005), Busse (2004), Blanton & Blanton (2007), Choi (2008), Guerin and Manzocchi (2009), Doces (2010). All these studies are grouped based of the measure of governance namely, voice and accountability, political stability, government effectiveness, regulation, law, corruption and aggregate governance.

Voice and accountability captures the extent to which citizens in a country have freedom of expression, freedom of association & media and have a voice in the government (Wernick and Haar, 2009). Voice and accountability can affect FDI by inclusion or exclusion of public opinion on investments which can in turn allow or deter foreign investments (Gani, 2007). Studies by Globerman and Shapiro (2002a), Jadhav (2012), Woo and Heo (2009), Busse and Hefeker (2005),

Zheng (2011), Li and Resnick (2003), Davis (2011), Gordon et al., (2012), Harms and Ursprung (2002), Jensen (2003), Jensen & McGillivray (2005), Busse (2004), Blanton & Blanton (2007), Choi (2008), Guerin and Manzocchi (2009) and Doces (2010) have reached mixed conclusions on the role of voice and accountability on inward FDI.

On the one hand, results reported by Globerman and Shapiro (2002a), Busse and Hefeker (2005), Zheng (2011), Harms and Ursprung (2002), Jensen (2003), Jensen & McGillivray (2005), Busse (2004), Blanton & Blanton (2007), Choi (2008) and Doces (2010) show that voice and accountability has a positive and significant effect on FDI. On the other hand Jadhav (2012) and Guerin and Manzocchi (2009) show that voice and accountability has a negative and significant effect on FDI. Others like Woo and Heo (2009), Li and Resnick (2003) and Gordon et al., (2012) report mixed results.

Political stability⁴ measures the solidity of government to political shocks, terrorism and domestic violence which can eventually reduce the risk of doing business and allow investments. Presumably foreign investors would like to invest in countries with political stability to ensure the continuity of policies by government. Studies focusing on this measure of governance are Globerman and Shapiro (2002a), Anghel (2004), Jadhav (2012), Singh and Jun (1995), Busse and Hefeker (2005), Baek and Qian (2011), Gordon et al., (2012), Busse et al., (2011) have generated mixed results. While Anghel (2004), Baek & Qian (2011) and Busse et al., (2011) found positive and significant effect, negative and insignificant effect is shown by Jadhav (2012).

Government effectiveness measures the quality of public services and the insulation of those services from political pressure. Through government effectiveness, government can exert discretionary power on economic activities by designing and implementing economic policies which can either deter or encourage investments (Globerman and Shapiro (2002a), Anghel, (2004)). Studies by Gastanaga et al., (1998), Arbatli (2011), Gordon, Loeb and Zhu (2012) and Jensen (2003) show mixed effects of government effectiveness on FDI under different models.

Regulation as one of the elements of governance indicators is the widest and diverse measure as it includes regulation related to aspects such as intellectual property rights, environment regulations, restrictive capital controls, accounting standards and corporate governance and tax and tariffs. Regulation captures the ability of a government in generating these policies and using them to promote private sector development. Through these policies regulation can affect FDI as they can

⁴ Similar to corruption political stability was considered in two ways – political stability and political instability. For aggregating these studies, political instability was transformed into political stability by inversing and multiplying both coefficient and t value with -1.

either speed up or delay the investments alongside affecting the cost of investments. There have been only three studies that have looked at the impact of this measure on FDI by Globerman and Shapiro (2002a), Jadhav (2012), Gordon et al., (2012) which reported positive and significant, positive and insignificant and mixed effect respectively leaving a scope for both further research and conclusive results.

Law can affect investments through various legal institutions and property rights protection. This measure also includes the quality of contract enforcement, the police, the courts and the likelihood of crime. In a country where there are weak legal institutions and property rights protection, very few foreign investors would like to invest as it would put their investments at risk and vice versa. Positive and significant effect is shown by Anghel (2004), Gani (2007), Jadhav (2012) and Fan et al., (2009). While Globerman and Shapiro (2002a) have shown positive and insignificant effect of rule of law, Arbatli (2011) has shown negative and insignificant effect. Studies by Busse and Hefeker (2005) and Gordon et al., (2012) have reported mixed effects.

Corruption is viewed as one of the important measures of governance as it has an important bearing on investments. Corruption measures the extent to which public goods are misused or used for private purposes by individuals. However, corruption⁵ cannot be considered in isolation from other governance related factors as bad governance is closely associated with corruption. Studies by Gastanaga et al., (1998), Globerman and Shapiro (2002a), Hsiao and Shen (2003), Anghel (2004), Gani (2007), Jadhav (2012), Habib and Zurawicki (2001), Wei (2000), Teksoz (2004), Voyer and Beamish (2004), Straub (2005), Dahlstrom and Johnson (2007), Khamfula (2007), Sadig (2009), Mathur and Singh (2013), Woo and Heo (2009), Goodspeed et al., (2011), Gordon et al., (2012) and Jensen (2003) have focused on the effect of corruption on inward FDI.

Corruption is considered to affect foreign investments in two ways – increase in cost of investments leading to decrease in profitability of such investments and increase in uncertainty levels in host country. Some studies have also shown that corruption 'greases the wheels' of investments rather than 'sands the wheels of investment' (Globerman and Shapiro (2002a), Gastanaga et al., (1998, Hsiao and Shen (2003) and Teksoz (2004)).

Finally, Globerman and Shapiro (2002b), Globerman and Shapiro (2004), Hur et al., (2007), Adeoye (2009), Wernick, Haar and Singh (2009), Ali et al., (2010), Muhammad et al. (2011), Luca and Spatafora (2012), Ahlquist (2008), Goodspeed et al., (2011), Gordon et al., (2012) have

⁵ Empirical studies focusing on corruption have considered the measure in two ways – corruption and control of corruption. For aggregating these studies, control of corruption was transformed into corruption by inversing and multiplying both coefficient and t value with -1.

focused on the effect of aggregate governance on inward FDI. Overall governance includes various political, legal and institutional factors in a country that can have a bearing on investments. While governance is expected to show a positive effect on foreign investments by providing impartial, effective and efficient conditions to operate, there is no conclusive evidence on this.

Mixed results and seemingly contradictory arguments on the empirical relationship between measures of governance and inward FDI can be attributed to various measurement, conceptual and methodological differences in these studies (appendix 2.7). Given this situation, policy makers may be uncertain as to what kind of policy they should propose in order to create a favourable investment climate for foreign investors in terms of economic governance.

In order to address the above inconclusiveness, as outlined in the introduction section this study has the following research aims; firstly, to deal with the effect of measures of governance on inward FDI and secondly with respect to heterogeneity. With regards to the effect, I raise the following two questions: firstly, is there any genuine effect of each measure of governance (voice and accountability, political stability, government effectiveness, regulation, corruption and rule of law) on the inward FDI into South and East Asia & Pacific countries? Secondly, what is the directionality of such effect? With respect to differences in reported results the following questions will be answered. Why do governance-FDI studies report such divergent results? Is the heterogeneity due to the data generating process or is it due to differences in research design? An overall summary of this chapter is given in appendix 2.8.

2.5 DISCUSSION OF RESULTS

I present and analyse results of simple meta-regression analysis (SMRA) and multiple metaregression analysis in this section. Before that, funnel plots and graphs of chronological order of estimates are presented. These graphs are used in order to offer a clear picture of the state of empirical knowledge in governance FDI studies.

2.5.1 FUNNEL PLOTS

FIGURE 2.2: FUNNEL PLOTS FOR MEASURES OF GOVERNANCE AND FDI ESTIMATES













Estimates of measures of governance and inward FDI are plotted on the funnel plot shown in the graphs above. Funnel plot is used to trace the relationship between the effect size which is measured using partial correlation (shown on X axis) and its precision measured as inverse of standard error (shown on Y axis). While high precision estimates are generally few and are compactly distributed at the top of the funnel, low precision estimates are at the bottom of the funnel and are widely distributed. One possible reason for the wide dispersion of estimates (which is the case in most of the graphs) is publication bias⁶ (Doucouliagos and Ulubasoglu, 2008). In each of the above graphs, the centre of the plot represents the estimated true underlying effect of respective measure on growth. In contrast to graphs of political stability, the other graphs show wide dispersion of governance-inward FDI values around the central value.

I have tested for publication bias using Funnel Asymmetric Test (FAT) and Precision Effect Test (PET) (appendix 2.9). Despite the presence of publication bias, PET results suggests that there is genuine effect of each measure of governance on FDI along with aggregate governance. However, they are not robust in case of corruption and aggregate governance.

FIGURE 2.3: CHRONOLOGICAL ORDER OF MEASURES OF GOVERNANCE AND FDI ESTIMATES













The graph above shows the chronological order of estimates of measures of governance on inward FDI. X-axis shows end year of sample period and Y axis shows partial correlation. Chronological ordering of graphs offers an insight into evolution of effect sizes and highlights the trends. With the exception of voice and accountability and political stability graphs, I see a downward trend in the estimates⁷. Downward trend has an important economic interpretation as it indicates that governance over a period of time has a declining effect on inward FDI as opposed to initial years of investment. As an alternative explanation, the downward trend can also be due to the fact that the econometric techniques have got better at controlling econometric problems and therefore smaller estimates are found.

⁷ I see the same downward trend in these graphs taking end year of sample period instead of average year.

2.5.3 SIMPLE META-REGRESSION ANALYSIS

	Political	Government	Regulation	Law	Corruption	Aggregate
	Stability	effectiveness				governance
	(Col. 1)	(Col. 2)	(Col. 3)	(Col. 4)	(Col. 5)	(Col. 6)
Un weighted	0.04	0.08	0.17	0.06	0.01	0.14
estimates, β0	(2.53)	(1.67)	(6.78)	(2.94)	(0.35)	(3.45)
(Row1)	R ² =0.04	R ² =0.01	R ² =0.33	R ² =0.09	$R^2=0.10$	R ² =0.002
Estimates	0.03	0.01	0.18	0.12	0.05	0.05
weighted by	(1.68)	(0.49)	(5.34)	(13.32)	(2.66)	(1.82)
precision, β0	R ² =0.08	R ² =0.01	R ² =0.39	R ² =0.16	R ² =0.07	R ² =0.01
(Row2)						
Number of	154	36	51	42	166	62
estimates						

TABLE 2.1: SIMPLE META-REGRESSION ANALYSIS RESULTS

Note: Values in parenthesis right below the estimate represent t-values. Each column represents models run with all estimates of that measure of governance. Despite of removing the effect of outliers, results for voice and accountability are infeasible and hence these are presented in appendix 2.11.

Table 2.1 shows unweighted and weighted simple meta-regression results of individual measures of governance on inward FDI. As can be noted, all unweighted estimates are with positive sign, indicating that a higher measure of each measure leads to more FDI. For instance, tighter regulations are associated with more FDI. In the case of corruption, results should be read inversely (due to rescaling) i.e. more corruption leads to less FDI. A positive effect of aggregate governance in the last column indicates that better governance is good for FDI.

Except for corruption, all the estimates are significant and unreliable as the R^2 value of each of these measures is very low (R^2 value ranges from 0.002 for aggregate governance to 0.33 for regulation). In addition to lower R^2 values, another shortcoming with this method of estimation is that the unweighted method treats all estimates equally with equal weight. Therefore studies with a large number of estimates can have an undue influence on the statistical assessment. Therefore these results can be biased and misleading. Hence, following Stanley and Doucouliagos (2012), I ran the above models using the weighted least squares method where estimates are weighed by

precision. I calculate precision as inverse of standard error as it is proven to be the optimal way of calculating weights from a statistical point of view.

When estimates are weighted by precision it is noted that, the size and significance of all measures has changed. A change in the size and significance of estimates indicates that undue influence by estimates is possibly removed. In terms of the effect, positive effect of regulation for instance indicates that more of regulation is good for FDI, whereas in the case of corruption, positive effect indicates that more corruption is still bad for FDI.

2.5.4 MULTIPLE META-REGRESSION ANALYSIS

It can be noted that in spite of weighting these estimates, R^2 values are still low indicating that the above models are weak in explaining the effect of governance on FDI. Hence similar to unweighted results these results can be misleading. One possible reason for a low R^2 value is due to the possible presence of heterogeneity. The expected value of governance FDI estimates will often depend on many other factors such as study, author and journal related. As these factors are unaccounted for, it is possible that both simple unweighted and weighted measures may capture the real effects of governance on FDI. Hence, I include the following moderator variables in order to validate simple meta-regression results. While some of the variables are included out of intuition (author specific variables) others are included as they are proved to have a significant effect by earlier meta studies (Doucouliagos and Ulubasoglu, 2008).

In terms of study related aspects, I have classified all studies into those that are published in journals and others that are not. Estimation techniques used have proven to have an important effect on reported estimates. I have classified studies into those using OLS, panel data, time series, instrumental and other techniques. In terms of the kind of data used, studies are grouped into panel, time series and cross sectional data. Sources of governance and FDI show different effects. In the case of FDI, data sources are grouped as World Bank, UNCTAD, IMF and others. Data sources on governance are classified into World Wide Governance indicators, ICRG, Polity, TI, PRS, Freedom House and others. To test the effect of real world factors, estimates are classified into different regions such as South Asia, East Asia, South East Asia and mixed countries. Dummies for China and South Korea are used to see if inclusion of these countries in the sample countries makes any difference to reported results.

Authors can differ in their values and beliefs which can influence the techniques they use and results they report. In order to capture this effect, I have classified authors based on the university the first author is from as American, European, South and East Asian, and others. I believe journals from different disciplines can differ in reported results due to the rhetorical purposes they aim to

fulfil and the different audience they target. Hence, I have classified journals into Economics and Finance, Business Management and Accounting, Policy and Development. The main results of governance on FDI are shown in table 2.2 and the effect of moderator variables are shown in table 2.3.

Estimation	Political	Government	Regulation	Law	Corruption	Aggregate
techniques	stability	effectiveness				governance
used	(Col. 1)	(Col. 2)	(Col. 3)	(Col. 4)	(Col. 5)	(Col. 6)
Estimates	0.26	0.82	0.63	-0.29	0.28	0.51
weighted by	(8.87)	(2.27)	(13.16)	(-3.69)	(1.08)	(6.10)
precision, β0	Adj.R ² =0	Adj.R ² =0.07	$Adj.R^2=0.$	Adj.R ² =	Adj.R ² =0.8	Adj.R ² =0.6
(Row1)	.94		85	0.85	8	7
Cluster	0.26	0.82	0.63	-0.29	0.28	0.51
analysis, β0	(15.18)	(3.03)	(15.18)	(-12.40)	(4.63)	(1.85)
(Row2)	R ² =0.94	$R^2=0.21$	R ² =0.86	$R^2=0.88$	R ² =0.89	R ² =0.67
Number of	154	34	51	42	166	62
estimates						

TABLE 2.2: MULTIPLE META-REGRESSION RESULTS⁸

Note: Values in parenthesis right below the estimate represent t-values. Each column represents model run with all estimates of each measure of governance.

Results of weighted (row1) multiple regression analysis for each measure of governance is shown in table 2.2. As I have several estimates taken from the same study, it can lead to the issue of potential dependence among estimates which causes bias in the reported results. This potential bias is removed by running MMRA using cluster analysis where each study is treated as a cluster. Results of cluster analysis are used to validate the results obtained by the weighted method.

Before I analyse the results, it is worth noting the following five points. First of all it is important to comment on the good overall fit of the models. With an adjusted R^2 value ranging from 0. 07 for government effectiveness to 0.94 for political stability, these models have done a reasonable job explaining the heterogeneity in governance FDI literature (Stanley and Docouliagos, 2012).

⁸ Results of Precision Effect Test (PET) suggest that there is genuine effect beyond publication bias in case of each measures of governance along with aggregate governance. However, PET results are not robust in case of corruption and aggregate governance.

As compared to R^2 values of simple meta-regression results, the explanatory power of these models has increased after inclusion of moderator variables. Hence, these estimates are more reliable as compared to simple meta-regression estimates.

Secondly, I could not test for endogeneity due to the limited number of estimates (in most cases it was less than 10). Therefore, the effects reported can be due to the possible presence of causality. Thirdly, in terms of the statistical significance, all estimates are statistically significant. In the fourth instance, robustness of all these results is confirmed by cluster analysis. In the fifth instance, with more than 140 estimates and an adjusted R^2 value of more than 0.88, my results are highly reliable for political stability and corruption. In the case of other measures, my results are slightly less reliable as either adjusted R^2 value is implausibly high or they have fewer numbers of estimates. In the sixth instance, all these results are retrieved after removing the effect of outliers⁹.

Firstly, in contrast to the results reported by Globerman and Shapiro (2002a), Zheng (2011), Li and Resnick (2003), Jensen (2003), Jensen & McGillivray (2005), Busse (2004), Blanton & Blanton (2007), Choi (2008) and Doces (2010) my results show that voice and accountability have a negative effect on inward FDI (appendix 2.11). Despite removing the effect of outliers, results for this measure of governance are remained negative and infeasible. These are presented in the appendix. Further research is needed, before any firm conclusions are reached. Nevertheless, negative effect of voice and accountability indicates that low levels of this measure in these countries is associated with high levels of FDI into them. These results reflect the tendency of MNC's to not to invest in countries where people are given voice to express their views and interests on government policies and processes.

Secondly, the overall effect of political stability on inward FDI is found to be positive and significant, which are in line with the findings reported by Anghel (2004), Baek and Qian (2011) and Busse et al., (2011). Therefore in general political stability does matter for foreign investors and it can be assumed that they like to invest in countries with high levels of stability. These results also suggest that foreign investors would not like to see frequent changes in the leadership and that they prefer long term government.

Thirdly, government effectiveness has positive and significant effect on FDI. A positive effect of government effectiveness indicates that higher levels of government effectiveness are correlated with higher levels of FDI. This contrasts the view that foreign investors are not happy with the cumbersome rules and tight procedures that effect the process and productiveness of investments (Khamfula, 2007; Gastanaga et al., 1998 and Arbatli, 2011). However, it is worth noting that with

⁹ Precision more than 200.
the lowest number of observations and a lower R^2 value, results for this measure are not strong enough. The lack of government effectiveness data may have caused biggest challenge in this area of research. Hence, further research is advised in this field of study before any strong conclusions can be made.

In the fourth instance, while on the one hand, effective and efficient policies along with incentives can attract foreign investments (Globerman and Shapiro, 2002a), on the other hand burdensome regulations can negate such investments (Jadhav, 2012). MMRA results on regulatory quality suggest that tighter regulations or regulations enforced in friendly manner are preferred by foreign investors as it has a positive and statistically significant impact on FDI. Therefore my results contrast the view that reducing the regulatory burden and making regulations easier for foreign investors would attract more FDI (Globerman and Shapiro, 2002b).

In the fifth instance, my results on rule of law contrast Arbatli (2011)'s view that a strong and impartial legal system is not preferred by foreign investors as the rule of law has a negative and statistically significant effect on inward FDI. As one would expect stronger laws to facilitate and protect investments, negative effect of law contradicts this view (Anghel, 2004; Gani, 2007; Jadhav, 2012; Fan et al., 2009). This shows a need for host country governments to develop their legal systems further and incline them in favour of foreign investors. Similar to the government effectiveness measure, despite a higher R^2 value, I have limited number of observations for this measure and hence these results must be interpreted carefully.

In the sixth instance, a positive sign of corruption indicates that the higher the corruption, lower is inward FDI. This suggests that foreign investors view corruption as an extra cost of operation rather than viewing it as helping hand. My results are not in line with the literature arguing that corruption is good for foreign investors (Gastanaga et al., 1998; Globerman and Shapiro, 2002a; Teksoz, 2004; Voyer and Beamish, 2004; Khamfula, 2007; Mathur and Singh, 2013). Negative effect inform us that investors prefer not to invest in countries with high corruption or where there is a lack of anti-enforcement laws. Results on corruption confirm the view that corruption sands the wheels of investment rather than greasing them.

Lastly, with 65 observations, aggregate governance has a positive effect on inward FDI. From this result it can be inferred that the higher the governance quality, the more attractive it is for foreign investors. While improved governance is important for the general wellbeing of the individuals, my results suggest that it also helps in attracting foreign investments. My results negate the view that, foreign investors are discouraged by extra cost and delays that are often associated with high levels of governance rather than seeing it as an advantage (Goodspeed et al., 2011). Nevertheless,

 R^2 value is only 0.67 suggesting that the model does not fully explain the effect of governance on FDI.

Based on the higher values of R^2 and with observations of more than 140, my results are strong enough for voice and accountability, political stability and corruption. Hence, I can safely suggest that the countries in South and East Asia & Pacific regions aiming to attract FDI must focus on these three measures of governance. In the case of the other four measures of governance, I see a need for further research to reach any conclusions.

2.5.5 MODERATOR VARIABLES ANALYSIS

	Political Stability		Government Effectiveness		Regulation		Law		Corruption		Aggregate Governance	
If the estimate is estimated using panel data. Reference: If the estimate is estimated using cross sectional data.									0.19 (11.78)	0.19 (43.94)		
If the estimate is taken from a study that is published. Reference: if the estimate is taken from unpublished source.	0.07 (2.86)	0.07 (8.48)										
If the estimate is estimated using levels of FDI. Reference: If the estimate is estimated using natural logarithms of FDI.												

TABLE 2.3: EFFECT OF MODERATOR VARIABLES

If the estimate belongs to a model that includes China in its list of countries. Reference: If the estimate belongs to a model that does not include China in its list of countries.								-0.81 (-3.10)	-0.81 (-24.6)		
If the estimate belongs to a model that includes South Korea in its list of countries. Reference: If the estimate belongs to a model that does not include South Korea in its list of countries.								0.67 (11.24)	0.67 (23.87)	0.94 (7.39)	0.94 (3.07)
If the estimate is estimated by an American author. Reference: If the estimate is estimated by other author.	-0.21 (-7.46)	-0.21 (-15.91)	1.21	1.21	0.20	0.20		0.05 (2.62)	0.05 (6.92)		
If the estimate is estimated by European author. Reference: If			-1.31 (-2.05)	-1.31 (-2.72)	-0.29 (-1.94)	-0.29 (-5.99)					

the estimate is estimated by other												
If the estimate is estimated by									0.71	0.71		
South and East Asian author.									(8.98)	(24.91)		
estimated by other author												
If the estimate is published in			0.60	0.60			0.40	0.40				
Economics and Finance journal.			(1.82)	(2.60)			(7.84)	(29.34)				
Reference: If the estimate is												
published in Law journal.												
If the estimate is published in					-0.28	-0.28					-1.45	-1.45
Business and Accounting						/						
journal. Reference: If the					(-7.96)	(-					(-7.29)	(-2.67)
estimate is published in Law						10.16)						
journal.												
If the estimate is published in	-0.12	-0.12					0.22	0.22				
Policy journal. Reference: If the												
estimate is published in Law	(-4.86)	(-12.27)					(11.16)	(47.61)				
journal.												

If the FDI data for the estimate is taken from IME database	0.72	0.72						-0.18	-0.18		
Reference: If the FDI data for the	(18.53)	(50.46)						(-3.38)	(-5.93)		
estimate is taken from World											
Bank database.											
If the FDI data for the estimate is										-0.45	-0.45
taken from OECD database.										(-5.38)	(-1.70)
estimate is taken from World											
Bank database.											
If the FDI data for the estimate is			-0.63	-0.63		-0.36	-0.36				
taken from other database. Reference: If the FDI data for the			(-1.85)	(-2.48)		(-6.83)	(-30.76)				
estimate is taken from World											
Bank database.											
If the FDI data for the estimate is										-0.42	-0.42
taken from UNCTAD database. Reference: If the FDI data for the										(-6.41)	(-3.52)

estimate is taken from World									
Bank database.									
If the governance data for the estimate is taken from Freedom House database. Reference: If the governance data for the estimate is taken from World Governance Indicators database.	-0.42 (-6.48)	-0.42 (-43.66)							
If the governance data for the estimate is taken from ICRG database. Reference: If the governance data for the estimate is taken from World Governance Indicators database.								-0.43 (-4.01)	-0.43 (-1.83)
If the governance data for the estimate is taken from PRS database. Reference: If the governance data for the estimate						0.21 (3.22)	0.21 (6.76)		

is taken from World Governance										
Indicators database.										
If the governance data for the									-0.43	-0.43
estimate is taken from Polity									(3.40)	(1.80)
database. Reference: If the									(-3.49)	(-1.60)
governance data for the estimate										
is taken from World Governance										
Indicators database.										
If the governance data for the							0.75	0.75		
estimate is taken from										
Transparency International							(14.03)	(23.86)		
database Reference. If the										
governance data for the estimate										
is taken from World Covernance										
Indiastors database										
Indicators database.										
If the estimate is taken from a	-0.05	-0.05			0.15	0.15				
model that is estimated using										
OLS techniques. Reference: If	(-1.78)	(-4.51)			(6.84)	(15.02)				

the estimate is taken from a											
model that is estimated using											
other techniques.											
If the estimate is taken from a	-0.12	-0.12			0.06	0.06					
model that is estimated using	(1 0 0)				(1,70)						
panel data techniques. Reference:	(-4.03)	(-7.26)			(1./8)	(0.60)					
If the estimate is taken from a											
model that is estimated using											
other techniques.											
If the estimate is taken from a							0.68	0.68			
model that is estimated using											
time series techniques							(7.86)	(35.84)			
Pafaranca: If the astimate is											
takan from a model that is											
taken nom a model mat is											
estimated using other techniques.											
If the estimate is estimated using			-0.78	-0.78	-0.43	-0.43	0.21	0.21	-0.33	-0.33	
yearly data of FDI. Reference: If											
the estimate is estimated using			(-2.2)	(-3.09)	(-9.49)	(-	(3.00)	(9.50)	(-7.39)	(-6.98)	
non-yearly data of FDI.						14.88)					

If the estimate is estimated using									-0.12	-0.12		
stock of FDI. Reference: If the									(2.92)	(
estimate is estimated using flow									(-3.63)	(-		
of FDI.										11.43)		
Number of observations	154	154	34	34	51	51	42	42	166	166	62	62
Adjusted R ² /R ²	0.94	0.94	0.07	0.21	0.85	0.86	0.85	0.88	0.88	0.89	0.63	0.67

Note: Only statistically significant variables are shown here. Values in parenthesis are t-values. See appendix 2.6 for full descriptive statistics of moderator variables included in multiple meta-regression.

Before I analyse the effect of moderating variables, it is important to note that except for regulation models using probit model all other results are robust including clustering on the regression. Using the general to specific model, insignificant factors were eliminated (Stanley and Doucouliagos, 2012). Twenty eight variables reflecting the characteristics of study, real world, author and journal have shown to have an important effect on reported estimates. For each of the governance measures, only factors that have caused a noticeable impact on reported results are presented in the table and only interesting, unexpected or surprising results are discussed below.

In the case of study related factors, whether a particular study has been published or not in an academic journal matters as it is statistically significant and have reported higher effects in the case of political stability as compared to estimates from unpublished studies. For instance, published studies on an average have reported a value of 0.33 as opposed to an overall effect of 0.26. Except in case of law, estimates using yearly data on FDI show a negative effect with reference to those using non-yearly data. This could presumably be because governance takes time to show its impact on FDI. There is also evidence to suggest that estimation techniques matter for governance FDI relationship. Models estimated using OLS and Probit techniques proved to be statistically significant compared to estimates estimated using other methods. Governance and FDI data sources also mattered.

Under real world factors, as expected, country composition of the sample did matter as there were few regional specific effects. For instance, models including China in their list of sample countries have reported an average effect of -0.81 which is lower than those which did not include China. Similarly, inclusion of South Korea mattered as reported results are higher (i.e. 0.67) in case of corruption as opposed to an overall effect of 0.28. Thus I infer that governance FDI association did alter with inclusion or exclusion of any particular region. These results are consistent with the notion that there can be many country specific factors that can have an important bearing on how governance works. It is interesting for future research to explore the reasons behind such differential impacts.

In the case of author related aspects, with the exception of political stability, law, corruption and aggregate governance, European authors seem to be consistently different in their results compared to other authors. For instance, reported results of government effectiveness and regulation are weak i.e. -1.31 and -0.29 respectively by European authors than other authors i.e. 0.82 and 0.63 respectively. Such an emphasis on these factors shows that European authors view these factors to be less important than others. Probably because they see government effectiveness and regulation as a part of life, they lay less stress on these factors. Similarly, American authors have emphasised

less on political stability and more on corruption. It is an interesting issue for future research to see why European and American experience is different in these aspects compared to other authors.

I also find that discipline specific journals are statistically significant. For instance, compared to studies from Law, those from Economics and Finance discipline tend to place more emphasis on government effectiveness and law. Surprisingly, studies from Business Management and Accounting discipline under emphasise the importance of regulations and overall governance in attracting FDI. One possible reason for this could be that these disciplines view regulations to be less important in attracting FDI than in protecting such investments. Studies from Policy discipline view law to be more important for FDI. While these results suggest that the type of estimates reported differ across different types of journals, it is interesting to explore this matter further to understand if it is really discipline that's causing the difference or if it is due to some other discipline related factors. The inclusion of other variables which are not reported in the table did not make any difference to reported results.

2.6 CONCLSUIONS

South and East Asia & Pacific countries have during the past decade or so begun liberalising their economic policies in order to create favourable governance environment for FDI. However, whether or not such governance has helped these countries to attract FDI remains inconclusive. The aim of this study was to assess the role of measures of governance on inward FDI in order to reduce the inconclusiveness in this field. Using 771 estimates from 48 empirical studies published from 1980 - 2012, this study meta-synthesised the overall effect of each measure of governance on inward foreign direct investment. The study has also identified factors that have caused heterogeneity in the reported results.

The main message of this study is that each measure of governance has an important effect on FDI. In comparison to less regulated and high corrupt countries meta-regression results show that countries with high regulation and low levels of corruption are able to attract more FDI. Countries with stronger legal systems are positively related to inward FDI. As expected, aggregate governance is found to have a positive effect on inward FDI. It is important to note that with a large number of observations and high R^2 values, my results are strong in the case of voice and accountability, political stability and corruption.

This study has also shown that various study, real world, author and journal related aspects have caused significant difference to reported results in this field of study. An interesting finding that has emerged from this study is that American authors have been shown to be consistently different in reporting effects of government effectiveness, political stability and aggregate governance.

Journal discipline did make a difference to the reported results. As expected, regional effects such as inclusion of China and South Korea in the list of sample countries did matter. Hence the effect of all moderating variables must be taken on board, while interpreting these results.

Despite the useful findings, this study is subject to a number of caveats. The first and foremost caveat of this study is to do with the choice of sample countries and time period. This limitation would mean that the results are restricted to South and East Asia & Pacific countries and can only be generalised to those countries with similar governance and investment conditions. Secondly, in addition to showing direct effects, it is possible that governance affects FDI indirectly through its interaction with macro-economic factors among others. This study has only assessed the direct effects of measures of governance on inward FDI mainly due to the limited and diverse nature of both interaction¹⁰ and non-linear terms¹¹. This has been a common problem with several other meta-analysis studies and thus highlights the need for more extensive research in this field with interaction and non-linear terms.

Thirdly, the quality of results in this study is as good as the quality of studies included for metaregression analysis. In the fourth instance, this chapter offers a general picture on the role of measures of governance on FDI. This limitation means that it does not look into the specific effects of sub measures of each measure of governance on FDI. Last but not least, it is important to note that governance can be measured in terms of the number of assassinations, riots and fines charged for violations of law and not just as a scale. However, I have only included studies which have defined governance as scale, and have excluded those that have defined it in terms of number. Whether or not the results of this study significantly differ if a wider definition of governance is considered is questionable.

The following directions for future research are suggested. Firstly, one important caveat of the empirical studies on measures of governance and inward FDI is that most of the studies have used country as a unit of analysis. Presumably, the effect of governance in attracting inward FDI can differ regionally and is also based on the motive of FDI within one nation. Whether results on the effect of governance on inward FDI would significantly differ if it were possible to carry out research at regional level or by sector is uncertain (Globerman and Shapiro, 2002b).

Secondly, most of the proxies used by existing studies in measuring economic governance in a country are subjective and perception based. The estimations reported by these studies are driven

¹⁰ There were about 15 different types of interaction terms ranging from a minimum of 1 to a maximum of 11 observations.

¹¹ There were only 2 different non-linear terms with less than 12 observations.

by subjective indices. In addition to this, the unanticipated negative effect of governance raises questions on whether these measures actually measure what has to be measured. This leaves an opportunity for future research to use more objective measures of governance by considering factual information on governance such as those provided by using the Business Database provided by World Bank (2006). Another interesting direction for future research would be analysing the effects of economic governance on inward FDI separately by taking up country level studies. This would be informative for the dynamic effects of measures of governance on inward FDI and would also control for country level heterogeneity.

Based on the results of this study it can be safely suggested that without designing and implementing governance in an appropriate manner, attracting high levels of FDI might not be possible. My results have important policy implications. Efforts towards raising the quality of institutions by designing and implementing policies that further political stability, regulation and overall governance is advised. Policy makers should design and enforce policies that lets government be more accountable for its actions along with appropriate legal systems. All possible formal and informal mechanisms that aid in enhancing the quality of accountability of government and those that give more voice to its citizens might be helpful.

As government effectiveness has been shown to have a negative effect on FDI, from an FDI point of view, continuing tighter rules and thereby speeding up the process and productiveness of investments is advised. It is important that the quality of policy formulation and enforcement are in favour of foreign investors along with staying committed to stated policies. Policy makers can focus on improving the regulatory quality to increase their openness to foreign capital. Overall, South and East Asia pacific countries striving to attract FDI should continue to design and implement governance quality in a way that encourages and facilitates investments from foreign investors rather than constraining such investments.

The main contributions of this chapter are twofold. Firstly, based on 771 estimates from 48 studies, this chapter has reduced the inconclusiveness on the role of governance on FDI. All measures of governance i.e. political stability, regulation, law, corruption and government effectiveness along with aggregate governance have an important effect on FDI. In contrast to less regulated and high corrupted countries, countries with tighter regulation and low levels of corruption are able to attract more FDI. On the one hand, countries with high voice and accountability and law are negatively related to FDI. Aggregate governance is found to have a positive effect on FDI.

Secondly, in terms of heterogeneity, studies which are published, those using a specific form of FDI, yearly data, studies published by American, European and Asian authors, studies including

China and South Korea in their sample countries, models estimated using techniques such as OLS, Panel data, instrumental variable and time series, studies using data on FDI from sources such as IMF, OECD, UNCTAD and other, those using data on governance from BERI, Freedom House, ICRG, Polity and other, those published in disciplines such as Economics & Finance, Accounting, Policy and Development studies have caused a significant difference in reported results.

CHAPTER 3

INWARD FOREIGN DIRECT INVESTMENT AND ECONOMIC GROWTH IN SOUTH AND EAST ASIA & PACIFIC REGION: EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALYSIS

PRESENTED AT CAMBRIDGE BUSINESS & ECONOMICS CONFERENCE (CBEC), 2014 SUBMITTED TO OXFORD JOURNAL: AN INTERNATIONAL JOURNAL OF BUSINESS & ECONOMICS (OJ)

3.1 INTRODUCTION

Economic growth is considered to be a function of investment and other factors. While investment can be both domestic and foreign, foreign direct investment in particular is considered to add new investible funds to a host country leading to enhanced economic growth. While there is a theoretical consensus on this aspect, empirically the role of inward FDI on economic growth has been and still is a subject of long and intense debate (Kottaridi and Stengos (2010), Le and Suruga (2005)). Although this continuous debate has provided some insights into the relationship between inward FDI and economic growth, the precise effect of inward FDI on economic growth is still not known either to researchers or to policy makers.

The objective of this study is to address the impact of inward FDI on economic growth empirically with a view to providing a meta-synthesis of the empirical evidence on the direct effects of inward FDI on economic growth in South and East Asia & Pacific countries¹². In particular this study raises the following questions: What do existing empirical studies tell us about the effect of inward FDI on economic growth? Is there any genuine effect of FDI on economic growth? What is the overall effect of inward FDI on economic growth? What is the empirical evidence reported in these studies.

In order to address the above set of questions, this study is outlined as follows. Section 1 gives a brief introduction to inward FDI and economic growth in the case of South and East Asia & Pacific countries followed by theories on economic growth. Section 2 outlines the methodology used in this study and section 3 presents a short and systematic literature review on the effects of FDI on growth. Section 4 presents results of meta-analysis followed by a discussion of results in section 5. The final section of this study has concluding remarks and some implications for policy and future research.

3.2 INWARD FDI AND ECONOMIC GROWTH IN SOUTH AND EAST ASIA & PACIFIC COUNTRIES

Foreign direct investment is an investment by the resident of one country in another with long lasting interest. Long lasting interest is seen when the investor owns a minimum of 10% of the voting power of the direct investment enterprise (OECD, 2008). The main objective of direct investment varies from portfolio investment whereby in the earlier case an investor would expect to influence the management of the direct investment enterprise. Foreign direct investments are

¹² As defined by World Bank and South Korea

made by investors, multinational corporations and other organisations from outside the country in which investment is made (Adeoye, 2009).

South and East Asia & Pacific countries have long pursued the traditional strategy of self-reliance. Foreign direct investments have become topical in South and East Asia region since the late 1980's when most of the countries in the region adopted an open door policy to welcome FDI (for example, India in 1981, China's open door policy in 1978) (Wang, 1995). This change is seen as a result of major political decision and economic development strategy so as to uplift the economies from their economic backwardness and reach their long term goals of development.

In recent times inward FDI into developing Asia has surged tremendously mainly with the liberalisation of investment policies and lowering of capital controls (ABD, 2007). Inward FDI has played a very important role in many regions of South and East Asia & Pacific countries development. While these countries have welcomed varying degrees of inward FDI into these regions, their effect on economic growth has been different based on the investment policies they have adopted. Some light is shed on economic growth and FDI trends in this region from 1980 to 2012.

Appendix 3.1 shows inward FDI and economic growth into these countries from 1980 – 2012. Needless to say, while macro environment in these countries has played a very important role in attracting inward FDI, an equally important role was played by FDI policies. As can be viewed from the graph, there is a clear positive pattern in inward FDI and economic growth in this region. Both FDI and economic growth were lowest in this region in the year 1980 and 1981 respectively and FDI peaked in the year 2002, while showing some steep falls between the periods 1998 and 1999, and 2002 and 2003.

FIGURE 3.1: FOREIGN DIRECT INVESTMENT NET INFLOWS AND GROSS DOMESTIC PRODUCT PER CAPITA IN SOUTH AND EAST ASIA & PACIFIC REGION (*Source: World*



Bank, 2013)

43



13

Fluctuating trends in FDI into South and East Asia & Pacific countries from 1996 can broadly be seen as a result of investment policies in these countries and also as a result of external factors such as currency appreciation (Figure 3.1). On the one hand, looking at the history of investment policies of East Asian countries from 1980, governments initially restricted FDI into these countries in order to promote and protect domestic companies. Countries such as Malaysia, Thailand and Indonesia had different policies for different industries. While investment was completely restricted in certain strategic industries, it was limited in others (Thomson, 1999).

Moreover, countries that have initially allowed FDI as a part of import substitution policy have later moved to export promotion strategies. In terms of external factors, currency appreciation of Yen around the 1980's has made it expensive to manufacture labour intensive goods. As a result, Japan started looking for other countries in Asia where labour costs were cheap. Yen appreciation has also created a wealth effect which led to an increase in inward investments to East Asian countries such as South Korea and Taiwan and later to China (Willem and Salike, 2013)

On the other hand, investment policies have been restrictive in South Asia until the 1990's when most of the countries in this region has opened up their doors and made it conducive for foreign investors (Sahoo, 2006). Most of the countries have also used tax incentive policies in order to attract FDI to promote employment opportunities, develop rural areas and the development of specific industries. Overall, inward FDI was regulated differently with differing degrees of efficiency by countries in this region.

¹³ Based on authors calculations. GDP per capita is calculated by dividing total GDP by total population of the region. Data on total GDP and total population was obtained from World Bank website.

3.3 METHODOLOGY

Methodology used in this study is informed by the same sources listed in chapter 1 section 1.3. Information relating to the first stage of methodology can be found in chapter 2, section 2.3.1. Search keywords were used for FDI and growth to search 'title', 'abstract', 'text' and 'keyword' in databases listed above with the time period as January 1980 – December 2012 are listed in appendix (appendix 3.2). Only studies published in the English language were used in this present study. Stages involved in the search process are detailed in the following diagram.

My initial search has retrieved 12863 studies that have looked at the effect of FDI on economic growth. From these studies 933 and 252 duplicate records were identified and removed by using duplicate search option in endnote and by hand search respectively leaving 11678 unique studies for the next stage. First stage screening of these unique studies was done by reading title and abstract only which resulted in 419 suitable for this study. The relevance of each study was ascertained by interrogating it with one question: Does the study estimate the relationship between inward FDI and economic growth? If a study does not, they are deselected and are not included in meta-analysis.



FIGURE 3.2: SUMMARY OF METHODOLOGY

The critical evaluation of full text of these studies was achieved based on PIOS (population - independent variable - outcome variable - study design) criteria as suggested by the University of York (CRD, 2009) (appendix 3.3). 32 empirical studies were found to satisfy all four criteria (appendix 3.4 and 3.5) to which a further 5 studies were added by hand search making a total of 37 empirical studies. Information on the composition of published and unpublished studies is given in appendix 3.6.

The following data were obtained from above retrieved 37 empirical studies:

a. Bibliographical information – name of the author, year of publication, type of paper (published paper, working paper or conference paper)

b. Study characteristics – Study type, study design, nature of data used, information on dependent and independent variables (functional form, data source)

c. Estimation methods used – ordinary least squares techniques, panel data techniques, time series techniques and instrumental variable techniques.

d. Outcome reported – estimated parameters for all independent variables, standard errors or t – statistics of the estimates. Effect sizes¹⁴ associated with linear, interaction and non-linear terms are all included in this study.

Two forms of econometric models were used in primary studies. First, models with only linear terms (equation 3.1) and second, models with linear, non-linear and interaction terms (equation 3.2). The econometric model with only linear terms can be expressed as follows:

$$Y_{it} = \alpha_0 + \alpha_1 F_{it} + \gamma X_{it} + \varepsilon_{it}$$
 equation (3.1)

The econometric model with linear, non-linear and interaction terms is expressed as follows:

$$Y_{it} = \alpha_0 + \alpha_1 F_{it} + \alpha_2 F_{it} \cdot K_{it} + \alpha_3 F^2_{it} + \gamma X_{it} + \varepsilon_{it} \qquad \text{equation (3.2)}$$

In equations 1 and 2, α_0 is the constant term and α_1 measures the marginal effect of F on Y; F stands for variable of interest i.e. inward FDI; F_{it} measures the linear effect of inward FDI on economic growth; $F_{it} \cdot K_{it}$ is the interaction term which measures the effect of F on economic growth conditional on the value of K; F^2 is non-linear term and α_3 measures the effect of F^2 on economic growth conditional on its own value. X_{it} is the vector of other variables that might affect

¹⁴ "Effect size is a measure of the strength (magnitude) and direct of a relationship between variables" (Littell, Corcoran and Pillai, 2008, p.80)

the dependent variable; y measures the marginal effect of X_{it} on Y; i and t are country and time indices respectively. E is the random error term. Interaction terms and non-linear terms are useful in identifying the marginal effect of inward FDI on economic growth.

Partial correlation is used as a standardised measure of the effect of FDI on economic growth. The beauty of partial correlation is that it allows for meaningful comparison across models. All values of α_1 , α_2 , α_3 were transformed into partial r using the formula: $r = [t/\sqrt{t^2 + dof}]$. Where, t stands for t –statistics of the multiple regression coefficient and dof stands for the degrees of freedom of the respective t –statistic. With the exception of the following variable, simple and multiple meta-regression equations (1.3, 1.4 and 1.5) used in this study are similar to the ones shown in Chapter 2, page. 12.

3.4 LITERATURE REVIEW

This section briefly reviews the literature on inward FDI and economic growth highlighting the inconsistencies between the empirical studies in order to shed some light on the reasons for the different findings and also to draw hypothesis to test using meta-regression analysis. The study aims to answer these questions specifically: 1. what is the effect of inward FDI on economic growth of the host country and how big is that effect? 2. What factors cause differences in empirical results within this field?

3.4.1 THEORETICAL VIEWS ON FDI-GROWTH NEXUS

Under the neoclassical growth model, FDI is considered to be a pure factor input and the long term effects of FDI are neutral. Studies based on neoclassical growth theory argue that the effects of FDI on the host country's economic growth are only in short term and it leaves long run growth unchanged. These scholars are of the view that long run growth can occur only when the quantity (for example population growth) and quality of resources (for example technological progress) in an economy are enhanced, both of which are considered to be exogenous. In contrast to this, under the endogenous growth model, FDI is considered to be a delivery vehicle to transfer technological, knowledge and know-how from the investing country to host country (Li and Liu (2005), Borensztein, Gregorio and Lee (1998), Balasubramanyam, Salisu and Sapsford (1996)). As a result, FDI will be able to have positive effects on the host country's economic growth in the long term (Makki and Somwaru (2004)).

From among these studies, positive and statistically significant results are reported by Alguacil, Cuadros and Orts (2011), Anwar and Cooray (2012), Ahmad and Hamdani (2003), Balasubramanyam, Salisu and Sapsford (1996), Basu and Guariglia (2003), Busse and Groizard (2008), Freckleton, Wright and Craigwell (2012), Hsiao and Shen (2003), Kotrajaras (2010), Kottaridi and Stengos (2010), Le and Suruga (2005), Lee, Lee and Kim (2011), Lensick and Morrissey (2006), Li and Liu (2005), Makki and Somwaru (2004), Sylwester (2005), Thangavelu, Yong and Chongvilaivan (2009), Vita and Kyaw (2009) and Wang and Wong (2010). Positive and statistically insignificant results are reported by Alfaro (2003), Alfaro, Chanda, Kalemli-Ozcan and Sayek (2004), Balasubramanyam, Salisu and Sapsford (1996), Carkovic and Levine (2002), Economidou, Lei and Netz (2006), Kottaridi and Stengos (2010), Makki and Somwaru (2004).

Negative effects of FDI can be attributed to Alfaro, Kalemli-Ozcan and Sayek (2009), Borensztein, Gregorio and Lee (1998), Durham (2004), Fry (1996), Hermes and Lensink (2003), Herzer (2012), Le and Suruga (2005), Vita and Kyaw (2009), Wang and Wong (2011). From these studies, significant results are reported by Borensztein, Gregorio and Lee (1998), Hermes and Lensink (2003), Le and Suruga (2005), Vita and Kyaw (2009), Wang and Wong (2011). In contrast to these studies, insignificant results are reported by Alfaro, Kalemli-Ozcan and Sayek (2009), Durham (2004) and Fry (1996).

In the case of single country studies, positive and statistically significant effects of FDI can be attributed to Baharumshah and Almansaied (2009) for Malaysia from 1974 - 2004, Acharyya (2009) for India from 1980 - 2003, Ahmed (2012) for Malaysia from 1999 - 2008, Ang (2009) for Thailand from 1970 - 2004, Chen, Chang and Zhang (1995) for China from 1968 - 1990, Hoang, Wiboonchutikula and Tubtimtong (2010) for Vietnam from 1995 - 2006, Quader (2009) for Bangladesh from 1990 - 2006, Yu and JingMei (2009) for China from 1991 - 2007. Choong, Yusop and Soo (2005) study on Malaysia from 1970 - 2001 finds negative and statistically significant results.

Empirical evidence reviewed so far on the growth effects of FDI are inconclusive or at least inconsistent. As noted above, the effect of FDI on economic growth can be positive and statistically significant, positive and statistically insignificant, no effect, negative and statistically insignificant and negative and statistically significant. While the differences in data, time period of study, methodology are generating these conflicts among empirical findings (appendix 3.8) the role and impact of FDI seems to be more country specific and can differ based on the host country's economic, institutional, technological and other factors (Li and Liu (2005)). Conflicting research results overwhelm any clear understanding on the effect of FDI on economic growth. This restricts the ability of researchers in suggesting and policy makers in implementing appropriate policies to promote economic growth.

As a remedy for inconclusive empirical results, various scholars have tried different methodologies by differentiating developed and developing countries, export promoting countries and import substitution countries (Balasubramanyam, Salisu and Sapsford (1996)) and by using advanced econometric techniques such as instrumental variable techniques in order to control for endogeneity problem (Alguacil et al., 2011, Alfaro et al., 2003, Alfaro et al., 2004), Anwar and Cooray 2012, Azman-Saini et al., 2010) (appendix 3.4.2.1). While these new techniques have created additional insights into this topic, empirical results still remain inconclusive. Hence, an intelligent summary of these findings is likely to lead to informed policy decisions (Stanley and Doucouliagos, 2012).

Despite differences in reported results, one common point among these studies is that they suggest that the growth enhancing effect of FDI is not automatic but is likely to depend on various country specific factors such as economic, technological and institutional. For instance, while on one hand Alfaro et al., (2003) shows that FDI effects are conditional upon sufficiently developed financial markets, on the other hand Balasubramanyam et al., (1996) suggest that the effect depend on upon trade policy. Despite this fact, it is important to remember that there are no widely accepted country specific factors that are identified by the literature. Hence, if the growth effects of FDI are positive or negative in some economies under some conditions, they may not be valid for all countries.

One problem in assessing the effects of FDI on economic growth is endogeneity, which arises due to interdependence of FDI and economic growth. FDI might have a positive impact on the host economy leading to market expansion. An expanded market in turn can attract further FDI. Hence, ignoring this problem might lead to reverse causality or simultaneity (Alguacil et al., 2011). Studies by Alguacil, et al., 2011, Alfaro et al., 2003, Alfaro et al., 2004), Anwar and Cooray 2012, Azman-Saini et al., 2010, Basu and Guariglia, 2003, Beugelsdijk et al., 2008, Borensztein et al., 1998, Busse and Groizard 2008, Durham 2004, Fry 1996, Kottaridi and Stengos 2010, Lensick and Morrissey 2006, Makki and Somwaru 2004, Thangavelu et al., 2009, Vita and Kyaw 2009 and Wang and Wong 2010 have used different instrumental techniques in order to understand the true effect of inward FDI on economic growth (appendix 3.8).

3.5 DISCUSSION OF RESULTS

I present and analyse empirical results in this section. I start with funnel plot and chronological order of estimates. These graphs are used to illustrate the distribution of empirical findings in FDI growth studies. Thereafter, simple and multiple meta-regression results are presented and analysed. An overview of meta-regression analysis is shown in appendix 3.9.



FIGURE 3.3: FUNNEL PLOT FOR FDI GROWTH ESTIMATES

633 estimates of FDI-growth nexus are plotted on funnel plot as shown in figure 1. Funnel plot shows association between the effect size and its precision. Effect size (partial r) is shown on X axis and weight of effect i.e. precision (calculated as inverse of standard error of each partial r) on Y axis.

Three observations can be inferred from the funnel plot. First, the average effect of FDI-growth is about 0.1369¹⁵. This is the reliable summary estimate of all estimates included in this study (the mean effect of the top 3% of estimates is about 0.2140). Secondly, there is a wide variation in the empirical estimates which are both large and small, and positive and negative. There are about 586 positive and 165 negative estimates. Thirdly, estimates with large precision (estimates with precision more than 500 are 18) are few and are compactly distributed on the top of the funnel while estimates with low precision are many and are widely distributed at the base of the funnel and form tails on both sides. Relatively there is more agreement among high precision estimates on FDI-growth effect as opposed to low precision estimates.

¹⁵ I have tested for publication bias using Funnel asymmetric test (FAT) and Precision effect test (PET) (appendix 3.10). Despite the presence of publication bias, PET results suggests that there is genuine effect of FDI on growth (except for South Asia, where PET results are not robust). I explore these aspects further in a different paper on publication bias in governance-growth studies.

3.5.2 CHRONOOGICAL ORDER OF ESTIMATES

FIGURE 3.4: CHRONOLOGICAL ORDER OF ESTIMATES BASED ON AVERAGE YEAR OF SAMPLE PERIOD



Figure 3.5.2 shows chronological order of FDI-growth estimates arranged in the order of average year of sample period. An upward trend can be seen in the results reported on the effects of FDI on economic growth. It can be noted that there is an increase in the number of positive estimates reported after 1995. This confirms the view that FDI takes time to show its positive effects on economic growth.

3.5.3 SIMPLE META-REGRESSION ANALYSIS

Statistic	All estimates	Estimates controlling	Estimates for
		for endogeneity	South East Asia
	(Col. 1)	(Col. 2)	(Col. 3)
Un weighted, β0	0.08	0.09	0.15
	(5.27)	(4.56)	(5.30)
	R ² =0.04	R ² =0.04	R ² =0.26
Weighted by	0.18	0.26	0.15
precision, β0	(17.29)	(27.09)	(5.12)
	$R^2=0.13$	R ² =0.30	$R^2=0.48$
Number of	624	229	77
estimates			

TABLE 3.1: SIMPLE META-REGRESSION RESULTS

Note: Values in parenthesis right below the estimate represent t-values.

Simple unweighted and weighted meta-regression results are presented in table 3.5.3. I have used four different models as follows: for all estimates, estimates controlling for endogeneity, for East Asia and for South East Asia in columns 1 to 4 respectively. Row 1 displays unweighted least square results and row 2 displays weighted least square estimates. Except for East Asia, unweighted estimates of FDI show positive effect on growth which indicates that FDI has a growth enhancing effect in all cases. However, these effects are unreliable for two reasons.

Firstly, because unweighted method treats all estimates with equal weight. This means if there are more estimates coming from one study, then they will have an undue influence on the overall effect. Secondly, R^2 value of each of these models is low (ranging from 0.04 for all estimates to 0.48 for South East Asia). These low values suggest that the models do not explain the complete effect of FDI on growth.

Hence, following Stanley and Doucouliagos (2012), I run weighted least squares model, where weight is defined as inverse of standard deviation. Once the estimates are weighed, size and the significance of the effects have changed. An interesting point here is that while column 1 for all estimates shows positive effect of FDI on growth, column 2 for estimates controlling for endogeneity also shows positive effect. These results tell us that after controlling for endogeneity, the true effect of FDI still remains to be positive. Hence, I infer that FDI has growth enhancing effects. Nevertheless, R^2 values have only improved a little which tells us that these models are

still showing unreliable effects of FDI on growth. Due to the presence of potential heterogeneity, simple unweighted and weighted measures may not capture the real effects of FDI on growth. I address this potential heterogeneity by using all coded moderator variables in multiple meta-regression analysis.

3.5.4 MULTIPLE META-REGRESSION ANALYSIS

The following moderator variables are included in the multiple meta-regression analysis. Most of these moderator variables are included as they are proven to be significant in other meta-analysis studies dealing with economic growth (Doucouliagos and Ulubasoglu, 2008; Doucouliagos and Paldam, 2007, 2009; Abreu et al., 2005).

Study characteristics – Here difference in study based on whether or not it is published in a journal is controlled. As authors use different functional forms and data sources for FDI and growth, this is controlled for. Estimation techniques have proven to be an important source of heterogeneity. Hence, I differentiate them into OLS, panel, time series, instrumental variable and other techniques. Studies using cross sectional data have been proven to report higher effects. Therefore I differentiate data used in these studies into panel, time series and cross sectional data. Researchers have also proved that average data removes any fluctuations in the growth, hence I control for this difference using yearly and average data variables.

I have differentiated FDI based on its purpose as Greenfield or Mergers and Acquisition. I control to see if observations reported in a study make any variation to the reported results as compared to studies not reporting observations. Omission of relevant explanatory variables such as education, population and domestic investment can have an impact on the estimated coefficient (Barro, 1991).

Real world factors – Firstly, I control for country composition of sample countries by grouping them into South Asia, East Asia, South East Asia and other countries. I also control for China and South Korea effect by using dummy variables.

Author characteristics – I merely wish to test if author origin makes any difference to FDI growth estimates. Hence, I differentiate authors based on the university of the first author as American, European, South and East Asian and others. I would also like to test if authors coming from prestigious universities like IVY league and Oxford/Cambridge report any different effects. Hence this difference is also controlled.

Journal characteristics – Differences in journals are controlled based on their discipline i.e. Economics and Finance, Science, Law, Development, Geography, Management and Policy. To see the impact of journal ranking and citations I use ABS 2010 rankings.

Statistic		All-estimates	Estimates	Estimates for South
			controlling for	East Asia
			endogeneity	
		(Col.1)	(Col.2)	(Col.3)
Weighted	by	0.83	0.29	0.61
precision,	β0	(20.25)	(39.76)	(11.36)
(Row1)		$Adj.R^2 = 0.92$	$Adj.R^2 = 0.79$	$Adj.R^2 = 0.78$
Cluster, β0		0.83	0.29	0.61
(Row2)		(7.33)	(79.59)	(18.28)
		$R^2 = 0.92$	$R^2 = 0.80$	$R^2 = 0.80$
Number	of	562	229	73
estimates				

TABLE 3.2: MULTIPLE META-REGRESSION ANALYSIS RESULTS¹⁶

Note: Values in parenthesis right below the estimate represent t-values.

Table 3.2 above shows multiple meta-regression analysis results. I have run four models, all estimates, estimates controlling for endogeneity, East Asia and South East Asia. Row 1 shows the results of weighted least squares and row 2 shows cluster regression analysis results which I use for robustness check. Due to limited number of estimates i.e. observations fewer than 30, results are less reliable for East Asia and South Asia (these are presented in appendix 3.11). All these results are retrieved after removing the effect of outliers¹⁷.

As expected, all estimates model shows a positive effect of FDI on growth. As this positive effect is also confirmed by estimates controlling for endogeneity, with an R^2 value of 0.92 and 562 observations, my results are in strong favour of the view that FDI has a growth enhancing effect in this region. I see four possible reasons for such positive effect. Firstly, this could be due to low reverse flows to home countries in the form of profits, dividends. Secondly, multinational companies in these countries have obtained limited concessions from the host country

¹⁶Results of Precision Effect Test (PET) suggests that there is genuine effect of FDI on economic growth in case of all models. However, PET results are not robust in case of South Asia estimates. Hence, the results in case of South Asia should be interpreted with caution.

¹⁷ Precision more than 800.

governments (Sahoo, 2006). Either of these two possibilities can result in the possible positive effect of FDI on growth. Thirdly, policy regime in these countries might have created a favourable climate to reap the benefits of FDI. In the fourth instance, positive effect of FDI on growth can arise when FDI does not crowd out domestic investment. As this study does not address the reasons behind such a positive effect, it is worthy of future studies to look into this.

Similar to all estimates model, in the case of South East Asia, FDI has a positive effect on economic growth. However, the effect is bigger compared estimates controlling for endogeneity. A positive sign indicates that FDI has growth enhancing effects in the case of South East Asia. By having an open policy regime, allowing foreign investments and increasing economic activity, it is not surprising to see such results (Sahoo, 2006). It is important to note here that I only examine direct effects of FDI on economic growth. It is also possible that FDI has an indirect positive effect on economic growth in these two cases through its interaction with factors such as technology, human capital and financial markets among many others. However, I could not test this due to the very diverse nature and few interaction terms reported in primary studies.

In the case of East Asia, an unforeseen negative sign is shown (as there are only 17 observations, these results are presented in appendix 3.11). Negative effect here indicates that FDI has a growth retarding effect for East Asia. While this result is surprising, it is also in contrast with those reported by Zhang (2001a, 2001b). Many factors can be identified from FDI growth literature that could have resulted in positive effects of FDI on growth. For instance, Balasubramanyam et al., (1996) and Mencinger (2003) show that growth enhancing effects of FDI are high in countries that follow export promotion policies as compared to import substitution policy. Borensztein et al., (1998) show that the growth promoting effects of FDI depend on the existing capital stock of the host countries. Alfaro et al., (2004) show that well developed financial markets aid in realising positive effects of FDI on growth. Despite, most of the East Asian countries following these policies, it is surprising to these results.

While the presence of the above noted conditions would have created an ideal climate for exploiting the potential of FDI in promoting economic growth in East Asia, my study does not explore the reasons behind such effect. Results for East Asia must be interpreted carefully as the number of observations is only 17. Further empirical research is advised before any firm conclusions are made in the case of both South Asia and East Asia. Overall, the results presented above suggests that FDI does not have a uniform direct effect on economic growth in all regions and that the effect is region specific. Future studies might want to study the causes behind region specific effects of FDI on growth.

3.5.5 MODERATOR VARIABLES ANALYSIS

I have identified several variables that have significantly influenced the reported effect of FDI on growth. I only discuss some interesting and unexpected results here.

TABLE 3.3: EFFECT OF MODERATOR VARIABLES

Moderator variables	All estima	tes	Estimates	controlling	South Ea	st Asian
			for endoge	eneity	estimates	5
	WLS	CLUST	WLS	CLUSTE	WLS	CLUSTE
		ER		R		R
If the estimate belongs to a model that is estimated for single	0.25	0.25				
country. Reference: If the estimate belongs to a model that is	(10.62)	(4.02)				
estimated for multiple countries.						
If the estimate is estimated using relative figures of FDI data.	0.13	0.13				
Reference: If the estimate is estimated using natural logarithm	(6.70)	(3.94)				
values of FDI.						
If the estimate is estimated using levels of FDI data. Reference: If	-0.26	-0.26				
the estimate is estimated using natural logarithm values of FDI	(-2.23)	(-7.15)				
If the estimate is estimated using OLS techniques. Reference: If	0.38	0.38			-0.42	-0.42
the estimate is estimates using other techniques.	(8.64)	(3.33)			(-2.14)	(20.73)
If the estimate is estimated using instrumental variable techniques.	-0.12	-0.12				
Reference: If the estimate is estimates using other techniques.	(-5.86)	(-5.86)				
If the estimate is estimated using time series techniques.	0.42	0.42				
Reference: If the estimate is estimates using other techniques.	(9.04)	(3.65)				
If the estimate is estimated for East Asian countries. Reference: If	-1.56	-1.56				
the estimate is estimated using mixed countries data.	(-27.3)	(-9.32)				

If the estimate is estimated for South East Asian countries.	-0.76	-0.76				
Reference: If the estimate is estimated using mixed countries data.	(-19.4)	(-6.95)				
If the estimate is estimated for South Asian countries. Reference:	-0.39	-0.39				
If the estimate is estimated using mixed countries data.	(-10.48)	(-3.92)				
If the estimate belongs to a model that includes China in its list of	-0.15	-0.15			-0.16	-0.16
countries. Reference: If the estimate belongs to a model that does	(-5.27)	(-2.39)			(-2.49)	(-2.81)
not have China in its list of countries.						
If the estimate belongs to a model that includes South Korea in its	0.18	0.18			-0.18	-0.18
list of countries. Reference: If the estimate belongs to a model that	(9.29)	(2.38)			(-3.10)	(-10.61)
does not have South Korea in its list of countries.						
If the estimate is estimated using Greenfield data. Reference: If the	-0.09	-0.09				
estimate is estimated using aggregate FDI.	(-1.78)	(-1.19)				
If the estimate is estimated using time series data. Reference: If the					0.59	0.59
estimate is estimated using cross sectional data.					(6.48)	(11.82)
If the estimate is estimated using regional level FDI data.	1.32	1.32				
Reference: If the estimate is estimated using economy level FDI	(18.13)	(7.21)				
data.						
If the estimate is estimated by American author. Reference: If the	-0.47	-0.47	0.04	0.04		
estimate is estimated by other author.	(-9.73)	(-4.17)	(2.90)	(8.08)		
If the estimate is estimated by European author. Reference: If the	-0.38	-0.38	0.03	0.03		
estimate is estimated by other author.	(-8.54)	(-2.73)	(2.07)	(4.77)		

If the estimate is estimated by South and East Asian author.					-0.35	-0.35
Reference: If the estimate is estimated by other author.					(-5.6)	(-19.65)
If the estimate belongs to Business and Accounting journal.	-0.67	-0.67				
Reference: If the estimate belongs to Law journal.	(-13.25)	(-5.25)				
If the estimate belongs to Development journal. Reference: If the	-0.61	-0.61				
estimate belongs to Law journal.	(-11.41)	(-6.41)				
If the estimate belongs to a model that includes population related	-0.15	-0.15	-0.15	-0.15		
variable. Reference: If the estimate	(-3.73)	(-1.78)	(-10.52)	(-19.50)		
No. of observations	562	562	229	229	73	73
Adjusted R2	0.92	0.92	0.92	0.92	0.78	0.80

Note: Only statistically significant variables are shown here. Values in parenthesis show t-values.

See appendix 3.7 for full descriptive statistics of moderator variables included in multiple meta-regression.

In terms of study related factors, type of FDI, data types, estimation techniques matter for the reported results. As shown by other meta-regression studies, estimation techniques mattered. On one hand, models estimated using OLS and times series techniques reported higher effects in case of model with all estimates compared to those estimated using other techniques. On the other hand, instrumental variable techniques have reported lower effect. For instance, instrumental variable techniques have reported on an average 0.71 which is lower by 0.12 than overall average effect of 0.83. In case of South East Asia, models estimated by OLS have reported lower effects of FDI on growth. As expected, I find that reported results differ among studies based on how researchers measure FDI. For instance, relative figures of FDI report lower effects in models with all estimates compared to these variables expressed in terms of natural logarithms. Those using FDI levels reported lower effect of 0.83.

The magnitude of effect also differed among studies based on real world factors. In case of all estimates model, while studies including China have reported lower effect by 0.15 and those including South Korea have reported lower effect by 0.18 than overall average effect of 0.83. Similarly, in case of South East model, estimates including China have reported an average of 0.45 and those including South Korea have reported an average of 0.43. These results suggest that, in spite of an increase in FDI flows into these regions, FDI in general has mixed effects on growth.

Author and journal related factors have shown noticeable effects on reported results. First, my intuition that the variation in the empirical estimates can partially be explained by the first author from different regions or universities is correct. In case of all estimates model, American and European authors have reported lower effects by -0.47 and -0.38 as compared to overall value of 0.83 reported by other authors. Possibly these authors value FDI to be less important for growth. Journals from Business Management and law discipline reported lower effects of FDI on economic growth. This could be because journals from these disciplines capture lower affect due to differences in the econometric techniques they use. The notion that estimated effects vary based on journal ranking and citations did not prove to be right in this study.

3.6 CONCLUSIONS

Using Meta-regression analysis, this study provided an average effect of inward FDI on economic growth obtained from weighted least squares for 633 estimates from 37 empirical studies for South and East Asia & Pacific countries. Meta-regression analysis is used to summarise and distil lessons from a body of econometric evidence in FDI-growth field. This study started by reviewing literature on FDI-growth systematically and identified possible reasons for variation in the empirical studies.

In case of model with all estimates, contrast to earlier studies on FDI growth (Borensztein, Gregorio and Lee (1998), Hermes and Lensink (2003), Le and Suruga (2005), Vita and Kyaw (2009), Wang and Wong (2011)), the results of this study indicate that FDI has a positive and significant effect. The same positive effect does hold true for estimates controlling for endogeneity and this could mean that FDI does have a genuine positive effect on FDI. FDI has shown a negative effect in the case of East Asia and a positive effect in the case of South East Asia. It is worth noting that the results in the case of East Asia and South Asia are less reliable as the number of observations are fewer than 30 (appendix 3.11). In terms of variations in studies, this study has identified many related, real life and journal related aspects that have caused a significant difference to the reported estimates.

Similar to any other meta-analysis studies, the present study has the following four caveats. Firstly, as the present study describes the research record in inward FDI and growth at a point in time, the results obtained cannot be used as a forecasting tool. Future research might consider updating this dataset and comparing the predictions made in this study with the subsequent ones to see if the findings of this study hold over time. Secondly, as the study has no control over primary econometric studies, any possible measurement or reporting error in primary studies is carried over to this study.

Thirdly, since there are a range of studies included in the present study, the issue of study quality and its effect on statistical inference can arise. This study has assigned more weight (based on precision) to estimates with higher quality and vice versa to address this issue (Doucouliagos, et al., 2010; Stanley and Doucouliagos, 2012). At last, data dependency can be seen as one problem in meta-analysis especially when there are multiple estimates reported in each study. This can violate assumptions of equation 3.1 and 3.2 which assume that estimates are statistically independent. In order to overcome this problem, clustered data analysis was used for robustness check that reduced the level of standard errors by clustering observations within a study (Doucouliagos et al., 2010).
In terms of research implications, the following three suggestions are made. Firstly, future research can focus more on country specific studies as the effect of FDI on economic growth varies from country to country based on its absorptive capacity. Currently there are very few studies examining FDI-growth relationship at country level (Acharyya (2009), Ahmed (2012), Ang (2009), Baharumshah and Almasaied (2009), Hoang et al., (2010), Quader (2009)). Secondly, it might also be interesting to analyse the reasons for the negative effect of FDI on growth.

Thirdly, Literature so far with the exception of Wang and Wong¹⁸ (2010) and (Beugelsdijik, et al¹⁹ (2008) has focused on understanding the effects of aggregate FDI on economic growth. Aggregate FDI does not always help in understanding the heterogeneous growth effects of different modes of FDI. Because cross border mergers and acquisitions involve buying existing entities and Greenfield investments involve starting up a new entity, these two forms of FDI are likely to have different effects on economic growth (Wang and Wong, 2010). Hence, future researchers can study this relationship by differentiating FDI into Greenfield and Brownfield.

Based on the results of this study, the following policy implications are suggested. South East Asian countries should continue to attract FDI as it has proved to have growth enhancing effects. A favourable economic environment that helps to reap the benefits of FDI for growth is suggested for these countries. As these countries already have FDI policies in place, it is worth focusing on appropriate policy enforcement so as to realise the positive effect of FDI on economic growth.

¹⁸ Wang and Wong (2010) differentiate inward FDI as Greenfield and mergers and acquisitions

¹⁹ Beugelsdijik, Smeets and Zwinkels (2008) differentiate US FDI as vertical and horizontal FDI

CHAPTER 4

ECONOMIC GOVERNANCE AND ECONOMIC GROWTH IN SOUTH AND EAST ASIA & PACIFIC REGION: EVIDENCE FROM SYSTEMATIC LITERATURE REVIEWS AND META-ANALYSIS

PRESENTED AT ASSOCIATION OF HETERODOX ECONOMICS CONFERENCE, 2014 SUBMITTED TO ADVANCES IN ECONOMICS AND BUSINESS

4.1 INTRODUCTION

Research on economic growth (hereafter referred to as growth) in general and particularly in the case of South and East Asia and Pacific counties has exploded in the last few years (Zhang, 2001). The economic growth literature is filled with empirical studies that have looked at the elusive and ever important question of what causes economic growth (Anwar and Cooray, 2012; Haggard and Tiede, 2011). The focus of most of the empirical studies in this field has been on conventional sources of economic growth such as domestic investment, education, foreign investment and others. With economists and policy makers recognising the role of economic governance (hereafter referred to as governance) for growth, recent research focus is on governance and its impact on growth. However, with ever growing number of studies using different methodologies, data sources and country groupings, a high amount of heterogeneity is created among reported results. This has left both policy makers and researchers having different views on the importance of governance for growth.

The relationship between governance and growth has been a highly debated topic in the Asian context. While some authors argue that governance shows positive effects on growth, others are of the view that it is not the case. As governance establishes the framework for economic activity within a country good governance on one hand can create an environment that promotes economic activity, provides incentives to invest and economic growth. Bad governance on the other hand can have detrimental effects on economic growth by increasing transaction costs and by causing delays in the investment process (Kaufmann et al, (1999a), Gani (2001)). This study is motivated by increased effort from both policy makers and researchers towards understanding the overall impact of governance on economic growth and improving the governance quality in general.

The aim of this study therefore is to contribute to evidence based policy making and to academic research on the governance growth relationship by providing meta-synthesis of empirical evidence on various measures of governance and growth. This study also identifies factors causing heterogeneity in results, pointing to policy implications of the results and identifying potential avenues for future research. In order to address the aims of this study, the following questions are raised in this study: Firstly, is there any genuine effect of governance on economic growth? Why do governance growth studies report such divergent results? Is the heterogeneity due to the data generating process or is it due to differences in research design?

After the above introduction, the rest of the paper is organised as follows. Section 4.2 outlines the methodology used in this study. Section 4.3 presents a brief overview of systematic literature in the case of South and East Asia and Pacific countries from 1980 - 2012. Section 4.4 presents

analysis and discusses the results, followed by Section 4.5 which concludes this study by outlining the limitations of the present study, together with some policy and research implications.

4.2 METHODOLOGY

Methodology used in this study is informed by the same sources as listed in chapter 1 section 1.3. Information relating to the first stage of methodology can be found in chapter 2, section 2.3.1. The key words that were used to search 'title', 'abstract', 'text', and 'keyword' in databases listed above are listed in the appendix (appendix 4.1). The time period of the search was January 1980 – December 2012. With regards to the language of publication, studies published in English language only were used.

FIGURE 4.1: SUMMARY OF METHODOLOGY



Searching databases for both published and unpublished studies, 5414 were retrieved. From these, 875 and 168 records were deleted through automatic and manual duplicate search respectively. This resulted in 4371 unique records that either analysed or estimated the relationship between economic governance and economic growth. First stage screening of these unique studies was done by reading title and abstract of each study which reduced the number of studies to 91 (fig 4.1). The relevance of each study was interrogated with two questions: Firstly, does the study estimate the relationship between economic governance and economic governance and economic governance and economic studies was interrogated with two questions: Firstly, does the study estimate the relationship between economic governance and economic growth? Secondly, does the

study analyse the relationship between economic governance and economic growth? Only studies which have estimated the relationship were considered for critical evaluation stage.

Critical evaluation of each of 91 studies was performed using PIOS criteria (Population, Independent variable, Outcome variable and Study design) (appendix 4.2). 32 studies have satisfied population criteria (studies including at least one of South and East Asia and Pacific countries), 29 studies have satisfied independent variable (i.e. economic governance), 18 studies have found to satisfy outcome variable (i.e. economic growth) and 32 studies satisfied study design. In total, 20 studies were found to satisfy all four criteria (appendices 4.3 and 4.4). Another 6 studies were added through hand search leaving a total of 26 studies for meta-regression analysis. The composition of published and unpublished studies is shown in appendix 4.5.

The general form of econometric models used in the above 26 studies with linear terms only (equation 4.1) and those with linear, non-linear and interaction terms (equation 4.2) appeared as follows.

$$Y_{it} = \alpha_0 + \alpha_1 F_{it} + \gamma X_{it} + \varepsilon_{it} \qquad \text{equation (4.1)}$$

$$Y_{it} = \alpha_0 + \alpha_1 F_{it} + \alpha_2 F_{it} \cdot K_{it} + \alpha_3 F^2_{it} + \gamma X_{it} + \varepsilon_{it} \qquad \text{equation (4.2)}$$

In equations 4.1 and 4.2, Y_{it} stands for dependent variable (economic growth); α_0 is the constant term and α_1 measures the marginal effect of F on Y; F stands for variable of interest i.e. various measures of governance; therefore, F_{it} measures the linear effect of measures of governance on economic growth; $F_{it} \cdot K_{it}$ is the interaction term which measures the effect of F on economic growth conditional on the value of K; F^2 is a non-linear term and α_3 measures the effect of F^2 on economic growth conditional on its own value. X_{it} is the vector of other variables that might affect the dependent variable; y measures the marginal effect of X_{it} on dependent variable; i and t are country and time indices respectively. E is the random error term.

The following data was obtained from the above studies:

- Information on Bibliography Bibliographical information of each study such as name of the first author, year of publication of study, type of study (published or unpublished), university of the first author was obtained.
- Study characteristics Study characteristics such as study type, study design, kind of data used, information on dependent and independent variables such as their functional form and data sources was obtained.

- Estimation methods used Data on estimation techniques such as ordinary least squares methods, panel data techniques, time series techniques, instrumental variable techniques and others were obtained.
- Outcome Data on outcome variable such as estimated parameters for all independent variables, t values, standard errors, p values, z values, F values of the estimates for linear, non-linear and interaction terms was obtained.

In order to allow for meaningful comparison across different models, partial correlation was used a standard measure. It is calculated using the formula $r = [t/\sqrt{t^2 + dof}]$, where, t stands for t – statistics of the multiple regression coefficient and dof stands for the degrees of freedom of the respective t –statistic. With the exception of the following variable, simple and multiple metaregression equations (1.3, 1.4 and 1.5) used in this study are similar to the ones shown in Chapter 2, page. 12.

4.3 LITERATURE REVIEW

4.3.1 THEORETICAL VIEWS ON GOVERNANCE AND GROWTH RELATIONSHIP

While the role of physical resources and human resources on economic growth cannot be undermined, institutions or economic governance plays an equally important role. Good governance in the form of rule of law, less political instability, low levels of corruption, necessary government effectiveness, high regulatory quality and appropriate levels of voice and accountability maximises economic incentives, reduces both information asymmetry and transaction costs. These contribute towards efficient allocation of resources and add to the smooth functioning of markets. This in turn encourages both domestic and foreign investors to invest further and also improves the confidence levels of existing investors. Overall, by building appropriate policies and laws governance builds all the necessary elements for the smooth functioning of markets and thereby contributes towards economic growth (Kaufmann et al, 1996; Busse and Groizard, 2008; Khamfula, 2007).

Theoretically, the relationship between economic governance and economic growth can be explained using North (1990)'s institutional framework. In view of this framework, institutions are important in shaping overall performance and growth of economies. Institutions in the form of political, economic and structural interactions are human-made constraints which aim to decrease the level of uncertainty and allow for firms and individuals to interact efficiently. Such an interaction can lead to effective and efficient allocation of resources that can add to economic growth. However, when these institutions function inefficiently it increases the transaction costs and hence discourages economic activities. In this context economic governance can be seen as an

institutional factor which can either have a progressive or regressive effect on economic growth (Dahlstrom and Johnson, 2007).

4.3.2 EMPIRICAL VIEWS ON GOVERNANCE AND GROWTH RELATIONSHIP

To date there has been a growing body of empirical literature that has examined the link between measures of governance and economic growth. These studies have provided continuous debate on the effects of various measures of governance and their impact on economic growth. While some studies have provided positive and significant effects of measures of governance, others have provided positive and insignificant, negative and significant, and negative and insignificant effects of such a relationship leading to overall inconclusiveness of results within this field.

Empirical evidence on effects of various measures of governance on economic growth in the case of South and East Asia and Pacific countries between 1980 and 2012 is provided by Adams and Mengistu (2008), Anwar and Cooray (2012), Butkiewicz and Yanikkaya (2004), Butkiewicz and Yanikkaya (2011), Campos and Nugent (1999), Evans and Rauch (1999), Evrensel (2010), Fernandez, Gonzalez and Suarez (2010), Haggard and Tiede (2011), Jalilian, Kirkpatrick and Parker (2007), Oliva and Rivera-Batiz (2002), Goldsmith (1995), Feeny (2005), Feeny and Mcgillivray (2010), Alonso (2010), Busse and Groizard (2008), Khamfula (2007), Mo (2001), Mauro (1995), Drury, Krieckhaus and Lusztig (2006), Assiotis and Sylwester (2012), Ekanayake and Chatrna (2010), Gani (2011), Seldadyo, Nugroho and Haan (2007), Commander and Nikoloski (2010), Klein (2005) and Law and Habibullah (2006).

Interestingly these studies have focused on either one or more governance measures such as voice and accountability, political instability, government effectiveness, corruption, regulatory quality, rule of law and have produced varied results. A brief and systematic summary of key aspects of the empirical studies are presented in appendix 10. It can be noted that differences in methodology, data sets, econometric methods and sample countries have produced mixed results. Inconclusiveness in empirical studies calls for a need for meta-regression analysis of these results in order to produce comparable, reliable and verifiable effect of measures of governance on economic growth (appendix 4.7).

4.4 DISCUSSION OF RESULTS

Empirical results are presented and analysed in this section. To start with, funnel plots and chronological order of estimates are used to offer a vivid picture on the state of empirical knowledge in governance growth studies. This is followed by simple and multiple meta-regression

results. An overview of measures of governance and growth meta-regression analysis is summarised in appendix 4.8.

4.4.1 FUNNEL PLOTS

FIGURE 4.2: FUNNEL PLOTS FOR MEASURES OF GOVERNANCE AND ECONOMIC GROWTH ESTIMATES







Estimates of various measures of governance and growth are plotted on the funnel plot shown in figures. Funnel plot traces the association between the effect size (partial correlation) and its precision (precision is calculated as inverse of standard error). I plot effect size on the X axis and precision on the Y axis. Estimates with high precision are normally few and are compactly distributed at the top of the funnel, while estimates with low precision are widely dispersed at the bottom of the funnel. Lack of consensus among estimates usually results in wide dispersion of the

estimates and vice versa indicating possible publication bias²⁰. Note that the reported estimates of law and aggregate governance and growth (except for corruption) are widely distributed around the central value of the funnel plot. While such a wide dispersion of values can arise due to real world factors, it can also be due to sampling error and due to differences in the research design (Doucouliagos and Ulubasoglu, 2008). In each of these graphs, the centre of the plot represents the estimated true underlying effect of respective measure on growth.

²⁰ I have tested for publication bias and its genuine effects using Funnel asymmetric test (FAT), Precision effect test (PET) (appendix 4.9). Except for voice and accountability measure, PET results suggests that despite the presence of publication bias, there is genuine effect of all measures of governance on FDI. I explore these aspects further in a different paper on publication bias in governance-growth studies.

4.4.2 CHRONOLOGICAL ORDER OF ESTIMATES

FIGURE 4.3: CHRONOLOGICAL ORDER OF ESTIMATES BASED ON AVERAGE YEAR OF SAMPLE PERIOD





I also plot chronological order of estimates reported from 1980 on various measures of governance against average year of sample period of each study. In case of law and corruption an upward trend can be seen in the estimates. While countries in South and East Asia & Pacific regions had governance well before the 1980's, an upward trend suggests that the effect of governance on growth started after the 1980's. In the case of voice and accountability and government effectiveness, I see fluctuations in estimates over the time period in focus.

4.4.3 SIMPLE META-REGRESSION ANALYSIS

The following table shows simple meta-regression results of various measures of governance on growth. Row1 shows unweighted estimates and row 2 shows weighted least squares estimates, weighted by precision. In the case of unweighted models, except for government effectiveness all measures show positive effect on growth. Hence, more of these measures is good for economic growth. Due to rescaling of governance measures, although corruption shows a positive sign, it should be interpreted inversely.

Negative effect of corruption indicates that less of this measure has growth enhancing effect. Nevertheless, these results should be interpreted carefully, both due to low R^2 values and fewer observations (especially for political stability and regulation measures). Another shortcoming of unweighted method is that it treats all observations with equal weight. This means studies reporting more than one observation can have an undue effect on the overall result.

	Law	Corruption	Aggregate
			Governance
	(Col.1)	(Col.2)	(Col.3)
Un weighted, β0	0.14	0.06	0.24
(Row1)	(5.37)	(2.83)	(2.73)
	R ² =0.13	R ² =0.04	R ² =0.03
Weighted by	0.07	-0.03	0.64
precision, β0	(3.51)	(-0.61)	(7.80)
(Row2)	R ² =0.13	R ² =0.04	R ² =0.14
Number of estimates	45	65	29

TABLE 4.1: SIMPLE META REGRESSION RESULTS

Note: Values in parenthesis right below the estimate represent t-values. Each column represents models run with all estimates of that measure of governance. Due to fewer number of observations, results for voice and accountability, political stability and regulation are shown in appendix 4.10.1.

In order to remove such undue effect, I use the weighted least squares model. Following Stanley and Doucouliagos (2012), weights are calculated as inverse of standard deviation. Once weights are applied, size and sign of a few measures of governance have changed. For instance cluster analysis results for corruption measures now show negative sign. However, R^2 values are still small, suggesting that these estimates are unreliable. One reason for lower R^2 could be due to the

fact that additional variables which can potentially show an effect on growth are not considered. Hence, I run multiple regression analysis including few moderator variables. These results are used to validate simple meta-regression results.

4.4.4 MULTIPLE META-REGRESSION ANALYSIS

I include the following moderator variables in multiple regression analysis. These variables are chosen as they are potentially important and some of which have been found to be significant in earlier meta-analysis studies (Doucouliagos & Paldam, 2008; Doucouliagos & Paldam, 2009; Doucouliagos & Ulubasoglu, 2008).

In terms of study related aspects I control for differences and whether or not a study is published, estimation techniques used by the studies, data related aspects such as the kind of data used (panel, time series and cross sectional data), whether or not studies report observations, and data sources for both governance and economic growth. In real world factors, I see regional effects by classifying the estimates into those belonging to South Asia, East Asia, South East Asia and others. I also use dummy variables for China and South Korea to see if inclusion of these variables makes any difference to results reported.

I wish to test if author related aspects such as the university of the first author. Based on this, I classify authors into American, European, South & East Asian and others. Journal related aspects such as journal discipline and ranking are controlled. Based on discipline, journals are classified into Economics & Finance, Business Management, Law, Science & Technology, Geography, Policy and Development. ABS 2010 journal rankings 1*, 2*, 3* and 4* are used to test if the future ranking of journal has any impact on results. While most of the study and real world related factors are proven to be important in earlier meta-regression studies, I merely wish to test the effects of journal and author related aspects. Main meta-regression results are presented in table 4.2 and the effect of moderator variables is shown in table 4.3.

	Law	Corruption	Aggregate
			Governance
	(Col.1)	(Col.2)	(Col.3)
Weighted by precision, $\beta 0$ (Row1)	0.18	-0.10	0.60
	(2.67)	(-0.36)	(7.40)
	Adj.R ² =0.38	Adj.R ² =0.86	Adj.R ² =0.92
Cluster analysis, β0	0.18	-0.10	0.60
(Row2)	(7.93)	(-0.26)	(7.98)
	R ² =0.42	R ² =0.88	R ² =0.93
Number of estimates	45	65	29

TABLE 4.2 MULTIPLE META-REGRESSION RESULTS

Note: Values in parenthesis right below the estimate represent t-values. Due to fewer number of observations, results for voice and accountability, political stability and regulation are shown in appendix 4.10.2.

The above table displays the results of multiple meta-regression analysis for each measure of governance. Row1 shows weighted least squares estimates and row 2 shows results of multiple regression analysis clustered by study. Under cluster analysis, each study is seen as a separate cluster and therefore the number of estimates of each study become the number of observations of each cluster (Doucouliagos and Ulubasoglu, 2008; Doucouliagos et al., 2010). Results in row 2 are used as a robustness check for the WLS results shown in row 1. It is important to note that all these results are retrieved after removing the effect of outliers²¹.

In case of voice and accountability, research literature has failed to provide evidence of genuine effect of it on growth. One would expect that either governance in the form of high levels of voice and accountability enhance economic growth as opposed to lower levels of voice and accountability or vice versa. These results are contrast to empirical results reported by Anwar and Cooray (2012), Campos and Nugent (1999) and Oliva and Rivera-Batiz (2002) who suggest that there is a ppositive effect of this measure on growth. These results are also different to the meta-regression results reported by Doucouliagos and Ulubasoglu (2008) for a broader group of countries in the world, which indicate that voice and accountability has no effect on economic growth.

²¹ Precision more than 40.

Surprisingly, political stability shows a negative and insignificant effect on economic growth. Hence, it can be deduced that political stability does not matter for economic growth. One would expect that stable and long term governments is good for growth. Positive effect can be realised when political stability removes uncertainty associated with uncertain political environment and increase investments along with the pace of economic growth. Despite this, there might be two possibilities where such a stability does not have such a positive effect. Firstly, if the political stability is achieved through oppression, it might show a negative effect on growth. Secondly, when political stability precludes any form of change and leads to some sort of stagnation that does not allow competition in economic activity. While this study does not examine the reasons for such insignificant effect, it is worthy of future research to focus on these aspects. These results must be interpreted with caution for two reasons. Firstly, because of fewer number of observations and secondly, as these results are not confirmed by cluster analysis. These are presented in appendix 4.10.2.

Government effectiveness shows a negative, robust and statistically significant effect on growth. These findings are in contrast with those reported by Jalilian et al. (2007). It is worth noting that government effectiveness measures have aggregated all studies that have looked at the effect of measures of governance such as quality of civil and public service, existence of red tape, quality of policy formulation and implementation and government's credibility to its stated policies. As one would expect that countries with better government effectiveness achieve high growth rates through high credit ratings and investments, it is surprising to see such negative effects. One possible reason for this can be due to fewer number of observations (appendix 4.10.2). This leaves scope for future research to examine those variables of government effectiveness that show a negative effect on economic growth and to differentiate them from those that cause negative effect.

Regulation is found to show a positive, robust and statistically significant effect on economic growth (appendix 4.10.2). In comparison to less regulated countries, my results indicate that highly regulated economies witness higher levels of economic growth. While these results are in line with institutionalist's view that tighter regulation promotes economic growth, it does not compare with other studies such as those by Goldsmith (1995) and Gani (2007). As most of the countries in this region have deregulated their economies post 1980's with the aim of removing the regulatory burden as well as to promote their economies (Jalialian et al., 2007), these results are not surprising. Similar to government effectiveness, it is important to note that empirical studies included under this measure have focused on various forms of regulations such as accounting and environmental among many others. It could be possible that any one form of the above regulations

is showing positive effect on growth and not others. This study only offers an overview on the effect of regulation on growth and does not focus on individual forms of it.

In contrast with the results reported by Ugur and Dasgupta (2011), Campos et al., (2010) and Mauro (1995), Butkiewicz and Yanikkaya (2004), Evrensel (2010), Drury et al., (2006) and Gani (2011), this study finds positive, robust and statistically insignificant effect of corruption on economic growth. Although negative sign on this measure should be read as corruption, having growth retarding effects²², this measure is not important for growth. This contradicts widely held views that corruption is either detrimental (Drury et al., 2006). Results of this study also contradict the view that corruption aids economic growth by 'greasing the wheels' of economic activity.

Although, the present study does not consider various forms and types of corruption, it is worth noting that certain forms of corruption are considered to be important and acceptable from growth point of view, while others are not (Leff, 1964). Hence, it is possible that the overall insignificant effect of corruption is caused due to more of certain forms of it than others. For instance, if bribes are paid to government officials to overcome bureaucratic delays in starting business and any inefficient rules associated with them, then allowing such corruption will facilitate investments and have a beneficial effect on economic activity and nullify the overall effect of it (Leff, 1964; Huntington, 1968 and Lui, 1985 are good papers to read on how corruption enhances economic growth). Overall, while corruption is unimportant from growth point of view, in order to suggest appropriate policy intervention within this area, further research is strongly advised especially to segregate the effects of different forms of corruption on growth.

One would either expect law to promote economic growth through various routes such as protection of property rights, institutional checks on government or by mitigating violence; (Haggard and Tiede, 2011 is a good paper to read more on this) or to have detrimental effects on growth through tighter laws. My study reveals that law is important for economic growth. The significant effect of law on growth is in line with positive and significant effects reported by Butkiewicz and Yanikayya (2004), Butkiewicz and Yanikayya (2011) (in case of developing countries), Fernandez et al., (2010), Haggard and Tiede (2011) and Busse and Groizard (2008). Nevertheless, these results are in contrast to those reported by Campos and Nugent (1999) and Oliva-Rivera-Batiz (2002). However, R² value for this measure is 0.38 suggesting that the model is poorly fit. Hence, the results should be interpreted carefully.

 $^{^{22}}$ Governance measures are rescaled as 0 – 1 high to low corruption. Hence, positive sign should be read as corruption having negative effect.

As expected, overall governance also shows a positive and significant effect on growth. The positive effect of aggregate governance on growth was expected as most of the individual measures of governance showed positive effect (except for corruption measure). As good governance is expected to be a prerequisite for economic growth by providing a favourable climate for investments and other economic activities, these results are not surprising (Globerman and Shapiro, 2002a). Hence, it can be inferred that, governance in these regions is serving as a helping hand for growth through less cumbersome and friendly rules and regulations, and by appropriate enforcement of law. These results are in line with the Dunning's OLI paradigm and institutionalist's view that institutions need not necessarily deter economic growth, they can instead aid it.

All the above results must be read and interpreted carefully by duly taking into account the following points. Firstly, results in case of law, corruption and aggregate governance are robust to cluster analysis. Secondly, while results on some measures of governance can be a bit surprising and unexpected, it is important to note that I do not have sufficient region specific estimates to see if these results are more specific to one region than others (i.e. East Asia or South East Asia or South Asia). Thirdly, I do not have sufficient estimates controlling for endogeneity to check if the results show genuine effect of governance measures on growth or whether the effect is due to causality. Finally, while my results are reliable with high R^2 value and econometrically sufficient observations, in the case of corruption, law and aggregate governance, they are less reliable for voice and accountability, regulation and political stability due to fewer observations (less than 30).

4.4.5 MODERATOR VARIABLES ANALYSIS

I now turn my attention towards exploring the factors that have caused heterogeneity in reported results of governance and growth studies. While I have included many study, real world, author and journal related factors, only few study, author and journal related factors proved to be important.

Many study related factors have proven to make a significant difference to reported results. I find that governance effect varied based on the data type used. My notion that governance takes time to show its effect on growth did not prove to be right. Studies including population related variable have reported higher effects of corruption by 0.20 on growth compared to an overall effect of - 0.10. Those including domestic investment have reported higher effects of aggregate governance on growth. I infer from these results that omitted variable bias does matter for governance and growth studies. While, governance and growth sources did matter, differences in defining FDI and growth did not make any difference to the stated results.

My study provides evidence that real world related factors did matter for governance growth studies. Studies including China in their list of countries have reported an average effect of 0.01 which is lower by 0.17 than overall affect. Similarly, models including South Korea in their list of sample countries conveyed bigger effects of law on growth by 0.11 than the overall effect of 0.18. Author and journal characteristics did make a difference. For example, American authors have emphasised less on corruption measure compared to other authors by -0.26 than an overall average effect of -0.10. As I expected, the rest of the author and journal related aspects such as authors from 'best' universities and journal ranking did not matter.

TABLE 4.3: EFFECT OF MODERATOR VARIABLES

Moderator variables	Law		Corruption		Aggregate	
					Governance	
If the growth data for the estimate is taken from World Bank Source.			0.24	0.24		
Reference: If the growth data for the estimate is taken from Non			(1.74)	(1.29)		
World Bank source.						
If the estimate is includes China in its list of sample countries.	-0.17	-0.17	-0.32	-0.32		
Reference: If the estimate does not include China in its list of sample	(-2.93)	(-9.30)	(-2.76)	(-2.11)		
countries.						
If the estimate is includes South Korea in its list of sample countries.	0.11	0.11	0.28	0.28	0.18	0.18
Reference: If the estimate does not include South Korea in its list of	(3.54)	(28.73)	(3.32)	(2.61)	(3.31)	(4.01)
sample countries.						
If the estimate is published in Economics and Finance Journal.					-0.46	-0.46
Reference: Estimate published in Science and Technology journals.					(-15.72)	(-93.74)
If the estimate is estimated by an American author. Reference:			-0.26	-0.26		
Estimate estimated by other author.			(-4.84)	(-3.79)		
If the estimate belongs to a model that includes population related			0.20	0.20		
variable. Reference: If the estimate belongs to a model that does not			(2.21)	(1.61)		
include population related variable.						

If the estimate belongs to a model that includes domestic investment						0.4	0.4
related variable. Reference: If the estimate belongs to a model that						(10.19)	(14.82)
does not include domestic investment related variable.							
If the governance data for the estimate is taken from Polity database.			-0.2	.5	-0.25		
Reference: If the governance data for the estimate is taken from			(-3.2	(-	-3.06)		
World Governance Indicators database. Reference:							
No. of observations	45	45	65	65		29	29
Adj R^2/R^2	0.38	0.42	0.86	0.88		0.92	0.93

Note: Only variables that have a significant effect are shown. Values in parenthesis show t-values.

See appendix 4.6 for full descriptive statistics of moderator variables included in multiple meta-regression.

4.5 CONCLUSIONS

This paper has meta-synthesised the empirical evidence on various measures of governance and economic growth in South and East Asia Pacific countries based on 29 studies with 554 estimates from 1980 – 2012. The empirical results show that while law is positively and significantly correlated to growth, corruption has insignificant effect on growth. In case of voice and accountability, research literature has failed to provide evidence of genuine effect of it on growth. Finally, overall governance is important for growth (see appendix 9 for summary of results). Various study, real world, estimation and author related aspects proved to have made difference to the stated results.

The main limitations of this study are as follows. Firstly, this study has focused only on South and East Asia and Pacific countries from 1980 – 2012. As the results of this study are confined to empirical results on measures of governance on growth during this period, they represent the research at one point in time and cannot be used as a forecasting tool. Another possible caveat of the research is that I have only focused on calculating the direct effects of measures of governance on economic growth. This study did not analyse the indirect effects of measures of governance on economic growth through their interaction with other physical and macro environment factors. To a large extent this has been due to the limited number of interaction and nonlinear terms of measures of governance. As an example, there were only 9 estimates of government effectiveness through regulation. Finally I would like to comment on the matter of the type of empirical studies included in this study. One of the main criteria in including a study has been that the measure of governance in the primary studies is expressed as a scale and not as a number (i.e. number of assassinations, number of riots amongst many others).

Few aspects of this study that require further research are identified. First, empirical studies on measures of governance and economic growth are relatively few in the case of South and East Asia and Pacific countries as opposed to studies on other determinants of growth. While one reason for this could be the unavailability of data in the past, recent years have seen a surge in data sources. More specifically, World Bank's project on worldwide governance indicators provides governance data on different measures of governance for 212 countries from 1996 onwards. Future research can make use of this data and conduct further research. In addition to this, as there is a possibility of reverse causality between measures of governance and growth, there is a need for controlling this aspect as well.

Secondly, most of the governance indicators used by the primary studies have used people's perceptions of governance in various countries derived from polls, surveys or expert opinions

(with the exception of Busse and Groizard, (2008) who uses objective data on regulation from the Doing Business database provided by World Bank (2006)). These measures are predominantly taken from sources such as Polity data set, ICRG and others. Such perception based measures are subjective and lack objective analysis of governance in addition to leading to a large margin of error (Gani, 2011). Hence, future researchers can use more reliable and objective data on institutions to measure their effect on economic growth or find weighted measures of governance by combining perception based data on measures of governance with that of objective data (Ugur and Dasgupta, 2011).

Thirdly, an important issue for future research concerns the indirect effects of measures of governance. Governance measures can transmit indirect effects on growth through factors such as human capital, physical capital amongst many other factors. My systematic search for empirical studies has found 12 out of 26 studies measuring such indirect effects. Governance measures are interacted with factors like domestic credit, private credit, capital account, money supply, bank market concentration amongst others. Due to the diverse nature of interaction terms and the limited number of observations under each category, I did not include them in my meta-regression analysis. This shows a clear scope for inclusion of indirect effects of governance on growth by future studies. The final important area where additional research is required is on the use of time series data. Authors of primary studies have mainly focused on panel studies (except Feeny (2005)). While panel studies help in getting more robust and econometrically efficient results, country specific studies will help in exploring country specific effects of various measures of governance on growth.

Based on the results of this study, I am convinced that without establishment and maintenance of economic governance in an appropriate manner, achieving economic growth might be difficult. My results have important policy implications. As a preface, it is important to point out that while any attempts by governments to enhance economic growth must focus on all measures of economic governance, some measures should be tighter than others. Policy implications of the results for South and East Asia & Pacific countries are that they can enhance their economic growth by improving governance, particularly by brining improvements law and overall governance. While corruption is found to have insignificant effect on the growth, my study does not suggest exact channels or forms through which it does or does not effect economic growth.

CHAPTER 5 CONCLUSION CHAPTER

5.1 INTRODUCTION

The relationship between governance, FDI and growth has been the subject of several theoretical and empirical studies. Good governance in host countries is able to provide a positive climate that encourages FDI and growth. However, the research findings in this field of study are quite conflicting and do not support informed policy actions. In this thesis, I made an attempt to contribute to both theory and practice by meta-synthesising the empirical estimates about the interaction of governance, FDI and growth in the case of South and East Asia and Pacific countries from 1980 – 2012. To achieve this aim, the thesis was structured to include three meta-regression analyses. The main conclusions of this thesis are that reforming institutions and making them favourable towards investors is a way forward to attracting FDI and achieving sustained economic growth to improve standards of living of people in South and East Asia & Pacific countries.

5.2 FINDINGS

The main findings of this thesis are as follows:

1. The main findings of the first meta-regression analysis in chapter 2 are that measures of governance such as political stability, regulation, law, government effectiveness and aggregate governance have important effects on FDI. As expected political stability, government effectiveness, regulation and aggregate governance are positively associated with FDI. While corruption is found to have no effect on FDI, Law is negatively associated. My results with respect to political stability and corruption are more reliable compared to other measures of governance, both due to the large number of observations and higher R² value.

In terms of heterogeneity, various study, real world, author and journal related factors have shown to cause an important difference in the reported results. One interesting result was that European authors have reported different results compared to other authors. They have placed less emphasis on aspects such as government effectiveness and regulation, and more on voice and accountability. American authors on the other hand placed less emphasis on political stability. As expected, regional factors such as inclusion of China and South Korea did matter.

2. FDI is shown to have a positive effect on growth in the case of all estimates, estimates controlling for endogeneity, and South East Asia. While the positive effect of all

estimates is confirmed by estimates controlling for endogeneity, I did have estimates controlling for endogeneity for other models to confirm their effect. Hence it is possible that the effect shown in the case of other models can be due to reverse causality effect. It is important to note that the findings for East Asia and South Asia are less reliable as the number of observations is fewer than 30 and hence these results are presented in appendix.

In terms of study related factors, type of FDI, data types, estimation techniques matter for the reported results. Models estimated using OLS and times series techniques reported higher effects in case of model with all estimates compared to those estimated using other techniques. In case of South East Asia, models estimated by OLS have reported lower effects of FDI on growth. The magnitude of effect also differed among studies based on real world factors. In case of all estimates model, while studies including China have reported lower effects, those including South Korea have reported higher effects. Author and journal related factors have shown noticeable effects on reported results. American and European authors have reported lower effects as compared to other authors.

3. The empirical results show that while law and aggregate governance are positively and significantly correlated to growth, corruption has insignificant effect. In case of voice and accountability, research literature has failed to provide evidence of genuine effect of it on growth. Finally, overall governance is important for growth.

Many study related factors have proven to make a significant difference to reported results. Studies including population related variable have reported higher effects of corruption and those including domestic investment have reported higher effects of aggregate governance on growth. I infer from these results that omitted variable bias does matter for governance and growth studies. My study provides evidence that real world related factors did matter for governance growth studies. Models including South Korea in their list of sample countries conveyed bigger effects of overall governance on growth. Author and journal characteristics did make a difference in governance and growth studies. American authors less on corruption measures compared to other authors.

5.3 CONTRIBUTION TO THEORY

The main contributions of this thesis are summarised below:

Findings of this thesis can be viewed as significant academic contributions to inconclusiveness of empirical results in the field of governance, FDI and growth. One significant and common contribution of chapters 2, 3 and 4 is systematic review of literature of all empirical studies on the effects of governance on FDI, FDI on growth and governance on growth respectively. Such review although it does not protect against publication bias, due to the well-defined methodology, it offers an unbiased view on literature in the respective fields of study. For all three studies I have predefined search strategies that have helped in detecting much of the relevant literature but these strategies are also documented so that future studies can assess the rigour and completeness of each one of the studies along with replicating them if necessary.

The second contribution of this thesis is that the findings of chapter 2 (governance and FDI) which have reduced the inconclusiveness of the empirical evidence on the role of governance on FDI based on 771 estimates from 48 studies. As expected political stability, government effectiveness, regulation and aggregate governance are positively associated with FDI. While corruption is found to have no effect on FDI, Law is negatively associated. My results with respect to political stability and corruption are more reliable compared to other measures of governance, both due to the large number of observations and higher R² value. Overall, main message of this chapter is that aggregate governance does matter for FDI and has a positive effect on it.

The third contribution of this thesis is that it has summarised 633 estimates of FDI and growth from 37 studies and has reduced the inconclusiveness in the field of FDI and growth. The main result of this chapter is that it shows that the effect of FDI on overall economic growth of this region is positive and statistically significant. Meta-regression results of this chapter confirm the view that the effect of FDI varies according to region. FDI has shown to have positive and significant effect in the case of all estimates, estimates controlling for endogeneity and South East Asia. Although FDI showed negative and significant effect in case of East Asia and South Asia, I did not have sufficient observations to reach firm conclusions.

The fourth significant contribution is that it has meta-synthesised the effect of measures of governance on growth using 554 estimates from 29 studies. Such an effect has reduced the inconclusiveness of empirical results within the fields of governance and growth. The main message that comes from this chapter is that different measures of governance have different

effects on economic growth. While corruption is found to have no effect on growth, the other measures of governance such as law and aggregate governance have positive effect on growth. Surprisingly, in case of voice and accountability, research literature has failed to provide evidence of genuine effect of it on growth. Finally, overall governance is important for growth These findings suggests that the effect of governance on growth is more complex than it looks in the first case and it also raises the question of whether corruption and voice and accountability measures used in primary studies actually measure what they have to measure since the results are unexpected for these two measures.

The last and most important contribution of this thesis is that it has identified various studies, real world, author and journal related factors that have caused significant difference to reported estimates within the fields of governance, FDI and growth. To my knowledge, this thesis is the first study which has highlighted that authors from different universities have an important effect on reported results within this field. For instance, throughout three studies I have highlighted that American and European authors consistently reported different effects of measures of governance on FDI, FDI on growth and measures of governance on growth compared to other authors. I also find that journals from different disciplines report different effects of governance on FDI and growth, and FDI on growth.

5.4 LIMITATIONS

It is important to note that despite the importance of systematic literature review and metaregression analysis, my thesis has not addressed the following issues for several reasons. Hence, the results of this thesis should be evaluated carefully after taking on board the following caveats. First of all, for the lack of sufficient data and as well as to ensure consistence of measures of governance, I have included only those empirical studies that have defined measures of governance as an index on a scale rather than as a number. For instance, if a study has defined governance as the number of riots or assassinations, or the amount of fines levied for breaking the law, then such studies are not included in this study. This leaves scope for future researchers to extend this study by including all such studies.

Secondly, while informal governance or institutions such as a country's culture can also have an important bearing on FDI and economic growth, it is worth noting that this thesis has only looked at the role of formal governance. Hence, future research might want to consider informal governance alongside formal governance to study its effect on both FDI and economic growth. Thirdly, publication bias is one of the important aspects of meta-regression analysis. As the main

focus of this thesis has been to find the overall effect of governance on FDI and growth and FDI on growth, I did not emphasise publication bias in this thesis.

The fourth caveat of this thesis is that it has only looked at the linear relationship between governance, FDI and growth. While this limitation is common to many other meta-regression studies, non-linear and interaction estimates were not estimated as the number of observations was quite limited. Fifth, despite including various study related, author related, journal related and real world factors for testing for heterogeneity, the possibility of unexplained residual heterogeneity cannot be ruled out. Sixth, in the three meta-regression studies, a dummy variable has not been used to test for structural break. Use of dummy variable would have been particularly useful as FDI inflows surged from 1980's. Exclusion of this can affect the reliability of the model in general. The last but most important caveat of this thesis is that my study only offers a general overview on the effect of measures of governance on growth. This caveat means that the study does not look into specific forms of each measure of governance. For instance, regulation can take many forms such as labour market regulations, environmental regulations, planning regulations among many others. Similarly corruption is a multifaceted phenomenon and its sub elements can have different impacts on FDI and growth (Teksoz, 2004).

5.5 RESEARCH IMPLICATIONS

First and foremost, in depth analysis of literature in the field of governance, FDI and growth using systematic search reveals that studies so far have explored the empirical relationship between governance, FDI and growth by using various proxies for all three terms on more or less the same data sources using similar econometric techniques. Despite this, empirical results are quite conflicting in this field of study suggesting that further research is clearly desirable in exploring the causes along with the reasons of such different results. Secondly, most of the studies in this field have used panel data sets to study the relationships between the three variables. While panel studies offer econometrically efficient results by blending the inter-individual differences and dynamics, they do not offer richer insights on how the relationship works in a specific country. Hence, future studies might derive more mileage by focusing on country specific studies as they are limited in number and also because they offer a richer picture of the effects of governance and investments on growth.

Thirdly, this thesis has looked at the one way meta-relationship between governance, investments and growth in chapters 2, 3 and 4. Future research might consider systematic literature review and meta-regression analysis by reversing the above relationships in order to address the overall effects of growth on FDI, FDI on governance and growth on governance. In addition to this, it might also

be useful to look at the micro level evidence of FDI so as to unearth the private returns and localised effects of FDI on growth. In the fourth instance, in comparison with samples of the first two meta-regression analyses, the sample size is relatively less in the third one. This leaves scope for future research to pay attention to this area of research. In the fifth instance, it might also be interesting to combine empirical evidence on the three relationships and explore it by use of Structural Equations Model (SEM). Whether the results reported in this thesis would significantly differ by use of SEM is uncertain. Last and most importantly, there is a rapid expansion of research publications and an increase in empirical studies in the fields of both Economics and International Business studies. This clearly leaves scope for future studies in using systematic literature review and meta-regression analysis so as to reduce the inconclusiveness of results and offer a general picture of a wide range of research findings (Stanley and Doucouliagos, 2012).

5.6 POLICY IMPLICATIONS

By employing objective standards, critical methodology and no preconceived notions or theories, this thesis has systematically synthesised the research findings in the fields of governance, growth and investments. After studying the relationship between governance, FDI and growth, the key findings emerged from the three meta-regression analyses in this thesis suggest the following policy implications. As a preface to these policy implications it is important to note two points. Firstly, as mentioned in the limitations section of this thesis, I did not study the effects of specific forms of each measure of governance. Hence the following policy implications might be different based on the effects of specific forms of governance. Secondly, it is necessary to take the next step forward for all three meta-regression analyses to delve deeper into the three relationships and the respective outcomes of such relationships before any policy intervention is made. Thirdly, these policy implications are subjective to the kind of institutional quality in each of South and East Asia & Pacific countries.

Firstly, South East Asian countries should continue to attract FDI as it has proved to have growth enhancing effects. A favourable economic environment that helps to reap the benefits of FDI for growth is advised in case of East Asian countries. As most of the countries in this region already have FDI policies in place, it is worth focusing on appropriate policy enforcement so as to realise the positive effect of FDI on economic growth.

Secondly, government effectiveness is shown to have a positive effect on FDI. Hence, there is a prima facie need for continuing and improving the quality of public services and insulating them from any political pressure. Process and productiveness of investments and economic activity can be slow and discouraging when there are any cumbersome rules or dishonest bureaucrats. A

tentative explanation for such a positive effect can also be due to the presence of friendly rules and bureaucracy and hence they should be continued. In the third instance, the expected positive effect of regulation on both FDI suggests that policy makers must continue to enforce regulations in order to maximise such a positive effect. In the fourth instance, policy implications of negative effect of law on FDI suggest that certain legal reforms aimed at reducing such a negative effect on FDI are necessary. In the fifth instance, corruption is shown to have no significant effect on both FDI and economic growth and hence both from FDI and growth point of view, no further policy intervention is advised for this measure of governance.

In the sixthth case, as aggregate governance is found to have a positive effect on both FDI and growth, country specific reforms aimed at improving governance are suggested. As Kaufmann et al (1999) note, good governance is important for economic growth of nations. Improving governance is critical in these countries in order to attract further FDI and to achieve economic growth. Designing and implementing policies that help in improving transparency and accountability, enforcing regulations appropriately and improving legal systems can be seen as a way forward towards furthering such positive effects of governance on both FDI and economic growth. Overall, countries in South and East Asia & Pacific region striving to attract FDI and increasing their economic growth levels need to design and implement governance quality in a way that it facilitates further investments and economic activity rather than constraining them both.

REFERENCES

- 1999. Indian Business and Foreign Investment. Economic and Political Weekly, 34, 2764-2765.
- 2000. FDI Blues. Economic and Political Weekly, 35, 1587-1588.
- 2006. The Impact of Regulatory Governance and Privatization on Electricity Industry Generation Capacity in Developing Economies. *World Bank Economic Review*, 20, 115-141.
- 2010. Foreign Investment: Impact on China's Economy. *Journal of Corporate Accounting and Finance*, 21, 25-41.
- APRIL .2003. Foreign Direct Investment in Developing Asia: Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations | *Asian Development Bank*.
- DECEMBER .2002. Technological Spillovers from Foreign Direct Investment--A Survey | Asian Development Bank.
- NOVEMBER 2008. Foreign Direct Investment, Innovation, and Exports: Firm-Level Evidence from People's Republic of China, Thailand, and Philippines | *Asian Development Bank*.
- ABBOTT, P., BENTZEN, J. & TARP, F. 2009. Trade and Development: Lessons from Vietnam's Past Trade Agreements. *World Development*, 37, 341-353.
- ABDOU, A. & MOSHIRI, S. 2009. Privatization and Capital Formation in Developing Countries: An Empirical Analysis. *International Review of Applied Economics*, 23, 557-575.
- ABRAHAM, F., KONINGS, J. & SLOOTMAEKERS, V. 2010. FDI Spillovers in the Chinese Manufacturing Sector. *Economics of Transition*, 18, 143-182.
- ABREU, M. DE GROOT, H.L.F, FLORAX, R.J.G.M. 2005. A Meta-Analysis of Beta-Convergence: The Legendary Two-Percent. *Journal of Economic Surveys*, 19 (3), 389 – 420.
- ACHARYYA, J. 2009. FDI, Growth and the Environment: Evidence from India on CO2 Emission During the Last Two Decades. *Journal of Economic Development*, 34, 43-58.
- ADAM, A. & FILIPPAIOS, F. 2007. Foreign Direct Investment and Civil Liberties: A New Perspective. *European Journal of Political Economy*, 23, 1038-1053.
- ADAMS, S. & MENGISTU, B. 2008. Privatization, Governance and Economic Development in Developing Countries. *Journal of Developing Societies (Netherlands)*, 24, 415-438.
- ADDISON, T. & HESHMATI, A. 2003. The New Global Determinants of FDI Flows to Developing Countries: The Importance of ICT and Democratization. WIDER Discussion Paper, 45.
- ADEOYE, A. 2009. Macro-Economic Level Corporate Governance and FDI in Emerging Markets: Is there a Close Relationship? *Available at SSRN* 1120816.
- AGARWAL, M. & SAMANTA, S. 2006. Structural Adjustment, Governance, Economic

Growth and Social Progress. *Journal of International Trade & Economic Development*, 15, 387-401.

- AGATIELLO, O. R. 2007. Is South-South Trade the Answer to Alleviating Poverty? *Management Decision*, 45, 1252-1269.
- AGGARWAL, M. R. 1980. Impact of Foreign Capital Inflows and Inflation on Domestic Savings in Developing Countries with Special Reference to India. *Indian Journal of Economics*, 61, 183-189.
- AGOSIN, M. & MACHADO, R. 2005. Foreign Investment in Developing Countries: Does it Crowd-In Domestic Investment? *Oxford Development Studies*, 33, 149-163.
- AHLQUIST, J. & PRAKASH, A. 2008. The Influence of Foreign Direct Investment on Contracting Confidence in Developing Countries. *Regulation and Governance*, 2, 316-340.
- AHLQUIST, J. S. & PRAKASH, A. 2010. FDI and the Costs of Contract Enforcement in Developing Countries. *Policy Sciences*, 43, 181-200.
- AHMAD, E. & HAMDANI, A. 2003. The Role of Foreign Direct Investment in Economic Growth. *Pakistan Economic and Social Review*, XLI, 29-43.
- AHMAD, M. H., ALAM, S., BUTT, M. S. & HAROON, Y. 2003. Foreign Direct Investment, Exports, and Domestic Output in Pakistan. *Pakistan Development Review*, 42, 715-723.
- AHMAD, N. & GHANI, E. 2005. Governance, Globalisation, and Human Development in Pakistan. *Pakistan Development Review*, 44, 585-596.
- AHMAD, N. A. & IMAN, A. H. M. 2011. The Relative Performance of ASEAN Countries in Foreign Direct Investment in Real Estate. In: TAO, F. (ed.) Sociality and Economics Development.
- AHMAD, S. 1990. Foreign Capital Inflow and Economic Growth A Two Gap Model for the Bangladesh Economy. *Bangladesh Development Studies*, XV111, 55-79.
- AHMED, E. M. 2010. The Role of FDI Intensity in Achieving Productivity Driven Growth in Malaysian Economy. *Applied Econometrics and International Development*, 10, 197-210.
- AHMED, E. M. 2012. Are the FDI Inflow Spillover Effects on Malaysia's Economic Growth Input Driven? *Economic Modelling*, 29.
- AHMED, N. 2003. Trade Liberalization and Endogenous Growth of Manufacturing Industries in Bangladesh: An Empirical Investigation. *Applied Economics*, 35, 305.
- AHN, Y. S., ADJI, S.S. & WILLETT, T. D. 1998. The Effects of Inflation and Exchange Rate Policies on Direct Investment to Developing Countries. *International Economic Journal*, 12, 1, 95 – 104.

- AHRENS, J. 1998. Economic Development, the State, and the Importance of Governance in East Asia. *Economic Systems*, 22, 23-51.
- AISEN, A. & VEIGA, F.J. 2011. How Does Political Instability Affect Economic Growth? *IMF Working Paper*. WP/11/12.
- AKISIK, O. & PFEIFFER, R. 2009. Globalization, US Foreign Investments and Accounting Standards. *Review of Accounting & Finance*, 8, 5-37.
- ALFARO, L., CHANDA, A., KALEMLI-OZCAN, S. & SAYEK, S. 2004. FDI and Economic Growth: The Role of Local Financial Markets. *Journal of International Economics*, 64, 89-113.
- ALFARO, L., CHANDA, A., KALEMLI-OZCAN, S. & SAYEK, S. 2010. Does Foreign Direct Investment Promote Growth? Exploring the Role of Financial Markets on Linkages. *Journal of Development Economics*, 91, 242-256.
- ALFARO, L. & CHARLTON, A. 2009. Intra-Industry Foreign Direct Investment. *The American Economic Review*, 99, 2096-2119.
- ALGUACIL, M., CUADROS, A. & ORTS, V. 2011. Inward FDI and Growth: The Role of Macroeconomic and Institutional Environment. *Journal of Policy Modeling*, 33, 481-496.
- ALI, F. A., FIESS, N. & MACDONALD, R. 2010. Do Institutions Matter for Foreign Direct Investment? *Open Economies Review*, 21.
- ALLEN, F., JUN, Q. & MEIJUN, Q. 2005. Law, Finance, and Economic Growth in China. *Journal of Financial Economics*, 77, 57-116.
- ALONSO, J. A. 2010. Colonisation, Institutions and Development: New Evidence. *Journal of Development Studies*, 47, 7, 937 958.
- AMARO, A. & MILES, W. 2006. Racing to the Bottom for FDI? The Changing Role of Labour Costs and Infrastructure. *The Journal of Developing Areas*, 40, 1-13.
- ANDREWS, M. 2010. Good Government Means Different Things in Different Countries. *Governance*, 23, 7-35.
- ANG, J. 2009. Foreign Direct Investment and its Impact on the Thai Economy: The Role of Financial Development. *Journal of Economics & Finance*, 33, 316-323.
- ANGHEL, B. 2004. Do Institutions Affect Foreign Direct Investment? International Doctorate in Economic Analysis, Universidad Autonoma De Barcelona.
- ANTRÀS, P., DESAI, M. A. & FOLEY, C. F. 2009. Multinational Firms, FDI Flows, and Imperfect Capital Markets. *The Quarterly Journal of Economics*, 124, 1171-1219.
- ANWAR, M. & AMAN, S. 2010. Aid Effectiveness in the Education Sector of Pakistan. *Pakistan Journal of Social Sciences*, 30, 355-372.
- ANWAR, S. & COORAY, A. 2012. Financial Development, Political Rights, Civil Liberties and

Economic Growth: Evidence from South Asia. Economic Modelling, 29, 974-981.

- ANWAR, S. & LAN PHI, N. 2010. Absorptive Capacity, Foreign Direct Investment-Linked Spillovers and Economic Growth in Vietnam. *Asian Business & Management*, 9.
- ANWAR, S. & NGUYEN, L. 2011. Financial Development and Economic Growth in Vietnam. Journal of Economics & Finance, 35, 348-360.
- ANWAR, S. & NGUYEN, L. P. 2010. Foreign Direct Investment and Economic Growth in Vietnam. *Asia Pacific Business Review*, 16, 183-202.
- APRIL, Y. 2011. Understanding an Aspect of China's Governance and Economic Growth. *Africa Insight*, 41, 106-122.
- ARBATLI, E. 2011. Economic Policies and FDI Inflows to Emerging Market Economies, USA, '*IMF eLibrary*'.
- ASIA, E. 1999. Capital Flows to East Asia. International Capital Flows.
- ASSIOTIS, A. & SYLWESTER, K. 2012. Do the Effects of Corruption upon Growth Differ Between Democracies and Autocracies? *University of Cyprus Working Papers in Economics*.
- ASTERIOU, D. & PRICE, S. 2005. "Uncertainty, Investment and Economic Growth: Evidence from a Dynamic Panel". *Review of Development Economics*, 9, 277-288.
- ATHUKORALA, P. 1995. Foreign Direct Investment and Manufacturing for Export in a New Exporting Country The Case of Srilanka. *World Economy*, 18.
- ATHUKORALA, P. 2002. Foreign Direct Investment and Manufacturing Exports: Opportunities and Strategies.
- ATHUKORALA, P. & MENON, J. 1995. Developing with Foreign Investment: Malaysia. *Australian Economic Review*, 9.
- ATHUKORALA, P.-C. & RAJAPATIRANA, S. 2003. Capital Inflows and the Real Exchange Rate: A Comparative Study of Asia and Latin America. *World Economy*, 26, 613-637.
- ATHUKORALA, P.-C. & TRAN QUANG, T. 2012. Foreign Direct Investment in Industrial Transition: The Experience of Vietnam. *Journal of the Asia Pacific Economy*, 17.
- AWASTHI, R. K. 1984. Evolution and Development of Urban-Society and Municipal Governance in India. *Man in India*, 64, 263-279.
- AYHAN KOSE, M., PRASAD, E. S. & TERRONES, M. E. 2009. Does Openness to International Financial Flows Raise Productivity Growth? *Journal of International Money and Finance*, 28, 554-580.
- AZAM, M., KHAN, H., HUNJRA, A. I., AHMAD, H. M. & CHANI, M. I. 2011. Institutional, Macro Economic Policy Factors and Foreign Direct Investment: South Asian Countries Case. African Journal of Business Management, 5, 4306-4313.

- AZHAR, M. 2011. New Exploration Licensing Policy (NELP) in India. *OPEC Energy Review*, 35, 174-188.
- AZID, T., KHALIQ, N., JAMIL, M. & KAZMI, A. A. 2006. Sectoral Volatility, Development, and Governance: A Case Study of Pakistan. *Pakistan Development Review*, 45, 797-818.
- AZMAN-SAINI, W. N. W., BAHARUMSHAH, A. Z. & LAW, S. H. 2010. Foreign Direct Investment, Economic Freedom and Economic Growth: International Evidence. *Economic Modelling*, 27, 1079-1090.
- AZÉMAR, C. & DESBORDES, R. 2010. Short Run Strategies for Attracting Foreign Direct Investment. *World Economy*, 33, 928-957.
- BAEK, K. & QIAN, X. 2011. An Analysis of Political Risks and the Flow of Foreign Direct Investment in Developing and Industrialized Economies. *Economics, Management & Financial Markets*, 6, 4, 60.
- BAGCHI, A. K. 2002. The Other Side of Foreign Investment by Imperial Powers: Transfer of Surplus from Colonies. *Economic and Political Weekly*, 37, 2229-2238.
- BAHARUMSHAH, A. Z. & ALMASAIED, S. W. 2009. Foreign Direct Investment and Economic Growth in Malaysia: Interactions with Human Capital and Financial Deepening. *Emerging Markets Finance and Trade*, 45.
- BAHARUMSHAH, A. Z. & THANOON, M. A.-M. 2006. Foreign Capital Flows and Economic Growth in East Asian Countries. *China Economic Review*, 17, 70-83.
- BAKHT, Z. & BHATTACHARYA, D. 1991. Investment, Employment and Value Added in Bangladesh Manufacturing Sector in 1980s - Evidence and Estimate. *Bangladesh Development Studies*, XIX, 1-1.
- BALASUBRAMANYAM, V. N. 2002. Foreign Direct Investment in Developing Countries: Determinants and Impact.
- BALASUBRAMANYAM, V. N., SALISU, M. & SAPSFORD, D. 1996. Foreign Direct Investment and Growth in EP and IS Countries. *Economic Journal*, 106, 92-105.
- BALASUBRAMANYAM, V. N., SALISU, M. & SAPSFORD, D. 1999. Foreign Direct Investment as an Engine of Growth. *Journal of International Trade and Economic Development*, 8, 27-41.
- BALASUBRAMANYAM, V. N. & SAPSFORD, D. 2007. Does India Need a Lot More FDI? *Economic and Political Weekly*, 42, 1549-1555.
- BANGA, R. 2006. The Export-Diversifying Impact of Japanese and US Foreign Direct Investments in the Indian Manufacturing Sector. *Journal of International Business Studies*, 37, 558-568.
- BARDHAN, P. 2002. Decentralization of Governance and Development. Journal of Economic

Perspectives, 16, 185-205.

- BARDHAN, P. 2009. India and China: Governance Issues of Development. *Journal of Asian Studies*, 68, 347-358.
- BARRO ROBERT J. 1991. Economic Growth in a Cross Section of Countries. *The Quarterly Journal of Economics*, 106, 2, 407 – 443.
- BASU, B. & YAO, J. 2009. Foreign Direct Investment and Skill Formation in China. *International Economic Journal*, 23, 163-179.
- BASU, P., CHAKRABORTY, C. & REAGLE, D. 2003. Liberalization, FDI, and Growth in Developing Countries: A Panel Co-integration Approach. *Economic Inquiry*, 41, 510.
- BASU, P. & GUARIGLIA, A. 2007. Foreign Direct Investment, Inequality, and Growth. *Journal of Macroeconomics*, 29, 824-839.
- BAYOUMI, T. & LIPWORTH, G. 1998. Japanese Foreign Direct Investment and Regional Trade. *Journal of Asian Economics - International Journal*, 9, 581-608.
- BEBBINGTON, A., DHARMAWAN, L., FAHMI, E. & GUGGENHEIM, S. 2006. Local Capacity, Village Governance, and the Political Economy of Rural Development in Indonesia. *World Development*, 34, 1958-1976.
- BEKAERT, G., HARVEY, C. R. & LUNDBLAD, C. 2011. Financial Openness and Productivity. *World Development*, 39, 1-19.
- BELADI, H., MARJI, S. & CHAKRABARTI, A. 2009. Tariff Jumping and Joint Ventures. *Southern Economic Journal*, 75, 1256-1269.
- BEN NACEUR, S., BAKARDZHIEVA, D. & KAMAR, B. 2012. Disaggregated Capital Flows and Developing Countries' Competitiveness. *World Development*, 40, 223-237.
- BENDE-NABENDE, A. & FORD, J. L. 1998. FDI, Policy Adjustment and Endogenous Growth: Multiplier Effects from a Small Dynamic Model for Taiwan, 1959-1995. World Development, 26, 1315-1331.
- BENDE-NABENDE, A., FORD, J. L. & SLATER, J. R. 2000. The Impact of FDI on the Economic Growth of the ASEAN-5 Economies, 1970-94: A Comparative Dynamic Multiplier Analysis from a Small Model, with Emphasis on Liberalization.
- BENDE-NABENDE, A. & JAMES, W. E. 2000. FDI, Regionalism, Government Policy and Endogenous Growth: A Comparative Study of the ASEAN-5 Economies, with Development Policy Implications for the Least Developed Countries. *Asian-Pacific Economic Literature*, 14, 53-54.
- BENG, C. S. 1980. The Role of Japanese Direct Investment in Malaysia. Malayan Economic Review, 25.

BERTHÉLEMY, J. C. & DÉMURGER, S. 2000. Foreign Direct Investment and Economic

Growth: Theory and Application to China. *Review of Development Economics*, 4, 140-156.

- BEUGELSDIJK, S., SMEETS, R. & ZWINKELS, R. 2008. The Impact of Horizontal and Vertical FDI on Host's Country Economic Growth. *International Business Review*, 17, 452-472.
- BEYER, J. 2002. "Please Invest in Our Country"- How Successful were the Tax Incentives for Foreign Investment in Transition Countries? *Communist and Post-Communist Studies*, 35.
- BHADURI, S. N. 2005. Investment, Financial Constraints and Financial Liberalization: Some Stylized Facts from a Developing Economy, India. *Journal of Asian Economics*, 16, 704-718.
- BHALLA, A. S. 1998. Sino-Indian Liberalization: The Role of Trade and Foreign Investment. Economic Change and Restructuring, 31, 151-174.
- BHAT, K. S. & RAJ, K. D. 2006. Foreign Direct Investment and Foreign Trade in India: What Causes What? *Indian Journal of Economics*, LXXXVII, 237-252.
- BHAT, K. S., SUNDARI, C. U. T. & RAJ, K. D. 2004. Causal Nexus between Foreign Investment and Economic Growth in India. *Indian Journal of Economics*, LXXXV, 171-185.
- BHATTI, A. H. & MALIK, M. M. 2001. Growth and Poverty in Pakistan: Implications for Governance [with Comments]. *The Pakistan Development Review*, 40, 831-844.
- BINICI, M., HUTCHISON, M. & SCHINDLER, M. 2010. Controlling Capital? Legal Restrictions and the Asset Composition of International Financial Flows. *Journal of International Money and Finance*, 29, 666 – 684.
- BLACKMAN, A. & WU, X. 1999. Foreign Direct Investment in China's Power Sector: Trends, Benefits and Barriers. *Energy Policy*, 27.
- BLAIR, H. W. 1978. Rural-Development, Class-Structure and Bureaucracy in Bangladesh. World Development, 6, 65-82.
- BLANTON, S. L. & BLANTON, R. G. 2007. What Attracts Foreign Investors? An Examination of Human Rights and Foreign Direct Investment. *Journal of Politics*, 69.
- BLEANEY, M. F. 1996. Macroeconomic Stability, Investment and Growth in Developing Countries. *Journal of Development Economics*, 48.
- BOISOT, M. & CHILD, J. 1990. Efficiency, Ideology and Tradition in the Choice of Transactions Governance Structures - The Case of China as a Modernizing Society.
- BORENSZTEIN, E., DE GREGORIO, J. & LEE, J. W. 1998. How does Foreign Direct Investment Affect Economic Growth? *Journal of International Economics*, 45, 115-136.
- BOSWORTH, B. P. & COLLINS, S. M. 1999. Capital Flows to Developing Economies: Implications for Saving and Investment. *Brookings Papers on Economic Activity*.
- BOUBAKRI, N., COSSET, J.-C. & GUEDHAMI, O. 2004. Privatization, Corporate Governance and Economic Environment: Firm-Level Evidence from Asia. *Pacific-Basin Finance Journal*, 12, 65.
- BRAHMASRENE, T. & JIRANYAKUL, K. 2009. Capital Mobility in Asia: Evidence from Bounds Testing of Co-integration between Savings and Investment. *Journal of the Asia Pacific Economy*, 14, 262-269.
- BRAMBILLA, I., HALE, G. & LONG, C. 2009. Foreign Direct Investment and the Incentives to Innovate and Imitate. *Scandinavian Journal of Economics*, 111, 835-861.
- BRANSTETTER, L. G. & FEENSTRA, R. C. 2002. Trade and Foreign Direct Investment in China: A Political Economy Approach. *Journal of International Economics*, 58.
- BRINK, C. 2003. Beyond Sweatshops: Foreign Direct Investment and Globalization in Developing Countries. *Millennium-Journal of International Studies*, 32.
- BROADMAN, H. G. 2002. Weak State, Strong Networks: The Institutional Dynamics of Foreign Direct Investment in China. *Journal of Asian Studies*, 61.
- BROOKS, D. H., FAN, E. X. & SUMULONG, L. R. 2003. Foreign Direct Investment in Developing Asia: Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations, Asian Development Bank.
- BROOKS, D. H. & HILL, H. 2004. Divergent Asian Views on Foreign Direct Investment and its Governance. *Asian Development Review*, 21, 1-36.
- BROUTHERS, L. E., GAO, Y. & JASON PATRICK MCNICOL 2008. Corruption and Market Attractiveness Influences on Different Types of FDI. *Strategic Management Journal*, 29, 673-680.
- BRUNETTI, A. & WEDER, B. 1998. Investment and Institutional Uncertainty: A Comparative Study of Different Uncertainty Measures. Weltwirtschaftliches Archiv-Review of World Economics, 134.
- BRUNO, RL; NAURO, C. 2011. A Systematic Review of the Effect of FDI on EconomicGrowth in Low Income Countries: A Meta-Regression- Analysis. (Systematic Reviews).DFID: London, United Kingdom.
- BUCKLEY, P. J., CLEGG, J., CROSS, A. R. & HUI, T. 2005. China's Inward Foreign Direct Investment Success: Southeast Asia in the Shadow of the Dragon. *Multinational Business Review (St. Louis University)*, 13, 3-31.
- BUCKLEY, P. J., CLEGG, J., PING, Z., SILER, P. A. & GIORGIONI, G. 2007a. The Impact of Foreign Direct Investment on the Productivity of China's Automotive Industry.

Management International Review (MIR), 47, 707-724.

- BUCKLEY, P. J., WANG, C. & CLEGG, J. 2007b. The Impact of Foreign Ownership, Local Ownership and Industry Characteristics on Spillover Benefits from Foreign Direct Investment in China. *International Business Review*, 16, 142-158.
- BUNYARATAVEJ, K., HAHN, E. D. & DOH, J. P. 2008. Multinational Investment and Host Country Development: Location Efficiencies for Services Offshoring. *Journal of World Business*, 43, 227-242.
- BURGER, M. J. & KARREMAN, B. 2010. Foreign Direct Investment, China and the World Economy. *Regional Studies*, 44.
- BURHOP, C. 2005. Foreign Assistance and Economic Development: A Re-Evaluation. *Economics Letters*, 86, 57-61.
- BURKE, P. J. & AHMADI-ESFAHANI, F. Z. 2006. Aid and Growth: A Study of South East Asia. *Journal of Asian Economics*, 17, 350-362.
- BURKI, S. J. 1992. Privatisation, Deregulation, Regulation and Good Governance A New Economic-Development Paradigm for Pakistan.
- BUSSCHE, S. V. & VERBEKE, A. 2008. The Impact of Administrative Distance and Good Governance on Multinational. *Perspective*, 2, 68-81.
- BUSSE, M. 2004. Transnational Corporations and Repression of Political Rights and Civil Liberties: An empirical analysis. *KYKLOS*, 57, 1, 45 66.
- BUSSE, M. & GROIZARD, J. L. 2008. Foreign Direct Investment, Regulations and Growth. The World Economy.
- BUSSE, M. & HEFEKER, C. 2005. Political Risk, Institution and Foreign Direct Investment. *HWWA Discussion Paper*, 315.
- BUSSE, M., NUNNENKAMP, P. & SPATAREANU, M. 2011. Foreign Direct Investment and Labour Rights: A Panel Analysis of Bilateral FDI Flows. *Applied Economics Letters*, 18, 149-152.
- BUSSIÈRE, M. & FRATZSCHER, M. 2008. Financial Openness and Growth: Short-Run Gain, Long-Run Pain? *Review of International Economics*, 16, 69-95.
- BUSSMANN, M. 2010. Foreign Direct Investment and Militarized International Conflict. Journal of Peace Research, 47, 143-154.
- BUTKIEWICZ, J. L. & YANIKKAYA, H. 2004. Institutional Quality and Economic Growth: Maintenance of the Rule of Law or Democratic Institutions, or Both? *Economic Modeling*, 23, 648-661.
- BUTKIEWICZ, J. L. & YANIKKAYA, H. 2011. Institutions and the Impact of Government Spending on Growth. *Journal of Applied Economics*, 14, 319-341.

- BÉNASSY-QUÉRÉ, A., COUPET, M. & MAYER, T. 2007. Institutional Determinants of Foreign Direct Investment. *The World Economy*, 30, 764-782.
- CALVO, G. A., LEIDERMAN, L. & REINHART, C. M. 1996. Inflows of Capital to Developing Countries in the 1990s. *Journal of Economic Perspectives*, 10.
- CALÌ, M. & SEN, K. 2011. Do Effective State Business Relations Matter for Economic Growth? Evidence from Indian States. *World Development*, 39, 1542-1557.
- CAMPOS, N. F. & KINOSHITA, Y. 2002. Foreign Direct Investment as Technology Transferred: Some Panel Evidence from the Transition Economies. *Manchester School*, 70, 398-420.
- CAMPOS, N. F. & KINOSHITA, Y. 2010. Structural Reforms, Financial Liberalization, and Foreign Direct Investment. *IMF Staff Papers*, 57, 326-365.
- CAMPOS, N. F. & NUGENT, J. B. 1999. Development Performance and the Institutions of Governance: Evidence from East Asia and Latin America. *World Development*, 27, 439-452.
- CARKOVIC, M. & LEVINE, R. 2002. Does Foreign Direct Investment Accelerate Economic Growth? University of Minnesota Department of Finance Working Paper.
- CENTRE FOR REVIEWS AND DISSEMINATION (CRD). 2009. Systematic Reviews: CRD's Guide for Undertaking Reviews in Health Care. *York, UK: Publishing Services*.
- CHAKRABORTY, C. & BASU, P. 2002. Foreign Direct Investment and Growth in India: A Co-Integration Approach. *Applied Economics*, 34.
- CHAKRABORTY, C. & NUNNENKAMP, P. 2008. Economic Reforms, FDI, and Economic Growth in India: A Sector Level Analysis. *World Development*, 36, 1192-1212.
- CHAND, S. 2007. Governance for Growth: Priorities for a Reform-Minded Papua New Guinea Government. *Pacific Economic Bulletin*, 22, 70-82.
- CHANG, N. 2012. The Empirical Relationship between Openness and Environmental Pollution in China. *Journal of Environmental Planning & Management*, 55, 783-796.
- CHANG, P.L. & LU, C.H. 2012. Risk and the Technology Content of FDI: A Dynamic Model. *Journal of International Economics*, 86, 306-317.
- CHANG, S.C. 2007. The Dynamic Interactions among Foreign Direct Investment, Economic Growth, Exports and Unemployment: Evidence from Taiwan. *Economic Change and Restructuring*, 38, 235-257.
- CHANG, Y. C., KAO, M. S., KUO, A. & CHIU, C. F. 2012. How Cultural Distance Influences Entry Mode Choice: The Contingent Role of Host Country's Governance Quality. *Journal of Business Research*, 65, 1160-1170.

CHANGWEN, Z. & JIANG, D. 2007. Causality between FDI and Economic Growth in China.

Chinese Economy, 40, 68-82.

- CHANGWEN, Z. & JIANG, D. 2009. Capital Formation and Economic Growth in Western China. *Chinese Economy*, 42, 7-26.
- CHANTASASAWAT, B., FUNG, K. C., IIZAKA, H. & SIU, A. 2010. FDI Flows to Latin America, East and Southeast Asia, and China: Substitutes or Complements? *Review of Development Economics*, 14, 533-546.
- CHARI, V. V. 2004. Discussion of Growth and Foreign Direct Investment: Does Policy Play a Role? *American Journal of Agricultural Economics*, 86, 802-804.
- CHATTERJEE, P. 2008. Democracy and Economic Transformation in India. *Economic and Political Weekly*, 43, 53-62.
- CHATTERJEE, S. 2009. An Economic Analysis of Foreign Direct Investment in India. Department of Economics, The Maharaja Sayajirao University of Baroda, Gujarat.
- CHAUDHARY, M. A. & QAISRANI, A. A. 2002. Trade Instability, Investment and Economic Growth in Pakistan. *Pakistan Economic and Social Review*, XL, 57-74.
- CHAUDHURI, S., YABUUCHI, S. & MUKHOPADHYAY, U. 2006. Inflow of Foreign Capital and Trade Liberalization in a Model with an Informal Sector and Urban Unemployment. *Pacific Economic Review*, 11, 87-103.
- CHEN, C., CHANG, L. & ZHANG, Y. 1995. The Role of Foreign Direct Investment in China's Post-1978 Economic Development. *World Development*, 23, 691-703.
- CHEN, C. M., MELACHROINOS, K. A. & CHANG, K. T. 2010a. FDI and Local Economic Development: The Case of Taiwanese Investment in Kunshan. *European Planning Studies*, 18, 213-238.
- CHEN, F., LEOW, K. W., NG, C. P., TUNESH, L. & WONG, V. Z. W. 2011a. The Effect of China's Foreign Direct Investment (FDI) on the Malaysian Economics Performance. *UTAR*.
- CHEN, J. J. H. 1997. China's Construction Industry and Foreign Investment. *Building Research and Information*, 25.
- CHEN, J. M., MA, S. C. & XU, H. Q. 2005. Analysis of FDI's Effect on Economic Growth in China by Panel Data Model.
- CHEN, W., XIE, T. & HAO, H. 2010b. A Research on Relationship among FDI, Foreign Trade and Economic Growth in Xingjiang Province of China.
- CHEN, Y., LI, B. & ZHANG, H. 2011b. Foreign Direct Investment, Consumption and Economic Growth in China.
- CHEN, Y.F. & FUNKE, M. 2011. Institutional Uncertainty, Economic Integration, and Vertical Foreign Direct Investment Decisions. *Open Economies Review*, 22.

- CHEN, Y.R., YANG, C., HSU, S.-M. & WANG, Y.-D. 2009. Entry Mode Choice in China's Regional Distribution Markets: Institution vs. Transaction Costs Perspectives. *Industrial Marketing Management*, 38.
- CHEN, Z., GE, Y. & LAI, H. 2011c. Foreign Direct Investment and Wage Inequality: Evidence from China. *World Development*, 39.
- CHENG, H. F. & ZHANG, S. Z. 2003. The Crowd-Out Effect of Foreign Direct Investment in China on Domestic Investment.
- CHENG, T.J. & HAGGARD, S. 1998. Institutions and Growth in Korea and Taiwan: The Bureaucracy. *Journal of Development Studies*, 34, 87.
- CHENGGANG, L., WEI, Y. & ZHONGWEI, Z. 2007. The Empirical Analysis on the Relationship between Foreign Direct Investment and Indigenous Innovation Capability in China.
- CHENGQI, W. & LI, Y. 2007. Do Spillover Benefits Grow with Rising Foreign Direct Investment? An Empirical Examination of the Case of China. *Applied Economics*, 39, 397-405.
- CHENG HSIAO & YAN SHEN 2003. Foreign Direct Investment and Economic Growth: The Importance of Institutions and Urbanization. *Economic Development and Cultural Change*, 51, 883-896.
- CHEUNG, K. Y. 2010. Spillover Effects of FDI via Exports on Innovation Performance of China's High-Technology Industries. *Journal of Contemporary China*, 19.
- CHEUNG, K. Y. & LIN, P. 2004. Spillover Effects of FDI on Innovation in China: Evidence from the Provincial Data. *China Economic Review*, 15.
- CHEW GING, L. 2009. Foreign Direct Investment, Pollution and Economic Growth: Evidence from Malaysia. *Applied Economics*, 41, 1709-1716.
- CHHATRE, A. & SABERWAL, V. 2006. Democracy, Development and (re-) Visions of Nature: Rural Conflicts in the Western Himalayas. *Journal of Peasant Studies*, 33, 678-706.
- CHING-MU, C., MELACHROINOS, K. A. & CHANG, K.T. 2010. FDI and Local Economic Development: The Case of Taiwanese Investment in Kunshan. *European Planning Studies*, 18, 213-238.
- CHOE, J. I. 2003. Do Foreign Direct Investment and Gross Domestic Investment Promote Economic Growth? *Review of Development Economics*, 7, 44-58.
- CHOI, S.-W. & SAMY, Y. 2008. Re-Examining the Effect of Democratic Institutions on Inflows of Foreign Direct Investment in Developing Countries. *Foreign Policy Analysis*, 4, 83-103.

- CHOONG, C.-K., YUSOP, Z. & SOO, S.C. 2005. Foreign Direct Investment and Economic Growth in Malaysia: The Role of Domestic Financial Sector. *Singapore Economic Review*, 50, 245-268.
- CHOUDHARY, K. & MONDAL, P. 2007. Globalisation, Governance Reforms and Development in India. *Indian Journal of Social Work*, 68, 437-443.
- CHOW, C. K. W. & FUNG, M. K. Y. 1996. Health Care Reform, Foreign Direct Investment and their Impact on the Pharmaceutical Industry in China. *Asian Economies*, 25, 55-74.
- CHOW, G. C. 2006. Globalization and China's Economic Development. *Pacific Economic Review*, 11, 271-285.
- CHOW, P. 2005. Financial Restructuring and Corporate Governance in Korea and Taiwan after 1997: Toward Sustainable Development in East Asia. *Joint U.S. Korea academic studies*, 15, 77-102.
- CHOW, Y.-F. & ZENG, J. 2001. Foreign Capital in a Neoclassical Model of Growth. *Applied Economics Letters*, 8, 613-615.
- CHOWDHURY, A. & MAVROTAS, G. 2006. FDI and Growth: What Causes What? *World Economy*, 29, 9-19.
- CHRISTENSEN, P. J., AUSTRALASIAN INST, M. I. N. & MET 1993. South-East Asia -Foreign-Investment in the Mining Sector.
- CHRISTERSON, B. 2000. Foreign Investment and Development: The Positive Effects of Embedded Foreign Investment in China. *Competition & Change*, 4, 325.
- CHUANG, Y.C. & LIN, C.M. 1999. Foreign Direct Investment, R&D and Spillover Efficiency: Evidence from Taiwan's Manufacturing Firms. *Journal of Development Studies - London*, 35, 117-138.
- CLARKE, T. & DU, Y. X. 1998. Corporate Governance in China: Explosive Growth and New Patterns of Ownership. *Long Range Planning*, 31, 239-251.
- CO, C. Y., LIST, J. A. & QUI, L. D. 2004. Intellectual Property Rights, Environmental Regulations, and Foreign Direct Investment. *Land Economics*, 80, 153-173.
- THE COCHRANE COLLABORATION. 2011. Cochrane Handbook for Systematic Reviews of Interventions. [online] Available at: http://handbook.cochrane.org/ [Accessed 27 Sep. 2012].
- COLE, M. A., ELLIOTT, R. J. R. & JING, Z. 2009. Corruption, Governance and FDI Location in China: A Province-Level Analysis. *Journal of Development Studies*, 45, 1494-1512.
- COLE, M. A., ELLIOTT, R. J. R. & ZHANG, J. 2011. Growth, Foreign Direct Investment, and the Environment: Evidence from Chinese Cities. *Journal of Regional Science*, 51, 121-138.

- COLEN, L., MAERTENS, M. & SWINNEN, J. 2009. Foreign Direct Investment as an Engine for Economic Growth and Human Development: A Review of the Arguments and Empirical Evidence. *Human Rights. & International Legal Discourse*, 3.
- COMMANDER, S. & NIKOLOSKI, Z. 2010. Institutions and Economic Performance: What can be Explained? *European Bank for Reconstruction and Development Working Paper Number*, 21.
- CONTRACTOR, F. J. 1995. Promoting Foreign Direct Investment in Developing Countries. International Trade Journal, 9, 107-142.
- COUPET, M. & MAYER, T. 2005. Institutional Determinants of Foreign Direct Investment. *CEPII Working Paper*.
- CURRIE, B. 1996. Governance, Democracy and Economic Adjustment in India: Conceptual and Empirical Problems. *Third World Quarterly*, 17, 787-807.
- CYRUS, T. L., ISCAN, T. B. & STARKY, S. 2006. Investor Protection and International Investment Positions: An Empirical Analysis. *International Finance*, 9, 197-221.
- CÁRDENAS, M. 1992. Capital Flows and Investment in the Growth Models: An Empirical Research in 81 Developing Countries. *Coyuntura Económica*, XXII, 113-138.
- DAHLSTROM, T. & JOHNSON, A. 2007. Bureaucratic, Corruption, MNEs and FDI. CESIS Electronic Working Paper Series, 82.
- DAI, X. 2011. Comparative Study on the Innovation Ability of Foreign Direct Investment in Major Capital Cities in China.
- DANHUI, Y. 2010. Foreign Direct Investment and Development of High-Growth Industrial Sectors in China, 1998-2006. *Chinese Economy*, 43, 93-114.
- DAR, A. & AMIRKHALKHALI, S. 2011. On the Impact of Openness and Regulatory Quality on Saving Investment Dynamics in Emerging Economies. *Applied Econometrics and International Development*, 11, 21-28.
- DAS, D. K. 2007. Foreign Direct Investment in China: Its Impact on the Neighbouring Asian Economies. *Asian Business & Management*, 6, 285-301.
- DASH, B. B. & RAJA, A. V. 2009. Institutions and the Quality of Governance: An Empirical Study on Interstate Differences in Economic Development in India. *Asia-Pacific Development Journal*, 16, 1-26.
- DASH, R. K. & SHARMA, C. 2011. FDI, Trade, and Growth Dynamics: New Evidence from the Post-Reform India. *International Trade Journal*, 25, 233-266.
- DAVIS, G. D. 2011. Regional Trade Agreements and Foreign Direct Investment. *Politics and Policy*, 39.

DE BEULE, F. & VAN DEN BULCKE, D. 2002. Foreign Invested Enterprises and Economic

Development. The Case of China. Status: published.

- DE LA TORRE, J. 1981. Foreign Investment and Economic Development: Conflict and Negotiation. *Journal of International Business Studies*, 12, 9-32.
- DE SOYSA, I. & ONEAL, J. R. 1999. Boon or Bane? Reassessing the Productivity of Foreign Direct Investment. *American Sociological Review*, 64, 766-782.
- DE VITA, G. & KYAW, K. S. 2009. Growth Effects of FDI and Portfolio Investment Flows to Developing Countries: A Disaggregated Analysis by Income Levels. *Applied Economics Letters*, 16.
- DEES, S. 1998. Foreign Direct Investment in China: Determinants and Effects. *Economics of Planning*, 31.
- DELLEPIANE-AVELLANEDA, S. 2010. Good Governance, Institutions and Economic Development: Beyond the Conventional Wisdom. *British Journal of Political Science*, 40, 195-224.
- DEMELLO, L. R. 1997. Foreign Direct Investment in Developing Countries and Growth: A Selective Survey. *Journal of Development Studies*, 34.
- DEMURGER, S. 1998. Interdependence between Foreign Investment and Growth in China: Evidence from Panel Data. *Revue Economique*, 49.
- DENG, Z., FALVEY, R. & BLAKE, A. 2012. Trading Market Access for Technology? Tax Incentives, Foreign Direct Investment and Productivity Spillovers in China. *Journal of Policy Modeling*, 34.
- DHANANI, S. & HASNAIN, S. A. 2002. The Impact of Foreign Direct Investment on Indonesia's Manufacturing Sector. *Journal of the Asia Pacific Economy*, 7, 61-94.
- DHAR, B. & ROY, S. S. 1996. Foreign Direct Investment and Domestic Savings-Investment Behaviour - Developing Countries' Experience. *Economic and Political Weekly*, 31.
- DIAO, X., RATTSØ, J. & STOKKE, H. E. 2005. International Spillovers, Productivity Growth and Openness in Thailand: An Inter Temporal General Equilibrium Analysis. *Journal of Development Economics*, 76, 429-450.
- DIMELIS, S. P. & PAPAIOANNOU, S. K. 2010. FDI and ICT Effects on Productivity Growth: A Comparative Analysis of Developing and Developed Countries. *European Journal of Development Research*, 22, 79-96.
- DIXIT, A. 2009. Governance Institutions & India's Development. *Indian Journal of Industrial Relations*, 44, 539-553.
- DOCES, J. A. 2010. The Dynamics of Democracy and Direct Investment: An Empirical Analysis. *Polity*, 42.
- DORAISAMI, A. G. 2007. Financial Crisis in Malaysia: Did FDI Flows Contribute to

Vulnerability? Journal of International Development, 19, 949-962.

- DOUCOULIAGOS, H. IAMSIRARAOJ, S. & ULUBASOGLU, M. A. 2010. Foreign Direct Investment and Economic Growth: A Real Relationship or Wishful Thinking? *Deakin University Working Paper* SWP 2010/14.
- DOUCOULIAGOS, H, & PALDAM, M. 2008. Aid Effectiveness on Growth: A Meta Study. *European Journal of Political Economy*, 24, 1, 1 – 24.
- DOUCOULIAGOS, H, & PALDAM, M. 2009. The Aid Effectiveness Literature: The Sad Results of 40 years of Research. *Journal of Economic Surveys*, 23, 3, 433 – 461.
- DOUCOULIAGOS, H, & ULUBAȘOĞLU, M. 2008, 'Democracy and Economic Growth: A Meta-Analysis', *American Journal of Political Science*, 52, 1, 61-83.
- DOYTCH, N. & UCTUM, M. 2011. Does the Worldwide Shift of FDI from Manufacturing to Services Accelerate Economic Growth? A GMM Estimation Study. *Journal of International Money and Finance*, 30, 410-428.
- DRIFFIELD, N., JONES, C. & CROTTY, J. 2012. International Business Research and Risky Investments, an Analysis of FDI in Conflict Zones. *International Business Review*.
- DRURY, A. C., KRIECKHAUS, J. & LUSZTIG, M. 2006. Corruption, Democracy and Economic Growth. *International Political Science Review*, 27, 2, 121 136.
- DU, J., LI, H. & WU, X. 2008a. Empirical Analysis on the Negative Technology Spillover
 Effect of Foreign Direct Investment in China. Asian Journal of Technology Innovation, 16.
- DU, J., LU, Y. & TAO, Z. 2008b. Economic Institutions and FDI Location Choice: Evidence from US Multinationals in China. *Journal of Comparative Economics*, 36, 412-429.
- DUA, P. & RASHID, A. I. 1998. Foreign Direct Investment and Economic Activity in India. *Indian Economic Review*, XXXIII, 153-168.
- DUNNING, J.H.1980. Toward An Eclectic Theory of International Production: Some Empirical Tests. *Journal of International Business Studies*, 11, 1, 9 31.
- DURHAM, J. B. 2004. Absorptive Capacity and the Effects of Foreign Direct Investment and Equity Foreign Portfolio Investment on Economic Growth. *European Economic Review*, 48, 285-307.
- DUTT, A. K. 1997. The Pattern of Direct Foreign Investment and Economic Growth. *World Development*, 25, 1925-1937.
- DUTTARAY, M., DUTT, A. K. & MUKHOPADHYAY, K. 2008. Foreign Direct Investment and Economic Growth in Less Developed Countries: An Empirical Study of Causality and Mechanisms. *Applied Economics*, 40.

DWIBEDI, J. K. & CHAUDHURI, S. 2010. Foreign Capital, Return to Education and Child

Labour. International Review of Economics & Finance, 19, 278-286.

EASTERLY, W. 1999. Life During Growth. Journal of Economic Growth, 4, 239 – 275.

- EBNER, A. 2007. Public Policy, Governance and Innovation: Entrepreneurial States in East
 Asian Economic Development. *International Journal of Technology & Globalisation*, 3, 6-6.
- ECONOMIDOU, C., LEI, V. & NETZ, J. S. 2006. International Integration and Growth: A Further Investigation on Developing Countries. *International Advances in Economic Research*, 12, 435-448.
- EGGER, P. & RADULESCU, D. M. 2011. Labour Taxation and Foreign Direct Investment. *The Scandinavian Journal of Economics*, 113, 603-636.
- EKANAYAKE, E. M. & CHATRNA, D. 2010. The Effect of Foreign Aid on Economic Growth in Developing Countries. *Journal of International Business and Cultural Studies*, 3, 2, 1 – 13.
- ELLINGSEN, T. & WÄRNERYD, K. 1999. Foreign Direct Investment and the Political Economy of Protection. *International Economic Review*, 40.
- ELO, K. Z. 2007. The Effect of Capital Controls on Foreign Direct Investment Decisions under Country Risk with Intangible Assets, USA, 'IMF eLibrary'.
- EPPI CENTRE. 2010. EPPI Centre Methods for Conducting Systematic Reviews. Evidence for Policy and Practice Information and Co-ordinating Centre, *Institute of Education*, *University of London*.
- ERDEM, D. 2012. Foreign Direct Investments, Energy Efficiency, and Innovation Dynamics. *Mineral Economics*, 1-15.
- ERICSSON, J. & IRANDOUST, M. 2001. On the Causality between Foreign Direct Investment and Output: A Comparative Study. *International Trade Journal*, 15, 1-27.
- ESCRIBANO, A., GUASCH, J. L., ORTE, M. D. & PENA, J. 2009. Investment Climate Assessment in Indonesia, Malaysia, the Philippines and Thailand: Results from Pooling Firm-Level Data. *Singapore Economic Review*, 54, 335-366.
- EVANS, P. & RAUCH, J. E. 1999. Bureaucracy and Growth: A Cross-National Analysis of the Effects of "Weberian" State Structures on Economic Growth. *American Sociological Review*, 64, 748-765.
- EVRENSEL, A. Y. 2010. Corruption, Growth, and Growth Volatility. *International Review of Economics & Finance*, 19, 501-514.
- FAN, E. X. 2003. Technological Spillovers from Foreign Direct Investment A Survey. Asian Development Review, 20, 34-56.
- FAN, J. P. H., MORCK, R., XU, L. C. & YEUNG, B. 2009. Institutions and Foreign Direct

Investment: China versus the Rest of the World. World Development, 37, 852-865.

- FEENY, S. 2005. The Impact of Foreign Aid on Economic Growth in Papua New Guinea. The Journal of Development Studies, 41, 6, 1092 – 1117.
- FEENY, S. & McGILLIVRAY, M. 2010. Aid and Growth in Small Island Developing States. Journal of Development Studies, 46, 5, 897 – 917.
- FEILS, D. J. & RAHMAN, M. 2011. The Impact of Regional Integration on Insider and Outsider FDI. *Management International Review*, 51, 41-63.
- FELIPE, J., LAVINA, E. & FAN, E. X. 2008. The Diverging Patterns of Profitability, Investment and Growth of China and India during 1980-2003. World Development, 36.
- FENG, Y. 2009. FDI in India and its Economic Effects on Service Industry. *International Journal of Trade and Global Markets*, 2, 179-193.
- FERNÁNDEZ, A. I., GONZÁLEZ, F. & SUÁREZ, N. 2010. How Institutions and Regulation Shape the Influence of Bank Concentration on Economic Growth: International Evidence. *International Review of Law and Economics*, 30, 28-36.
- FLOYD, D. & SUMMAN, S. 2008. Understanding the Main Motives for Foreign Direct Investment, an East-West Country Contrast: Is the Host Country Legislation an Important Factor? *Corporate Governance (Bradford)*, 8, 661-668.
- FOSTER, M. J. 2011. Distribution of FDI Across China: Common Policies but Differing Impacts by Region. *Journal of Chinese Economic and Foreign Trade Studies*, 4, 125-138.
- FRANKS, T. 1989. Bureaucracy, Organization Culture and Development. *Public Administration* & Development, 9, 357-368.
- FRECKLETON, M., WRIGHT, A. & CRAIGWELL, R. 2012. Economic Growth, Foreign Direct Investment and Corruption in Developed and Developing Countries. *Journal of Economic Studies*, 39, 639-652.
- FREEMAN, N. 2004. Harnessing Foreign Direct Investment for Economic Development and Poverty Reduction. *Journal of the Asia Pacific Economy*, 9, 209-223.
- FRY, M. J. 1996. How Foreign Direct Investment in Pacific Asia Improves the Current Account. Journal of Asian Economics, 7, 459-486.
- FU, X. 2008. Foreign Direct Investment, Absorptive Capacity and Regional Innovation Capabilities: Evidence from China. Oxford Development Studies, 36, 89-110.
- FU, X. & GONG, Y. 2011. Indigenous and Foreign Innovation Efforts and Drivers of Technological Upgrading: Evidence from China. World Development, 39, 1213-1225.
- FU, X. L. & BALASUBRAMANYAM, V. N. 2005. Exports, Foreign Direct Investment and Employment: The Case of China. World Economy, 28.

- FUNG, H.-G., ZHANG, K. H., LEUNG, W. K., LO, W. C., CHAN, K. C., CHENG, L. T. W., FUNG, J. K. W., XU, X. E. & PEI, C. 2001. Financial Reforms and Foreign Investments in China: Challenges and Opportunities. *Chinese Economy*, 34, 3-100.
- FUNG, K. C., IIZAKA, H. & TONG, S. Y. 2004. Foreign Direct Investment in China: Policy, Recent Trend and Impact. *Global Economic Review*, 33, 99-130.
- FUNG, M.-Y., ZENG, J. & ZHU, L. 1999. Foreign Capital, Urban Unemployment, and Economic Growth. *Review of International Economics*, 7, 651-665.
- GALLAGHER, K. P. & ZARSKY, L. 2005. No Miracle Drug: Foreign Direct Investment and Sustainable Development. International Investment for Sustainable Development: Balancing Rights and Rewards, 45.
- GAMBER, E. N. & SCOTT, A. K. S. 2007. A Threshold Analysis of the Relationship between Governance and Growth. *International Economic Journal*, 21, 255-278.
- GANGTI, Z. & KONG YAM, T. 2000. Foreign Direct Investment and Labour Productivity: New Evidence from China as the Host. *Thunderbird International Business Review*, 42, 507-528.
- GANI, A. 2007. Governance and Foreign Direct Investment Links: Evidence from Panel Data Estimations. *Applied Economics Letters*, 14, 753-756.
- GANI, A. 2011. Governance and Growth in Developing Countries. *Journal of Economic Issues* (*M.E. Sharpe Inc.*), 45, 19-40.
- GAO, T. 2005. Foreign Direct Investment and Growth under Economic Integration. *Journal of International Economics*, 67, 157-175.
- GAO, Y. & QI, Z. 2007. A Research on the Relation between Real Estate Investment and Economic Growth in China.
- GARCÍA, C. 2007. Capital Inflows, Policy Responses, and their Adverse Effects: Thailand,
 Malaysia, and Indonesia in the Decade before the Crisis. *Problemas del desarrollo*, 38, 9-40.
- GASTANAGA, V. M., NUGENT, J. B. & PASHAMOVA, B. 1998. Host Country Reforms and FDI Inflows: How Much Difference do they make? *World Development*, 26.
- GAZIOGLOU, S. & MCCAUSLAND, W. D. 2002. The Costs of Inward Direct Foreign Investment to Developing Countries. *Applied Economics Letters*, 9.
- GE, Y. 2006. The Effect of Foreign Direct Investment on the Urban Wage in China: An Empirical Examination. Urban Studies, 43.
- GELAN, A. 2004. The Effects of Multinational Enterprises Investment in the Non-Traded Sector of Developing Economies. *Journal of Economic Development*, 29, 41-64.
- GEMMA ESTRADA, D. P., & ARIEF RAMAYANDI .2010. Financial Development and

Economic Growth in Developing Asia. *Financial Development and Economic Growth in Developing Asia.* Asian Development Bank.

- GENG, Q. & SHEN, K. R. 2004. FDI and Innovation in China: Evidence from Panel Data of Patent 1987-2002.
- GHOSH, D. N. 1992. Making Foreign Investment Flow. *Economic and Political Weekly*, 27, 1585-1586.
- GHOSH, D. N. 2005. FDI and Reform: Significance and Relevance of Chinese Experience. *Economic and Political Weekly*, 40, 5388-5392.
- GIRMA, S., GONG, Y. & GÖRG, H. 2008. Foreign Direct Investment, Access to Finance, and Innovation Activity in Chinese Enterprises. *World Bank Economic Review*, 22, 367-382.
- GIRMA, S. & YUNDAN, G. 2008. FDI, Linkages and the Efficiency of State-Owned Enterprises in China. *Journal of Development Studies*, 44, 728-749.
- GIROUD, A. 2003. The Impact of Foreign Direct Investment on the Economic Development of ASEAN Economies: A Preliminary Analysis Hafiz Mirza.
- GIROUD, A. 2004. Regionalisation, Foreign Direct Investment and Poverty Reduction: The Case of ASEAN Hafiz Mirza.
- GLOBERMAN, S. & SHAPIRO, D. 2002a. Global Foreign Direct Investment Flows: The Role of Governance Infrastructure. *World Development*, 30.
- GLOBERMAN, S. & SHAPIRO, D. 2002b. Governance Infrastructure and US Foreign Direct Investment. *Journal of International Business Studies*, 34, 19-39.
- GLOBERMAN, S. & SHAPIRO, D. 2009. Assessing Recent International Mergers and Acquisitions as a Mode of Foreign Direct Investment.
- GOLDSMITH, A. A. 1995. Democracy, Property Rights and Economic Growth. *The Journal of Development Studies*, 32, 2, 157 174.
- GOLUB, S. S. 2009. Openness to Foreign Direct Investment in Services: An International Comparative Analysis. *World Economy*, 32, 1245-1268.
- GONG, X. & WEN, S.-H. 2008. The Impact of FDI and Technological Innovation on Eastern, Middle and Western China's Economics.
- GOODSPEED, T., MARTINEZ-VAZQUEZ, J. & ZHANG, L. 2011. Public Policies and FDI Location: Differences between Developing and Developed Countries. *Finanzarchiv*, 67.
- GORDON, L. A., LOEB, M. P. & ZHU, W. 2012. The Impact of IFRS Adoption on Foreign Direct Investment. *Journal of Accounting and Public Policy*, 31.
- GOUNDER, R. 2004. Fiji's Economic Growth Impediments: Institutions, Policies and Coups. *Journal of the Asia Pacific Economy*, 9, 301-324.
- GRINDLE, M. S. 2004. Good Enough Governance: Poverty Reduction and Reform in

Developing Countries. Governance, 17, 525-548.

- GROENEWOLD, N. & SAM HAK KAN, T. 2007. Killing the Goose that Lays the Golden Egg: Institutional Change and Economic Growth in Hong Kong. *Economic Inquiry*, 45, 787-799.
- GROTH, A. J. 2006. Regimes and Economic Development: China, India and Some Others. *The Journal of East Asian Affairs*, 20, 181-202.
- GU, J., HUMPHREY, J. & MESSNER, D. 2008. Global Governance and Developing Countries: The Implications of the Rise of China. *World Development*, 36, 274-292.
- GU, S. 2003. Progress and Problems with the Rule of Law in China. *Contemporary Chinese Thought*, 34, 55-67.
- GUERIN, S. S. & MANZOCCHI, S. 2009. Political Regime and FDI from Advanced to Emerging Countries. *Review of World Economics - Weltwirtschaftliches Archiv*, 145.
- GUYOT, J. F. 1970. Bureaucracy and Rural Development in Malaysia Study of Complex Organizations in Stimulating Economic Development in New States. *Economic Development and Cultural Change*, 18, 476-480.
- GÖRG, H. & GREENAWAY, D. 2004. Much Ado about Nothing? Do Domestic Firms Really Benefit from Foreign Direct Investment? *World Bank Research Observer*, 19, 171-198.
- HABIB, M. & ZURAWICKI, L. 2001. Country-Level Investments and the Effect of Corruption - Some Empirical Evidence. *International Business Review*, 10.
- HAGGARD, S. 1999. Governance and Growth: Lessons from the Asian Economic Crisis. *Asian-Pacific Economic Literature*, 13, 30.
- HAGGARD, S. 2004. Institutions and Growth in East Asia. *Studies in Comparative International Development*, 38, Issue 4, pp. 53 – 81.
- HAGGARD, S., MACINTYRE, A. & TIEDE, L. 2008. The Rule of Law and Economic Development. *Annual Review of Political Science*, 11, 205-234.
- HAGGARD, S. & TIEDE, L. 2011. The Rule of Law and Economic Growth: Where Are We? *World development*, 39, 673-685.
- HAISHUN, S. 1999. Impact of FDI on the Foreign Trade of China. *Journal of the Asia Pacific Economy*, 4, 317.
- HALE, G. & LONG, C. 2011. Are there Productivity Spill overs from Foreign Direct Investment in China? *Pacific Economic Review*, 16.
- HAQ, R., ZIA, U. & ARIF, G. M. 2006. Governance and Pro-Poor Growth: Evidence from Pakistan. *Pakistan Development Review*, 45, 761-776.
- HAQUE, M. S. 2001. Recent Transition in Governance in South Asia: Contexts, Dimensions, and Implications. *International Journal of Public Administration*, 24, 1405-1436.

- HAQUE, M. S. 2004. Governance based on Partnership with NGOs: Implications for Development and Empowerment in Rural Bangladesh. *International Review of Administrative Sciences*, 70, 271-290.
- HARMS, P. & LUTZ, M. 2006. Aid, Governance and Private Foreign Investment: Some Puzzling Findings for the 1990s. *Economic Journal*, 116, 773-790.
- HARMS, P. & URSPUNG, H. W. 2002. Do Civil and Political Repression Really Boost Foreign Direct Investments? *Economic Inquiry*, 40, 4, 651 – 663.
- HAVRYLCHYK, O. & PONCET, S. 2007. Foreign Direct Investment in China: Reward or Remedy? *World Economy*, 30.
- HE, C., WANG, J. & CHENG, S. 2011. What Attracts Foreign Direct Investment in China's Real Estate Development? *Annals of Regional Science*, 46, 267-293.
- HE, C. & ZHU, Y. 2010. Real Estate FDI in Chinese Cities: Local Market Conditions and Regional Institutions. *Eurasian Geography and Economics*, 51.
- HELLER, P., HARILAL, K. N. & CHAUDHURI, S. 2007. Building Local Democracy: Evaluating the Impact of Decentralization in Kerala, India. World Development, 35, 626-648.
- HENDRICKS, L. 2000. Equipment Investment and Growth in Developing Countries. *Journal of Development Economics*, 61.
- HENG, T. M. & LOW, L. 1994. The State of Play of Direct Foreign Investment in Asia. *Journal* of Asian Economics, 5, 65-84.
- HERMES, N. & LENSINK, R. 2003. Foreign Direct Investment, Financial Development and Economic Growth. *Journal of Development Studies London*, 40, 142-164.
- HERZER, D. 2012. How Does Foreign Direct Investment Really Affect Developing Countries' Growth? *Review of International Economics*, 20.
- HERZER, D., KLASEN, S. & NOWAK-LEHMANN D, F. 2008. In search of FDI-led Growth in Developing Countries: The Way Forward. *Economic Modelling*, 25, 793-810.
- HIEMENZ, U. 1987. Foreign Direct Investment and Industrialization in ASEAN Countries. Weltwirtschaftliches Archive, 123, 121-139.
- HILL, D. 2009. Governance, Labour and Uneven Development: The Political Economy of the Port Sector in South and South-East Asia.
- HILL, H. 1997. Towards a Political Economy Explanation of Rapid Growth in ASEAN: A Survey and Analysis. *ASEAN Economic Bulletin*, 14, 131-149.
- HILL, H. & JOHNS, B. 1985. The Role of Direct Foreign-Investment in Developing East Asian Countries. *Weltwirtschaftliches Archive-Review of World Economics*, 121.
- HILL, M. 2001. Democracy, Governance, and Economic Performance: East and Southeast Asia.

Singapore Journal of Tropical Geography, 22, 311-313.

- HITAM, M. B. & BORHAN, H. B. 2012. FDI, Growth and the Environment: Impact on Quality of Life in Malaysia. *Procedia Social and Behavioural Sciences*, 50, 333-342.
- HOANG, T. T., WIBOONCHUTIKULA, P. & TUBTIMTONG, B. 2010. Does Foreign Direct Investment Promote Economic Growth in Vietnam? ASEAN Economic Bulletin, 27, 295-311.
- HOI QUOC, L. & POMFRET, R. 2011. Technology Spillovers from Foreign Direct Investment in Vietnam: Horizontal or Vertical Spillovers? *Journal of the Asia Pacific Economy*, 16.
- HOMLONG, N. & SPRINGLER, E. 2010. Economic Development and Foreign Direct
 Investment: How to Create Sustainable Development an Analysis of the Potential for
 Sustainable Development on the Indian Subcontinent. *Panoeconomicus*, 57, 333-348.
- HONG, E. & SUN, L. 2011. Foreign Direct Investment and Total Factor Productivity in China: A Spatial Dynamic Panel Analysis. *Oxford Bulletin of Economics and Statistics*, 73.
- HONGLIN ZHANG, K. 2006. Does International Investment Help Poverty Reduction in China? *Chinese Economy*, 39, 79-90.
- HOUSTON, D. A. 2007. Can Corruption Ever Improve an Economy? Cato journal, 27, 325-342.
- HSIAO, C. & SHEN, Y. 2003. Foreign Direct Investment and Economic Growth: The Importance of Institutions and Urbanization. *Economic Development & Cultural Change*, 51, 883.
- HSIU-LING, W., CHIEN-HSUN, C. & LI-TING, C. 2007. Foreign Trade in China's Electronics Industry. *Eurasian Geography & Economics*, 48, 626-642.
- HUANG, C.H., TENG, K.-F. & TSAI, P.-L. 2010. Inward and Outward Foreign Direct Investment and Poverty: East Asia vs. Latin America. *Review of World Economics*, 146.
- HUANG, J.T. 2004. Spillovers from Taiwan, Hong Kong, and Macau Investment and from other Foreign Investment in Chinese Industries. *Contemporary Economic Policy*, 22, 13-25.
- HUANG, L., LIU, X. & XU, L. 2012. Regional Innovation and Spillover Effects of Foreign Direct Investment in China: A Threshold Approach. *Regional Studies*, 46.
- HUAY HUAY, L. & HUI BOON, T. 2006. Technology Transfer, FDI and Economic Growth in the ASEAN Region. *Journal of the Asia Pacific Economy*, 11, 394-410.
- HUR, J., PARINDURI, R. A. & RIYANTO, Y. E. 2011. Cross-border M&A Inflows and Quality of Country Governance: Developing Versus Developed Countries. *Pacific Economic Review*, 16.
- HUNTINGTIN, S. 1968. Political Order in Changing Societies. New Haven. CT: Yale University Press.
- HWANG, K. K. 1996. South Korea's Bureaucracy and the Informal Politics of Economic

Development. Asian Survey, 36, 306-319.

- HYE, Q. M. A. 2011. Financial Development Index and Economic Growth: Empirical Evidence from India. *Journal of Risk Finance*, 12, 98-111.
- IHARA, R. & IWAHASHI, R. 2007. Attracting Foreign Investment: Optimal ODA policy with Trade Liberalization. *Journal of International Trade & Economic Development*, 16, 193-211.
- IM, H. 2007. Implications of Korean Foreign Direct Investment in China on its Trade Balance: Evidence from Korean Subsidiaries Survey Data. *Journal of Korea Trade*, 11.
- INGHAM, B. & KALAM, A. K. M. 1992. Decentralization and Development: Theory and Evidence from Bangladesh. *Public Administration & Development*, 12, 373-385.
- INOGUCHI, M. 2009. Did Capital Controls Decrease Capital Flows in Malaysia? *Journal of the Asia Pacific Economy*, 14, 27-48.
- IONESCU, L. 2012. Global Competitiveness, Bureaucracy and the Quality of Institutions. *Economics, Management & Financial Markets*, 7, 733-740.
- IRAM, S. & NISHAT, M. 2009. Sector Level Analysis of FDI-Growth Nexus: A Case Study of Pakistan. Pakistan Development Review, 48, 875-884.
- ISHIDA, K. 1998. Japan's Foreign Direct Investment in East Asia: Its Influence on Recipient Countries and Japan's Trade Structure.
- ISLAM, R. 2005. Bureaucracy and Administrative Development in Bangladesh and Nigeria: A Comparative Analysis. *International Journal of Public Administration*, 28, 1009-1030.
- ISLAM, S. 1994. Foreign-Investment Effects on Balance of Payments a Bangladesh Case-Study. *Asian Survey*, 34.
- ITO, B., YASHIRO, N., XU, Z., CHEN, X. & WAKASUGI, R. 2012. How do Chinese Industries Benefit from FDI Spillovers? *China Economic Review (1043951X)*, 23, 342-356.
- J. SAÚL LIZONDO, D. J. M. Foreign Direct Investment, USA, 'IMF eLibrary'.
- JADHAV, P. 2012. Determinants of Foreign Direct Investment in BRICS Economies: Analysis of Economic, Institutional and Political Factor. *Procedia – Social and Behavioral Sciences*, 37, 5 – 14.
- JALILIAN, H., KIRKPATRICK, C. & PARKER, D. 2007. The Impact of Regulation on Economic Growth in Developing Countries: A Cross-Country Analysis. World Development, 35, 87-103.
- JANSEN, K. 1995. The Macroeconomic Effects of Direct Foreign Investment: The Case of Thailand. World Development, 23, 193-210.
- JANSEN, W. J. 2003. What Do Capital Inflows Do? Dissecting the Transmission Mechanism

for Thailand, 1980-1996. Journal of Macroeconomics, 25.

- JARVIS, D. S. 2012. Foreign Direct Investment and Investment Liberalisation in Asia: Assessing ASEAN's Initiatives. *Australian Journal of International Affairs*, 66.
- JAYASURIYA, K. 2001. Democracy, Governance and Economic Performance: East and Southeast Asia. *Australian Journal of Political Science*, 36, 396-397.
- JENKINS, R. 1991. The Impact of Foreign Investment on Less Developed Countries Cross Section Analysis Versus Industry Studies.
- JENSEN, C. 2006. Foreign Direct Investment and Economic Transition: Panacea or Pain Killer? *Europe-Asia Studies*, 58, 881-903.
- JENSEN, E. 1998. Good Governance and Pakistan's Economic Prospects for Progress. *Pakistan Journal of Applied Economics*, XIV, 63-88.
- JENSEN, N. 2008. Political Risk, Democratic Institutions, and Foreign Direct Investment. Journal of Politics, 70.
- JENSEN, N. & MCGILLIVRAY, F. 2005. Federal Institutions and Multinational Investors: Federalism, Government Credibility, and Foreign Direct Investment. *International Interactions*, 31.
- JENSEN, N. M. 2003. Democratic Governance and Multinational Corporations: Political Regimes and Inflows of Foreign Direct Investment. *International Organization*, 57, 587-616.
- JENSEN, O. & OECD. 2002. Foreign Direct Investment in India and South Africa: A Comparison of Performance and Policy.
- JEONG-YEON LEE & MANSFIELD, E. 1996. Intellectual Property Protection and U.S. Foreign Direct Investment. *The Review of Economics and Statistics*, 78, 181-186.
- JEPSEN, E. M. 2010. Governance in Federal India: Democracy and Economic Development.
- JHA, V. 2000. The role of Foreign Direct Investment The Case of India.

JIA, H., MA, J. & IEEE. 2009. FDI Spillovers and Economic Growth in China.

- JIANG, C. X., YANG, Q., LI, S. & WANG, Y. 2011. The Moderating Effect of Foreign Direct Investment Intensity on Local Firms' Intangible Resources Investment and Performance Implications: A Case from China. *Journal of International Management*, 17.
- JIANG, D. C., CHEN, J. A. & ISAAC, D. 1998. The Effect of Foreign Investment on the Real Estate Industry in China. *Urban Studies*, 35.
- JIANG, Y. & ZHOU, K. 2012. Theoretical Model and Empirical Analysis: Effects of Foreign Direct Investment on Income Gaps in China.
- JIN, F., LEE, K. & KIM, Y.K. 2008. Changing Engines of Growth in China: From Exports, FDI and Marketization to Innovation and Exports. *China & World Economy*, 16.

- JINDRA, B. & ROJEC, M. FDI and Knowledge Spillovers in Transition Economies Is there a Need for More Policy Coordination?
- JING, Z. & XIAOLAN, F. 2008. FDI and Environmental Regulations in China. *Journal of the Asia Pacific Economy*, 13, 332-353.
- JINGJING, Z. & ROGERS, J. D. 2009. The Technological Innovation Performance of Chinese Firms: The Role of Industrial and Academic R&D, FDI and the Markets in Firm Patenting. *International Journal of Technology Management*, 48, 518-543.
- JOCELYN GLASS, A. & SAGGI, K. 2002. Licensing versus Direct Investment: Implications for Economic Growth. *Journal of International Economics*, 56, 131-154.
- JON, M. 1999. Patterns of Corruption and Development in East Asia. *Third World Quarterly*, 20, 569-587.
- JONAS, M. 2010. Foreign Direct Investments as an Engine of Technological Change in Malaysia: The Role of Social Interactions. *International Journal of Technological Learning, Innovation and Development*, 3, 164-186.
- JR-TSUNG, H. 2004. Spillovers from Taiwan, Hong Kong, and Macau Investment and from Other Foreign Investment in Chinese Industries. *Contemporary Economic Policy*, 22, 13-25.
- JUN YEUP, K. & LE-YIN, Z. 2008. Formation of Foreign Direct Investment Clustering-A New Path to Local Economic Development? The Case of Qingdao. *Regional Studies*, 42, 265-280.
- JUNG, K. H. 1999. Foreign Direct Investment and Corporate Restructuring in East Asia. *Pacific Review*, 12.
- KAILEI, W., SHUJIE, Y. & AYING, L. 2009. Foreign Direct Investment and Regional Inequality in China. *Review of Development Economics*, 13, 778-791.
- KAJIWARA, H. 1994. The Effects of Trade and Foreign Investment Liberalization Policy on Productivity in the Philippines. *Developing Economies*, 32.
- KAN, S., TANG, L. & KONG, J. 2011. Analysis of the Impact on Foreign Direct Investment to Energy Consumption Intensity in China.
- KASIBHATLA, K. M., STEWART, D. B. & KHOJASTEH, M. 2008. The Role of FDI in High Medium, Low Medium and Low Income Countries during 1970 - 2005: Empirical Tests and Evidence. *Journal of Business & Economic Studies*, 14, 60-72.
- KASUGA, H. 2007. Evaluating the Impacts of Foreign Direct Investment, Aid and Saving in Developing Countries. *Journal of International Money and Finance*, 26, 213-228.
- KATHURIA, V. 2010. Does the Technology Gap Influence Spillovers? A Post-liberalization Analysis of Indian Manufacturing Industries. *Oxford Development Studies*, 38, 145-170.

- KAUFMANN, D., KRAAY, A. & ZOIDO-LOBATON, P. 1999a. Aggregating Governance Indicators. *World Bank Policy Research Department Working Paper No.* 2195.
- KAUFMANN, D., KRAAY, A., & ZOIDO-LOBATON, P. 1999b. Governance Matters. *Policy Research Working Paper No.* 2196.
- KAUFMANN, D., KRAAY, A., & ZOIDO-LOBATON, P. 2007. Growth and Governance: A Reply. The Journal of Politics, 69, 2, 555 562.
- KEPILI, E. I. Z. & MASRON, T. A. 2011. Real Estate-Foreign Direct Investment-Growth in Malaysia: Re-Framing Eclectic Paradigm. *In:* TAO, F. ed. *Economics, Trade and Development.*
- KERKVLIET, B. J. T. 2010. Governance, Development, and the Responsive Repressive State in Vietnam. *Forum for Development Studies*, 37, 33-59.
- KESSING, S. G., KONRAD, K. A. & KOTSOGIANNIS, C. 2007. Foreign Direct Investment and the Dark Side of Decentralization. *Economic Policy*.
- KHADKA, N. 1993. Democracy and Development in Nepal: Prospects and Challenges. *Pacific Affairs*, 66, 44-71.
- KHAMFULA, Y. 2007. Foreign Direct Investment and Economic Growth in EP and IS Countries: The Role of Corruption. *World Economy*, 30, 1843-1855.
- KHAN, M. S. & REINHART, C. M. 1990. Private Investment and Economic Growth in Developing Countries. World Development, 18.
- KHAN, S. M. T.H. MORAN, E.M. GRAHM, M. BLOMSTROM. 2006. Does Foreign Direct Investment Promote Development? Institute for International Economics, Washington, DC. Journal of Asian Economics, 17, 1128-1130.
- KHAN, S. M. & KHAN, Z. S. 2007. World Investment Report 2006, FDI from Developing and Transition Economies: Implications for Development by United Nations Conference on Trade and Development (UNCTAD). United Nations Publications, New York (2006). *Journal of Asian Economics*, 18, 553-561.
- KHOLDY, S. 1995. Causality between Foreign Investment and Spill over Efficiency. *Applied Economics*, 27, 745.
- KIM, D.-H. & TRUMBORE, P. F. 2010. Transnational Mergers and Acquisitions: The Impact of FDI on Human Rights, 1981-2006. *Journal of Peace Research*, 47, 723-735.
- KIM, J. Y. & ZHANG, L.Y. 2008. Formation of Foreign Direct Investment Clustering-A New Path to Local Economic Development? The Case of Qingdao. *Regional Studies -Abingdon*, 42, 265-281.
- KIM, S. 2010. Productivity and Timing of Foreign Direct Investment: Evidence Based on Korean Direct Investment in China. *Economic Papers. Bank of Korea*, 13, 37-51.

- KIM, S. & YANG, D. Y. 2011. The Impact of Capital Inflows on Asset Prices in Emerging Asian Economies: Is Too Much Money Chasing Too Little Good? *Open Economies Review*, 22.
- KIMURA, K. 2012. Does Foreign Direct Investment Affect the Growth of Local Firms? The Case of China's Electrical and Electronics Industry. *China & World Economy*, 20.
- KINOSHITA, T. 1986. Japanese Investment in Indonesia Problems and Prospects. *Bulletin of Indonesian Economic Studies*, 22.
- KIONG, W. H. & JOMO, K. S. 2005. Before the Storm: The Impact of Foreign Capital Inflows on the Malaysian Economy, 1966-1996. *Journal of the Asia Pacific economy*, 10, 56-69.
- KLASRA, M. A. 2010. Foreign Direct Investment, Trade Openness and Economic Growth in Pakistan and Turkey: An Investigation Using Bounds Test. *Quality and Quantity*, 45, 223-232.
- KLASRA, M. A. 2011. Foreign Direct Investment, Trade Openness and Economic Growth in Pakistan and Turkey: An Investigation Using Bounds Test. *Quality & Quantity*, 45.
- KLEIN, M. 2005. Capital Account Liberalization, Institutional Quality and Economic Growth: Theory and Evidence. *NBER Working Paper*, 11112.
- KNIGHT, G. R. & VAN SCHAIK, A. 2001. State and Capital in Late Colonial Indonesia The Sugar Industry, Braakhuur, and the Colonial Bureaucracy in North Central Java. *Bijdragen Tot De Taal- Land- En Volkenkunde*, 157, 831-859.
- KOHPAIBOON, A. 2003. Foreign Trade Regimes and the FDI-Growth Nexus: A Case Study of Thailand. *Journal of Development Studies London*, 40, 55-70.
- KONINGS, J. 2001. The Effects of Foreign Direct Investment on Domestic Firms Evidence from Firm-Level Panel Data in Emerging Economies. *Economics of Transition*, 9.
- KOTRAJARAS, P. 2010. Foreign Direct Investment and Economic Growth: A Comparative Study among East Asian Countries. *Journal of Applied Economics*, 17, 2, 12 26.
- KOTTARIDI, C. 2005. FDI, Growth and the Role of Governance: Changing the Rules of the Game. *Contributions to Political Economy*, 24, 79-79.
- KOTTARIDI, C. & STENGOS, T. 2010. Foreign Direct Investment, Human Capital and Non-Linearities in Economic Growth. *Journal of Macroeconomics*, 32, 858-872.
- KOTWAL, A., RAMASWAMI, B. & WADHWA, W. 2011. Economic Liberalization and Indian Economic Growth: What's the Evidence?? *Journal of Economic Literature*, 49, 1152-1199.
- KRAMMER, S. 2010. International R&D Spillovers in Emerging Markets: The Impact of Trade and Foreign Direct Investment. *Journal of International Trade and Economic Development*, 19, 591-623.

- KUMAR, N. 2005. Liberalisation, Foreign Direct Investment Flows and Development: Indian Experience in the 1990s. *Economic and Political Weekly*, 40, 1459-1469.
- KUNDU, A. 2011. Urban System in India Trends, Economic Base, Governance, and a Perspective of Growth under Globalization.
- KWOK, C. C. Y. & SOLOMON TADESSE. 2006. The MNC as an Agent of Change for Host-Country Institutions: FDI and Corruption. *Journal of International Business Studies*, 37, 767-785.
- LAAKSONEN-CRAIG, S. 2004. Foreign Direct Investments in the Forest Sector: Implications for Sustainable Forest Management in Developed and Developing Countries. *Forest Policy and Economics*, 6.
- LAI, M., PENG, S. & BAO, Q. 2006. Technology Spill overs, Absorptive Capacity and Economic Growth. *China Economic Review (1043951X)*, 17, 300-320.
- LAKSHMANASAMY, T. 2003. New Information, New Institutions and Economic Development. *Asian Economic Review*, 45, 425-439.
- LALL, S. & NARULA, R. 2004. Foreign Direct Investment and its Role in Economic Development: Do We Need a New Agenda? *European Journal of Development Research*, 16, 447-465.
- LALWANI, M. 2002. The Impact of Foreign Direct Investment on Domestic Savings and Investment in Select Developing Economies. *International Studies*, 39, 79-87.
- LAM, W. F. 1996. Improving the Performance of Small Scale Irrigation Systems: The Effects of Technological Investments and Governance Structure on Irrigation Performance in Nepal. World Development, 24, 1301-1315.
- LAN, J., KAKINAKA, M. & HUANG, X. 2012. Foreign Direct Investment, Human Capital and Environmental Pollution in China. *Environmental & Resource Economics*, 51.
- LANE, J.E. 2010. Development without Freedom? East and South East Asia: Vibrant Markets but a Rule of Law Deficit. *Asia Pacific Journal of Public Administration*, 32, 177-186.
- LARDY, N. R. 1995. The Role of Foreign Trade and Investment in China's Economic Transformation. *China Quarterly*.
- LARTEY, E. K. K. 2008. Capital Inflows, Resource Reallocation and the Real Exchange Rate. *International Finance*, 11, 131-152.
- LAURIDSEN, L. S. 2004. Foreign Direct Investment, Linkage Formation and Supplier Development in Thailand during the 1990s: The Role of State Governance. *European Journal of Development Research*, 16, 561-586.
- LAW, S. H. & AZMAN-SAINI, W. N. W. 2012. Institutional Quality, Governance, and Financial Development. *Economics of Governance*, 13, 217-236.

- LAW, S. H. & HABIBULLAH, M. S. 2006. Financial Development, Institutional Quality and Economic Performance in East Asian Economies. *Review of Applied Economics*, 2, 2, 201 – 216.
- LE, H. Q. & POMFRET, R. 2011. Technology Spill Overs from Foreign Direct Investment in Vietnam: Horizontal or Vertical Spill overs? *Journal of the Asia Pacific Economy*, 16, 183-201.
- LE, M. V. & SURUGA, T. 2005. Foreign Direct Investment, Public Expenditure and Economic Growth: The Empirical Evidence for the Period 1970-2001. *Applied Economics Letters*, 12, 45-49.
- LEAN, H. H. & TAN, B. W. 2011. Linkages between Foreign Direct Investment, Domestic Investment and Economic Growth in Malaysia. *Journal of Economic Cooperation and Development*, 32, 75-96.
- LEE, C.C. & CHANG, C.P. 2009. FDI, Financial Development, and Economic Growth: International Evidence. *Journal of Applied Economics*, 12, 249-271.
- LEE, C. G. 2009. Foreign Direct Investment, Pollution and Economic Growth: Evidence from Malaysia. *Applied Economics*, 41.
- LEE, C. H. 2002. The State and Institutions in East Asian Economic Development: The Past and the Future. *The Journal of the Korean Economy*, Vol. 3, No.1, pp. 1 17.
- LEE, H., LEE, J. & KIM, H.H. 2011. Foreign Direct Investment, Technology Diffusion, and Host Country Productivity Growth. Asian Development Bank Economics Working Paper Series No. 272.
- LEE, H. H. & TAN, H. B. 2006. Technology Transfer, FDI and Economic Growth in the ASEAN Region. *Journal of the Asia Pacific Economy*, 11, 394-411.
- LEE, J. 2009. Trade, FDI, and Productivity Convergence: A Dynamic Panel Data Approach in 25 Countries. *Japan and the World Economy*, 21, 226-238.
- LEE, J. R., CHEN, W. R. & KAO, C. 2003. Determinants and Performance Impact of Asymmetric Governance Structures in International Joint Ventures: An Empirical Investigation. *Journal of Business Research*, 56.
- LEE, J-Y. & MANSFIELD, E. 1996. Intellectual Property Protection and U.S. Foreign Direct Investment. *The Review of Economics and Statistics*, 78, 2, 181 – 186.
- LEE, S. C., LEE, K. B. & RYU, J. Y. 2003. Technology Transfer of Foreign Direct Investment in China. *Geography*, 88.
- LEE, Y.S. 2005. Foreign Direct Investment and Regional Trade Liberalization: A Viable Answer for Economic Development? *Journal of World Trade*, 39, 701-717.
- LEFF, N. 1964. Economic Development through Bureaucratic Corruption. American

Behavioural Scientist, 82, 337 – 41.

- LEFTWICH, A. 1993. Governance, Democracy and Development in the Third World. *Third World Quarterly*, 14, 605-624.
- LEMI, A. 2004. Foreign Direct Investment, Host Country Productivity and Export: The Case of U.S. and Japanese Multinational Affiliates. *Journal of Economic Development*, 29, 163-188.
- LENSINK, R. & MORRISSEY, O. 2006. Foreign Direct Investment: Flows, Volatility, and the Impact on Growth. *Review of International Economics*, 14, 478-494.
- LEUNG, C. K. 1996. Foreign Manufacturing Investment and Regional Industrial Growth in Guangdong Province, China. *Environment and Planning A*, 28.
- LEUNG, H. M., THOBURN, J. T., CHAU, E. & TANG, S. H. 1991. Contractual Relations, Foreign Direct Investment, and Technology Transfer - The Case of China. *Journal of International Development*, 3, 277-292.
- LEWER, J. J. & TERRY, N. 2003. Capital Account and Foreign Direct Investment Policies in the Late Nineties. *ASEAN Economic Bulletin*, 20, 256-271.
- LI, B. & YI, S. 2009. A VAR Model Analysis on China's FDI, Export and Economic Growth under Energy Constraint.
- LI, G. & HAN, W. 2010. The Empirical Analysis of the Reverse Spillover Effect of China's Technological Sourcing Foreign Direct Investment.
- LI, Q. 2006. Democracy, Autocracy, and Tax Incentives to Foreign Direct Investors: A Cross-National Analysis. *Journal of Politics*, 68.
- LI, Q. 2009. Democracy, Autocracy, and Expropriation of Foreign Direct Investment. *Comparative Political Studies*, 42.
- LI, Q. & RESNICK, A. 2003. Reversal of Fortunes: Democratic Institutions and Foreign Direct Inflows to Developing Countries. *International Organization*, 57, 175-211.
- LI, Q. & SHERALI, H. D. 2003. An Approach for Analyzing Foreign Direct Investment Projects with Application to China's Tumen River Area development. *Computers & Operations Research*, 30.
- LI, X. & FANG, C. 2008. Research on Technological Spill over Effects of MNCs' R&D Investment in China.
- LI, X. & LIU, X. 2005. Foreign Direct Investment and Economic Growth: An Increasingly Endogenous Relationship. *World Development*, 33, 393-408.
- LIANG, Y. 2007. Does Foreign Direct Investment Provide Desirable Development Finance? The Case of China. *China and World Economy*, 15, 104-120.
- LIANGSHU, Q. 2007. The Relationship between Growth, Total Investment and Inward FDI:

Evidence from Time Series Data. *International Review of Applied Economics*, 21, 119-133.

- LIEFNER, I. 2008. Foreign Direct Investment and Knowledge Transfer to China. *Geographische Rundschau*, 60, 4-11.
- LIEFNER, I. & WEI, Y. D. 2011. Geography and the Research on Contemporary China Introduction to the Special Issue Foreign Direct Investment, Innovation and Regional Economic Development in China. *Erdkunde*, 65.
- LIM, D. 1987. Export Instability, Investment and Economic Growth in Developing Countries. *Australian Economic Papers*, 26, 318-327.
- LIN, A. L. 1995. Trade Effects of Foreign Direct Investment: Evidence for Taiwan with Four ASEAN Countries. *Weltwirtschaftliches Archive-Review of World Economics*, 131.
- LIN, C.H., LEE, C.M. & YANG, C.H. 2011. Does Foreign Direct Investment Really Enhance China's Regional Productivity? *Journal of International Trade & Economic Development*, 20.
- LIN, P., LIU, Z. & ZHANG, Y. 2009. Do Chinese Domestic Firms Benefit from FDI Inflow?: Evidence of Horizontal and Vertical Spillovers. *China Economic Review (1043951X)*, 20, 677-691.
- LIN, Q., CHEN, G., DU, W. & NIU, H. 2012. Spillover Effect of Environmental Investment: Evidence from Panel Data at Provincial Level in China. *Frontiers of Environmental Science & Engineering*, 6.
- LINDA FUNG-YEE, N. & CHYAU, T. 2005. Industry Technology Performance of Manufacturing FDI: Micro-level Evidence from Joint Ventures in China. *International Journal of Technology Management*, 32, 246-263.
- LINDSEY, C. W. 1992. Foreign Direct Investment and Industrialization in Malaysia, Singapore, Taiwan, and Thailand. *Journal of Asian Studies*, 51.
- LIO, M. & LIU, M. C. 2008. Governance and Agricultural Productivity: A Cross National Analysis. *Food Policy*, 33, 504-512.
- LIPSEY, R. & SJOHOLM, F. 2011. Foreign Direct Investment and Growth in East Asia: Lessons for Indonesia. *Bulletin of Indonesian Economic Studies*, 47, 35-63.
- LIPSEY, R. & SJÖHOLM, F. 2011. The Role of South-South FDI in the Economies of Developing Asia. Asian Development Bank Economics Working Papers Series No. 273.
- LIU, H. & LU, J. 2010. The Home Employment Effect of Foreign Direct Investment: In the Case of China.
- LIU, M. 2009. Empirical Analysis on the Effects and Mechanisms of FDI on Development of China's Service Industry.

- LIU, S. 2005. Corporate Governance and Development: The Case of China. *Managerial and Decision Economics*, 26, 445-449.
- LIU, S. & SUN, Q. 2003. An Empirical Analysis of the Effect of FDI on China's Economic Growth.
- LIU, X. H., BURRIDGE, P. & SINCLAIR, P. J. N. 2002. Relationships between Economic Growth, Foreign Direct Investment and Trade: Evidence from China. *Applied Economics*, 34.
- LIU, X. M., WANG, C. G. & WEI, Y. Q. 2001. Causal Links between Foreign Direct Investment and Trade in China. *China Economic Review*, 12.
- LIU, X. Y. & WANG, P. Z. 2005. Trade Effects Induced by Taiwanese Investment in the Mainland China.
- LIU, Z. 2010. Researches on the Effect of Foreign Direct Investment in the Food Processing Industry in China.
- LIU, Z. Q. 2002. Foreign Direct Investment and Technology Spillover: Evidence from China. *Journal of Comparative Economics*, 30.
- LIU, Z. Q. 2003. The Economic Impact and Determinants of Investment in Human and Political Capital in China. *Economic Development and Cultural Change*, 51.
- LOMBARD, M. & LOMBARD, A. 2011. A Comparative Analysis of the Effects of Foreign Direct Investment on China's and India's Economic Development in Recent Years. *Economic Papers [Australia]*, 30, 522-529.
- LOPES, J., RUDDOCK, L. & RIBEIRO, F. L. 2002. Investment in Construction and Economic Growth in Developing Countries. *Building Research and Information*, 30.
- LOUIE, H. J. & ROUSSLANG, D. J. 2008. Host Country Governance, Tax Treaties and U.S. Direct Investment Abroad. *International Tax and Public Finance*, 15.
- LU, Y., YU, J., SUN, B. & QIN, X. 2011. Impacts of FDI Technology Spillover on Dynamic Changes of Export Structure in China: An Analysis Based on the Mechanism of Human Capital.
- LUCA, O. & SPATAFORA, N. 2012. Capital Inflows, Financial Development and Domestic Investment: Determinants and Inter Relationships. *IMF Working Paper Number*, 120.
- LUDDEN, D. 2005. A Usable Past for a Post National Present: Governance and Development in South Asia. *Journal of the Asiatic Society of Bangladesh*, 50, 259-292.
- LUI, F.T. 1985. An Equilibrium Queuing Model of Bribery. *Journal of Political Economy*, 93, 760 81.
- LUO, Q. 2002. Foreign Direct Investment in China: Determinants and Impact. China Quarterly.
- LV, L., WEN, S. & XIONG, Q. 2010. Determinants and Performance Index of Foreign Direct

Investment in China's Agriculture. China Agricultural Economic Review, 2.

- LÉVY, B. 2005. Global Competition and Economic Development: Key Governance Issues. *Competitiveness Review*, 15, 130-139.
- MA, Y. 2002. Foreign Direct Investment in China: Determinants and Impact. China Journal, 48.
- MACÍAS, C. M., OLIVIÉ, I. & PÉREZ, A. 2011. Opening the Black Box of FDI and Development. 19-22.
- MAH, J. S. 2010a. Foreign Direct Investment Inflows and Economic Growth of China. *Journal of Policy Modeling*, 32.
- MAH, J. S. 2010b. Foreign Direct Investment Inflows and Economic Growth: The Case of Korea. *Review of Development Economics*, 14, 726-736.
- MAHMOOD, A. 1997. The Role of Foreign Aid in Economic Development of Pakistan (1960-61 to 1994-95). *Pakistan Economic and Social Review*, 35, 57-90.
- MAHMUD, W., AHMED, S. & MAHAJAN, S. 2010. Economic Reforms, Growth, and Governance: The Political Economy Aspects of Bangladesh's Development Surprise.
- MAI, P. H. 2001. Regional Economic Development and Foreign Direct Investment Flows in Vietnam, 1988-98. *Journal of the Asia Pacific Economy*, 7, 182-203.
- MAITI, D. & MUKHERJEE, A. 2013. Governance, Foreign Direct Investment and Domestic Welfare. *University of Nottingham, GEP Discussion Papers*.
- MAJEED, M. T. & AHMAD, E. 2008. Human Capital Development and FDI in Developing Countries. *Journal of Economic Cooperation among Islamic Countries*, 29, 79-104.
- MAKKI, S. S. & SOMWARU, A. 2004. Impact of Foreign Direct Investment and Trade on Economic Growth: Evidence from Developing Countries. *American Journal of Agricultural Economics*, 86.
- MALIAR, L., MALIAR, S. & SEBASTIÁN, F. 2008. Sovereign Risk, FDI Spillovers, and Growth. *Review of International Economics*, 16, 463-478.
- MANACSA, R. C. & TAN, A. C. 2012. 'Strong Republic' Side Tracked: Oligarchic Dynamics, Democratization, and Economic Development in the Philippines. *Korea Observer*, 43, 47-88.
- MAPALAD, M. C. M. 1998. Foreign Capital Inflows and Domestic Savings in the Philippines. Savings and Development, XXII, 5-25.
- MARSH, I., BLONDEL, J., INOGUCHI, T. & COATE, R. A. 2005. Democratic Governance and Economic Performance: East and Southeast Asia. *Global Society*, 19, 445-455.
- MARSH, I., BLONDEL, J., INOGUCHI, T. & HILL, M. 2001. Democracy, Governance, and Economic Performance: East and Southeast Asia. *Singapore Journal of Tropical Geography*, 22, 311-312.

- MARSHALL, I., BLONDEL, J., INOGUCHI, T. & JAYASURIYA, K. 2001. Democracy, Governance and Economic Performance: East and Southeast Asia. *Australian Journal of Political Science*, 36, 396-397.
- MARTIN SCHINDLER, M. B., MICHAEL HUTCHISON. 2009. Controlling Capital? Legal Restrictions and the Asset Composition of International Financial Flows, USA, '*IMF eLibrary*'.
- MARWAH, K. & TAVAKOLI, A. 2004. The Effect of Foreign Capital and Imports on Economic Growth: Further Evidence from Four Asian Countries (1970-1998). *Journal of Asian Economics - International Journal*, 15, 399-414.
- MASTROMARCO, C. 2008. Foreign Capital and Efficiency in Developing Countries. *Bulletin* of Economic Research, 60, 351-374.
- MASTROMARCO, C. & GHOSH, S. 2009. Foreign Capital, Human Capital, and Efficiency: A Stochastic Frontier Analysis for Developing Countries. *World Development*, 37, 489-502.
- MASUYAMA, S. 2000. The Role of Japan's Direct Investment in Restoring East Asia's Dynamism. *Restoring East Asia's Dynamism*, 24.
- MATHUR, A. & SINGH, K. 2013. Foreign Direct Investment, Corruption and Democracy. *Applied Economics*, 45, 8, 991 – 1002.
- MATHUR, K. 1999. Strengthening Bureaucracy: State and Development in India. *Indian Social Science Review*, 1, 7-28.
- MAURO, P. 1995. Corruption and Growth. *The Quarterly Journal of Economics*, 110, 3, 681 712.
- MAYER, T. 2006. Policy Coherence for Development: A Background Paper on Foreign Direct Investment, OECD Publishing.
- MAYER, W. & MOURMOURAS, A. 2010. IMF Surveillance as a Signal to Attract Foreign Investment. *International Review of Economics & Finance*, 19.
- MCCLOUD, N. & KUMBHAKAR, S. C. 2012. Institutions, Foreign Direct Investment and Growth: A Hierarchical Bayesian Approach. *Journal of the Royal Statistical Society -Series A*, 175, 83-106.
- MEDVEDEV, D. 2012. Beyond Trade: The Impact of Preferential Trade Agreements on FDI Inflows. *World Development*, 40, 49-61.
- MEHMET, O. & TAVAKOLI, A. 2003. Does Foreign Direct Investment Cause a Race to the Bottom? *Journal of the Asia Pacific Economy*, 8, 133.
- MEHTA, P. B. 2010. India: Governance and Growth in State Capacity. *Governance*, 23, 381-384.
- MEI, L. & SUN, H. 2006. The Effect of FDI on Economic Growth in China.

- MEI, W. 2007. Foreign Direct Investment, Regional Market Conditions and Regional Development. *Economics of Transition*, 15, 125-151.
- MELEISEA, M. 2000. Governance, Development and Leadership in Polynesia: A Micro study from Samoa. *In:* HOOPER, A. (ed.) *Culture and Sustainable Development in the Pacific*.
- MELLO, J. L. R. D. 1997. Foreign Direct Investment in Developing Countries and Growth: A Selective Survey. *Journal of Development Studies*, 34, 1-34.
- MENCINGER, J. 2003. Does Foreign Direct Investment Always Enhance Economic Growth? *Kyklos*, 56, 4, 491 – 508.
- MENON, N. & SANYAL, P. 2007. Labour Conflict and Foreign Investments: An Analysis of FDI in India. *Review of Development Economics*, 11, 629-644.
- MEON, P.-G. & WEILL, L. 2005. Does Better Governance Foster Efficiency? An Aggregate Frontier Analysis. *Economics of Governance*, 6, 75-90.
- MEON, P.-G. & WEILL, L. 2010. Is Corruption an Efficient Grease? *World Development*, 38, 244-259.
- MERCEREAU, B. 2005. FDI Flows to Asia: Did the Dragon Crowd Out the Tigers?, USA, 'IMF eLibrary'.
- MESO, P., DATTA, P. & MBARIKA, V. 2006. Moderating Information and Communication Technologies' Influences on Socioeconomic Development with Good Governance: A Study of the Developing Countries. *Journal of the American Society for Information Science & Technology*, 57, 186-197.
- MESO, P., MUSA, P., STRAUB, D. & MBARIKA, V. 2009. Information Infrastructure, Governance, and Socio Economic Development in Developing Countries. *European Journal of Information Systems*, 18, 52-65.
- MEYER, K. E. 2003. FDI Spill overs in Emerging Markets: A Literature Review and New Perspectives. *Copenhagen Business School (Mimographed.)*.
- MEYER, K. E., ESTRIN, S., BHAUMIK, S. K. & PENG, M. W. 2009. Institutions, Resources, and Entry Strategies in Emerging Economies. *Strategic Management Journal*, 30.
- MICHE, J. 2002. Foreign Direct Investment and 'Human Capital Enhancement' in Developing Countries. *Competition & Change*, 6, 363.
- MIELNIK, O. & GOLDEMBERG, J. 2002. Foreign Direct Investment and Decoupling between Energy and Gross Domestic Product in Developing Countries. *Energy Policy*, 30.
- MILBERG, W. 1999. Foreign Direct Investment and Development: Balancing Costs and Benefits. *International Monetary and Financial Issues for the 1990s*, 11.
- MINH NGUYEN, T. B. & NURUL AMIN, A. T. M. 20q02. The Role of Foreign Direct Investment in Urban Environmental Management: Some Evidence from Hanoi, Vietnam.

Environment Development and Sustainability, 4, 279-298.

- MINQUAN, L., LUODAN, X. & LIU, L. 2004. Wage Related Labour Standards and FDI in China: Some Survey Findings from Guangdong Province. *Pacific Economic Review*, 9, 225-243.
- MIRZA, H. & FREEMAN, N. J. 2007. Foreign Direct Investment in East Asia's Transitional Economies: Perspectives on Development and Transition. *International Business Review*, 16.
- MIRZA, H. & GIROUD, A. 2004. Regionalization, Foreign Direct Investment and Poverty Reduction. *Journal of the Asia Pacific Economy*, 9, 223-248.
- MO, P. H. 2001. Corruption and Economic Growth. *Journal of Comparative Economics*, 29, 66 79.
- MODY, A. & MURSHID, A. P. 2005. Growing up with Capital Flows. *Journal of International Economics*, 65, 249-266.
- MOHAMMED, A.-M. & STROBL, E. 2011. Good Governance and Growth in Developing Countries: A Case Study of Regulatory Reforms in the Telecommunications Industry. *Journal of Industry, Competition & Trade*, 11, 91-107.
- MONDAL, P. 2007. Globalisation, Governance Reforms and Development in India. *Indian Journal of Social Work*, 68, 437-443.
- MONI, M. H. 2006. Japan-Bangladesh Economic Partnership. Focus on Aid, Trade and Investments. *International Studies*, 43, 395-410.
- MONTES, M. F. 1997. Direct Foreign Investment and Technology Transfer in ASEAN. ASEAN Economic Bulletin, 14, 176-189.
- MONTFORT MLACHILA, M. T. 2011. FDI from BRICs to LICs: Emerging Growth Driver?, USA, 'IMF eLibrary'.
- MORAN, T. 2002. The Relationship between Trade, Foreign Direct Investment, and Development: New Evidence, Strategy, and Tactics under the Doha Development Agenda Negotiations. *Asian Development Bank's Study on Regional Integration and Trade: Emerging Policy Issues for Selected Developing Member Countries, September.*
- MORAN, T. H. 1998. Foreign Direct Investment and Development: The New Policy Agenda for Developing Countries and Economies in Transition, *Peterson Institute*.
- MORRISSEY, O. & UDOMKERDMONGKOL, M. 2012. Governance, Private Investment and Foreign Direct Investment in Developing Countries. *World Development*, 40.
- MUELLER, D. C. 2006. Corporate Governance and Economic Performance. *International Review of Applied Economics*, 20, 623-643.
- MUHAMMAD, A., HASHIM, K., IMRAN, H. A., MUSHTAQ, A. H., & IRFAN, C. M. 2011.

Institutions, Macroeconomic Policy and Foreign Direct Investment: South Asian Countries Case. *Munich Personal RePEc Archive paper*, 32480.

- MUKHERJEE, A. & SUETRONG, K. 2009. Privatization, Strategic Foreign Direct Investment and Host Country Welfare. *European Economic Review*, 53, 775-785.
- MUKHERJEE, A., WANG, L. F. S. & TSAI, Y. 2012. Governance and Foreign Direct Investment: Is there a Two Way Relationship? *Trade and Development Review*, 5.
- MUKHERJI, R. 2008. The Politics of Telecommunications Regulation: State-Industry Alliance Favouring Foreign Investment in India. *Journal of Development Studies*, 44, 1405-1423.
- MULLEN, J. K. & WILLIAMS, M. 2005. Foreign Direct Investment and Regional Economic Performance. *Kyklos*, 58, 265-283.
- MYTELKA, L. K. & BARCLAY, L. A. 2004. Using Foreign Investment Strategically for Innovation. *European Journal of Development Research*, 16, 531-560.
- MZENDA, V. M. & BUYS, A. J. 2006. The Ambivalence of Technology Transfer through Foreign Direct Investments, 1827-1835.
- MÉNDEZ-PICAZO, M.T., GALINDO-MARTÍN, M.Á. & RIBEIRO-SORIANO, D. 2012. Governance, Entrepreneurship and Economic Growth. *Entrepreneurship and Regional Development*, 24, 865-877.
- MÉON, P.-G. & WEILL, L. 2010. Is Corruption an Efficient Grease? *World Development*, 38, 244-259.
- NAIR-REICHERT, U. & WEINHOLD, D. 2001. Causality Tests for Cross Country Panels: A New Look at FDI and Economic Growth in Developing Countries. *Oxford Bulletin of Economics and Statistics*, 63.
- NAURO F. CAMPOS, Y. K. 2003. Why Does FDI Go Where it goes? New Evidence from the Transition Economies, USA, 'IMF eLibrary'.
- NAYYAR, D. 1998. Economic Development and Political Democracy: Interaction of Economics and Politics in Independent India. *Economic and Political Weekly*, 33, 3121-3131.
- NEGARA, S. D. & ADAM, L. 2012. Foreign Direct Investment and Firms' Productivity Level. *ASEAN Economic Bulletin*, 29, 116-127.
- NESADURAI, H. E. S. 1999. Economic Developnt and Prospects in the ASEAN: Foreign Investment and Growth. *Pacific Review*, 12.
- NG, L. F.-Y. & TUAN, C. 2006. Spatial Agglomeration, FDI, and Regional Growth in China: Locality of Local and Foreign Manufacturing Investments. *Journal of Asian Economics -International Journal*, 17, 691-714.
- NG, L. F. Y. & TUAN, C. 2001. FDI Promotion Policy in China: Governance and Effectiveness. *World Economy*, 24.

- NGUYEN, T. T. & VAN DIJK, M. A. 2012. Corruption, Growth, and Governance: Private vs. State Owned Firms in Vietnam. *Journal of Banking & Finance*, 36, 2935-2948.
- NIGH, D. & SCHOLLHAMMER, H. 1987. Foreign Direct Investment, Political Conflict and Co-Operation: The Asymmetric Response Hypothesis. *Managerial and Decision Economics*, 8, 307-312.
- NONGKYNRIH, A. K. 2008. Who Decides? Who Implements? Political Structure, Governance and Rural Development in Meghalaya, India. *Indian Journal of Social Work*, 69, 301-319.
- NORTH, D. C. 1990. Institutions, Institutional Change and Economic Performance. *Cambridge University Press.*
- NUNNENKAMP, P. & SPATZ, J. 2004. Intellectual Property Rights and Foreign Direct Investment: A Disaggregated Analysis. *Review of World Economics*, 140, 3, 393 – 414.
- NUNNENKAMP, P. & STRACKE, R. 2008. Foreign Direct Investment inPost Reform India: Likely to Work Wonders for Regional Development? *Journal of Economic Development*, 33, 54-84.
- NWABUZOR, A. 2005. Corruption and Development: New Initiatives in Economic Openness and Strengthened Rule of Law. *Journal of Business Ethics*, 59, 121-138.
- OANA LUCA, N. S. 2012. Capital Inflows, Financial Development, and Domestic Investment: Determinants and Inter-Relationships, USA, '*IMF eLibrary*'.
- OFAIRCHEALLAIGH, C. 1985. Foreign Investment and Development in Less Developed Countries. *Inter-American Economic Affairs*, 39.
- OH, C. H. & OETZEL, J. 2011. Multinationals' Response to Major Disasters: How Does Subsidiary Investment Vary in Response to the Type of Disaster and the Quality of Country Governance? *Strategic Management Journal*, 32.
- OLIVA, M.-A. & RIVERA-BATIZ, L. A. 2002. Political Institutions, Capital Flows, and Developing Country Growth: An Empirical Investigation. *Review of Development Economics*, 6, 248.
- OLSON JR, M., SARNA, N. & SWAMY, A. V. 2000. Governance and Growth: A Simple Hypothesis Explaining Cross Country Differences in Productivity. *Public Choice*, 102, 341-364.
- ONYEMAECHI, U. 2001. Democracy, Governance, and Economic Performance: East and Southeast Asia. *Journal of Government Information*, 28, 594-596.
- OURA, H. Financial Development and Growth in India: A Growing Tiger in a Cage?, USA, 'IMF eLibrary'.
- OUYANG, P. & FU, S. 2012. Economic Growth, Local Industrial Development and Inter-

Regional Spillovers from Foreign Direct Investment: Evidence from China. *China Economic Review*, 23.

- PAJUNEN, K. 2008. Institutions and Inflows of Foreign Direct Investment: A Fuzzy Set Analysis. *Journal of International Business Studies*, 39.
- PAN, S. & LI, H. 2011. Impact of FDI on Economic Growth in China: an Empirical Analysis.
- PAPAGEORGIOU, C. & TURNBULL, G. K. 2005. Economic Development and Property Rights: Time Limits on Land Ownership. *Economic Development Quarterly*, 19, 271-283.
- PAREDES, T. A. 2005. Corporate Governance and Economic Development. *Regulation*, 28, 34-39.
- PARK, S. R. & LEE, X. 2005. Foreign Direct Investment and Technology Spillovers Empirical Evidences from China.
- PARK, Y. C. 1990. Development Lessons from Asia: The Role of Government in South. *American Economic Review*, 80, 118.
- PEERENBOOM, R. 2002. Social Networks, Rule of Law and Economic Growth in China: The Elusive Pursuit of the Right Combination of Private and Public Ordering. *Global Economic Review*, 31, 1-19.
- PERRITT, J. H. H. & CLARKE, R. R. 1998. Chinese Economic Development, Rule of Law, and the Internet. *Government Information Quarterly*, 15, 393.
- PERRY, A. 2000. Effective Legal Systems and Foreign Direct Investment: In Search of the Evidence. *International and Comparative Law Quarterly*, 49.
- PING, Y. U., CHEN, K. C. & XIAOJIN, S. U. N. 2010. Foreign Direct Investment and Economic Growth in China: Evidence from a Two Sector Model. *Journal of Financial Management & Analysis*, 23, 1-9.
- PINTO, M. R. 1992. Rural Development and Bureaucracy in India. *Indian Journal of Political Science*, LIII, 279-296.
- PINTO, S. 2004. Development without Institutions: Ersatz Medicine and the Politics of Everyday Life in Rural North India. *Cultural Anthropology*, 19, 337-364.
- POLPAT, K., BANGORN, T. U. & PAITOON, W. 2011. Does FDI Enhance Economic Growth? New Evidence from East Asia. *ASEAN Economic Bulletin*, 28, 183-202.
- POMFRET, R. 1999. Foreign Direct Investment and Economic Growth in China. *Journal of Comparative Economics*, 27.
- PRADHAN, J. P. 2003. Foreign Direct Investment and Economic Growth in Developing Countries: Further Evidence from Panel Data. *Asian Economic Review*, 45, 197-217.

PRASAD, B. C. 2008. Institutions, Good Governance and Economic Growth in the Pacific

Island Countries. International Journal of Social Economics, 35, 904-918.

- PRASAD, E. S., RAJAN, R. G. & SUBRAMANIAN, A. 2007. Foreign Capital and Economic Growth. *Brookings Papers on Economic Activity*, 153-209.
- PRASANNA, N. 2010. Impact of Foreign Direct Investment on Export Performance in India. Journal of Social Sciences, 24, 65-71.
- PRESBITERO, A. F. 2006. Institutions and Geography as Sources of Economic Development. Journal of International Development, 18, 351-378.
- QI, J., ZHENG, Y., LAURENCESON, J. & LI, H. 2009. Productivity Spillovers from FDI in China: Regional Differences and Threshold Effects. *China and World Economy*, 17, 18-36.
- QI, L. 2007. The Relationship between Growth, Total Investment and Inward FDI: Evidence from Time Series Data. *International Review of Applied Economics*, 21, 119-133.
- QIAN, X. SANDOVAL-HERNANDEZ, J. & GARRETT, J. Z. 2012. Corruption Distance and Foreign Direct Investment. *SSRN Working Paper Series*.
- QIN, D., CAGAS, M. A., QUISING, P. & HE, X.H. 2006. How Much Does Investment Drive Economic Growth in China? *Journal of Policy Modeling*, 28.
- QUADER, S. M. 2009. Foreign Direct Investment in Bangladesh: An Empirical Analysis on its Determinants and Impacts. *Asian Economic Review*, 52, 1-16.
- QUIBRIA, M. G. 2006. Does Governance Matter? Yes, No or Maybe: Some Evidence from Developing Asia. *Kyklos*, 59, 99-114.
- RAI, R. K. 2009. Effects of the TRIPS-Mandated Intellectual Property Rights on Foreign Direct Investment in Developing Countries: A Case Study of the Indian Pharmaceutical Industry. *The Journal of World Intellectual Property*, 11, 5/6, 404 – 431.
- RADELET, S., CLEMENS, M. & BHAVNANI, R. 2005. Aid and Growth. *Finance & Development*, 42, 16-20.
- RADHAN, J. P. 2002. Foreign Direct Investment and Economic Growth in India: A Production Function Analysis. *Indian Journal of Economics*, LXXXII, 581-586.
- RAMASAMY, B. & YEUNG, M. 2010. A Causality Analysis of the FDI Wages Productivity Nexus in China. *Journal of Chinese Economic and Foreign Trade Studies*, 3, 5-23.
- RAMMAL, H. G. & ZURBRUEGG, R. 2006. The Impact of Regulatory Quality on Intra Foreign Direct Investment Flows in the ASEAN Markets. *International Business Review*, 15.
- RATNASINGAM, J. & IORAS, F. 2009. Foreign Direct Investment (FDI), Added Value and Environmental Friendly Practices in Furniture Manufacturing: The Case of Malaysia and Vietnam. *International Forestry Review*, 11.

- RAWSKI, T. G. 2002. Will Investment Behaviour Constrain China's Growth? *China Economic Review*, 13.
- RAZIN, A. & SADKA, E. 2003. Gains from FDI Inflows with Incomplete Information. *Economics Letters*, 78, 71-78.
- REDEK, T. & SUSJAN, A. 2007. The Importance of Institutional Quality for Foreign Direct Investment in Emerging Economies: A Macro/Micro Perspective. *Economia Internazionale*, LX, 207-229.
- REIS, A. B. 2001. On the Welfare Effects of Foreign Investment. *Journal of International Economics*, 54, 411.
- REITER, S. L. & STEENSMA, H. K. 2010. Human Development and Foreign Direct Investment in Developing Countries: The Influence of FDI Policy and Corruption. *World Development*, 38, 1678-1692.
- RIZVI, S. Z. A. & NISHAT, M. 2009. The Impact of Foreign Direct Investment on Employment Opportunities: Panel Data Analysis: Empirical Evidence from Pakistan, India and China. *Pakistan Development Review*, 48, 841-852.
- ROBERTSON, G. B. & TEITELBAUM, E. 2011. Foreign Direct Investment, Regime Type, and Labor Protest in Developing Countries. *American Journal of Political Science*, 55.
- ROTHGEB, J. M. 1995. Investment Penetration, Agrarian Change, and Political Conflict in Developing Countries. *Studies in Comparative International Development*, 30.
- ROY, D. K. 2005. Governance and Development: The Challenges for Bangladesh. *Bangladesh Development Studies*, 31, 99-136.
- RUEDA-SABATER, E. J. 2000. Corporate Governance and the Bargaining Power of Developing Countries to Attract Foreign Investment. *Corporate Governance-an International Review*, 8.
- SAADI, M. 2011. Technology Transfer, Foreign Direct Investment, Licensing and the Developing Countries' Terms of Trade. *Margin: The Journal of Applied Economic Research*, 5, 381-420.
- SADIG, A. A. 2009. The Effects of Corruption on FDI Inflows. Cato Journal, 29, 2.
- SADOI, Y. 2008. Technology Transfer in Automotive Parts Firms in China. *Asia Pacific Business Review*, 14, 147-163.
- SAGGI, K. 2002. Trade, Foreign Direct Investment, and International Technology Transfer: A Survey. *The World Bank Research Observer*, 17, 191-235.
- SAHOO, D. & MATHIYAZHAGAN, M. K. 2003. Economic Growth in India: Does Foreign Direct Investment Inflow Matter? *Singapore Economic Review*, 48, 151-172.
- SAHOO, R. K. 1999. Foreign Direct Investment and Economic Development: A Firm Level

Analysis of Manufacturing Sector in Post-Reform India. *Asian Economic Review*, 41, 33-42.

- SAHOO, P. 2006. Foreign Direct Investment in South Asia: Policy, Trends, Impact and Determinants, *ADB Institute Discussion Paper* No.56.
- SAMUEL CHAN, Y.S. & JOHN LEE WAI, S. 2007. Tax Incentives in Hong Kong for Offshore Funds and Investment Schemes. *International Tax Journal*, 33, 13-43.
- SAPSFORD, D. 1999. Foreign Investment and Economic Development in China: 1979-1996. World Economy, 22.
- SARAVANAMUTTU, P. 2000. Sri Lanka in 1999 The Challenge of Peace, Governance, and Development. *Asian Survey*, 40, 219-225.
- SASIDHARAN, S. & KATHURIA, V. 2011. Foreign Direct Investment and R&D: Substitutes or Complements—A Case of Indian Manufacturing after 1991 Reforms. *World Development*, 39, 1226-1239.
- SCHAUMBURG-MÜLLER, H. 2003. Rise and Fall of Foreign Direct Investment in Vietnam and its Impact on Local Manufacturing Upgrading. *European Journal of Development Research*, 15, 44-66.
- SCHELKLE, W. 1999. Institutions and Economic Development. Growth and Governance in Less Developed and Post Socialist Countries. *Kyklos*, 52, 107-109.
- SEKIGUCHI, S. & LI, S. H. 2005. Foreign Direct Investment and Productivity in China. Journal of Tokyo Keizai University, 241, 189-204.
- SELDADYO, H., NUGROHO, E. P. & DE HAAN, J. 2007. Governance and Growth Revisited. *Kyklos*, 60, 279-290.
- SEMYONOV, M. & SHENHAV, Y. 1988. Investment Dependence, Economic Development, and Female Employment Opportunities in Less Developed Countries. *Social Science Quarterly*, 69.
- SERBU, S. 2006. FDI Role in Promoting the Economic Growth-A Problem Still Ambiguous. *Available at SSRN 962346*.
- SEYOUM, B. 1996. The Impact of Intellectual Property Rights on Foreign Direct Investment. *The Columbia Journal of World Business*, 31, 1, 50 – 59.
- SHABBIR, T., MAHMOOD, A. & NIAZI, S. A. 1992. The Effects of Foreign Private Investment on Economic Growth in Pakistan. *Pakistan Development Review*, 31, 831-831.
- SHAMSHAD, B., SIDDIQI, J. S. & SOCIETY, I. C. 2009. Impact of Foreign Direct Investment in Pakistan.
- SHARMA, B. & GANI, A. 2004. The Effects of Foreign Direct Investment on Human
Development. Global Economy Journal, 4.

SHARMA, S. D. 2007. Democracy, Good Governance, and Economic Development. *Taiwan Journal of Democracy*, 3, 29-62.

SHEN, K. R. & GENG, Q. 2001. FDI, Technology Spillovers, and Economic Growth in China.

- SHEN, W. 2011. Is Safe Safe Now?–Foreign Exchange Regulatory Control Over Chinese Outbound and Inbound Investments and a Political Economy Analysis of Policies.
 Foreign Exchange Regulatory Control over Chinese Outbound and Inbound Investments and a Political Economy Analysis of Policies. *Journal of World Investment & Trade,* Forthcoming.
- SHOHELER, M. & CHOWDHURY, R. 2006. The Impact of Foreign Aid on the Economic Growth of Bangladesh. *Asian Economic Review*, 48, 361-372.
- SHUJIE, Y. 2006. On Economic Growth, FDI and Exports in China. *Applied Economics*, 38, 339-351.
- SIDDHARTHAN, N. S. 2004. Business Environment, Investment Climate and FDI: Chinese and Indian Experiences. *Economic and Political Weekly*, 39, 3986-3988.
- SIMEON, H. 1992. Trade Strategy and the Dependency Hypothesis: A Comparison of Policy, Foreign Investment, and Economic Growth in Latin America and East Asia. *Economic Development and Cultural Change*, 40, 495-521.
- SINGH, A. K. & SINGH, U. B. 2004. Governance and Development Deprivation in India. *Indian Journal of Public Administration*, L, 277-284.
- SINGH, H. & JUN, K. W. 1995. Some New Evidence on Determinants of Foreign Direct Investment in Developing Countries. *Policy Research Working Paper*, 1531.
- SIT, V. F. S. 1993. Transnational Capital Flows, Foreign Investments, and Urban Growth in Developing Countries.
- SJOHOLM, F. 1999. Productivity Growth in Indonesia: The Role of Regional Characteristics and Direct Foreign Investment. *Economic Development and Cultural Change*, 47.
- SJOHOLM, F. 2002. The Challenge of Combining FDI and Regional Development in Indonesia. Journal of Contemporary Asia, 32.
- SMYTH, R. 2001. Foreign Investment and Economic Development in China: 1979-1996. Journal of Contemporary Asia, 31.
- SPAR, D. 1996. Trade, Investment, and Labor: The Case of Indonesia. Columbia Journal of World Business, 31.
- SPENCER, J. W. 2008. The Impact of Multinational Enterprise Strategy on Indigenous Enterprises: Horizontal Spillovers and Crowding Out in Developing Countries. Academy of Management Review, 33, 341-361.

- SRIDHAR, K. S. & REDDY, A. V. 2011. Investment and Economic Opportunities: Urbanization, Infrastructure and Governance in the North and South of India. *Asia-Pacific Development Journal*, 18, 1-46.
- SRIVASTAVA, S. 2006. The Role of Foreign Direct Investment in India's Services Exports: An Empirical Investigation. *Singapore Economic Review*, 51, 175-194.
- STANLEY, TD. 2001. Wheat from Chaff: Meta-Analysis as Quantitative Literature Review. *Journal of Economic Perspectives*, 15, 3, pp. 131-150.
- STANLEY, T. D & DOUCOULIAGOS, H. 2012. Meta-Regression Analysis in Economics and Business. *Oxford Routledge*.
- STANLEY, T, DOUCOULIAGOS, C, & JARRELL, S. 2008. Meta-Regression Analysis as the Socio Economics of Economics Research, *Journal of Socio-Economics*, 37, 1, pp. 276-292.
- STANLEY, T, DOUCOULIAGOS, H, GILES, M, HECKEMEYER, J, JOHNSTON, R, LAROCHE, P, NELSON, J, PALDAM, M, POOT, J, PUGH, G, ROSENBERGER, R, & ROST, K. 2013. Meta-Analysis of Economics Research Reporting Guidelines, *Journal of Economic Surveys*, 27, 2, pp. 390-394.
- STRANGE, R., FILATOTCHEV, I., LIEN, Y.C. & PIESSE, J. 2009. Insider Control and the FDI Location Decision Evidence from Firms Investing in an Emerging Market. *Management International Review*, 49.
- STRAUB, S. 2005. Opportunism, Corruption and the Multinational Firm's Mode of Entry. *Journal of International Economics*, 74.
- SUBRAMANIAN, A. 2007. The Evolution of Institutions in India and its Relationship with Economic Growth. *Oxford Review of Economic Policy*, 23, 196-220.
- SULLIVAN, J. D., ROGERS, J. & BETTCHER, K. E. 2007. The Importance of Property Rights to Development. SAIS Review, XXVII, 31-43.
- SUMNER, A. 2005. Is Foreign Direct Investment Good for the Poor? A Review and Stocktake. *Development in Practice*, 15, 269-286.
- SUMULONG, D. H. B. A. L. R. 2003. Foreign Direct Investment: The Role of Policy. Asian Development Bank Series No. 23.
- SUN, H. S. 1998. Macroeconomic Impact of Direct Foreign Investment in China: 1979-96. World Economy, 21.
- SUN, H. S. 2001. Foreign Direct Investment and Regional Export Performance in China. *Journal of Regional Science*, 41.
- SUN, H. S. & PARIKH, A. 2001. Exports, Inward Foreign Direct Investment (FDI) and Regional Economic Growth in China. *Regional Studies*, 35.

- SUN, M. & TAN, B. 2009. The Effects of FDI on Capital Formation in Central China-Analysis Based on Dynamic Panel Data with System GMM Estimation.
- SUN, S. 2010. Heterogeneity of FDI Export Spillovers and Its Policy Implications: The Experience of China. *Asian Economic Journal*, 24, 289-303.
- SUN, S. 2011. Foreign Direct Investment and Technology Spillovers in China's Manufacturing Sector. *Chinese Economy*, 44, 25-42.
- SUN, Y. 2010. What Matters for Industrial Innovation in China: R&D, Technology Transfer or Spillover Impacts from Foreign Investment? *International Journal of Business and Systems Research (IBSR)*, 4, 621-647.
- SUYANTO, BLOCH, H. & SALIM, R. A. 2012. Foreign Direct Investment Spillovers and Productivity Growth in Indonesian Garment and Electronics Manufacturing. *Journal of Development Studies*, 48, 1397-1411.
- SUYANTO, SALIM, R. A. & BLOCH, H. 2009. Does Foreign Direct Investment Lead to Productivity Spillovers? Firm Level Evidence from Indonesia. World Development, 37, 1861-1876.
- SYLWESTER, K. 2005. Foreign Direct Investment, Growth and Income Inequality in Less Developed Countries. *International Review of Applied Economics*, 19, 289-301.
- TAKII, S. 2011. Do FDI Spillovers Vary among Home Economies? Evidence from Indonesian Manufacturing. *Journal of Asian Economics*, 22, 152-163.
- TAN, Z. A. 2002. Product Cycle Theory and Telecommunications Industry Foreign Direct Investment, Government Policy, and Indigenous Manufacturing in China. *Telecommunications Policy*, 26.
- TANG, S., SELVANATHAN, E. A. & SELVANATHAN, S. 2008. Foreign Direct Investment, Domestic Investment and Economic Growth in China: A Time Series Analysis. World Economy, 31.
- TAUBE, M. 2005. Foreign Direct Investment: Boosting Economic Development in China. *Wirtschaftspolitische Blätter*, 52, 89-101.
- TE VELDE, D. & MORRISSEY, O. 2004. Foreign Direct Investment, Skills and Wage Inequality in East Asia. *Journal of the Asia Pacific Economy*, 9, 348-369.
- TEKIN, R. B. 2012. Economic Growth, Exports and Foreign Direct Investment in Least Developed Countries: A Panel Granger Causality Analysis. *Economic Modelling*, 29.
- TEKSOZ, A. U. 2004. Corruption and Foreign Direct Investment: An Empirical Analysis. *Working Paper*.
- THANGAVELU, S. M., WEI YONG, Y. & CHONGVILAIVAN, A. 2009. FDI, Growth and the Asian Financial Crisis: The Experience of Selected Asian Countries. *World Economy*, 32,

1461-1477.

- THEE, K. W. 2001. The Role of Foreign Direct Investment in Indonesia's Industrial Technology Development. *International Journal of Technology Management*, 22.
- THOMPSON, E. R. 2002. Clustering of Foreign Direct Investment and Enhanced Technology Transfer: Evidence from Hong Kong Garment Firms in China. *World Development*, 30.
- THOMSON, S. 1999. South East Asia: The Role of Foreign Direct Investment Policies In Development, *Working Paper Series, OECD*.
- THU THI, H., PAITOON, W. & BANGORN, T. 2010. Does Foreign Direct Investment Promote Economic Growth in Vietnam? *ASEAN Economic Bulletin*, 27, 295-311.
- THUN, E. 2006. Changing Lanes in China: Foreign Direct Investment, Local Governments, and Auto Sector Development, *Cambridge University Press*.
- THUY THU, N. & VAN DIJK, M. A. 2012. Corruption, Growth, and Governance: Private vs. State Owned Firms in Vietnam. *Journal of Banking & Finance*, 36, 2935-2948.
- TIAN, X. 2010. Managing FDI Technology Spillovers: A Challenge to TNCs in Emerging Markets. *Journal of World Business*, 45, 276-284.
- TIAN, X. W., LIN, S. L. & LO, V. I. 2004. Foreign Direct Investment and Economic Performance in Transition Economies: Evidence from China. *Post-Communist Economies*, 16.
- TIKKU, A. 1998. Indian Inflow: The Interplay of Foreign Investment and Intellectual Property. *Third World Quarterly*, 19, 87-113.
- TODO, Y. & MIYAMOTO, K. 2006. Knowledge Spillovers from Foreign Direct Investment and the Role of Local R&D Activities: Evidence from Indonesia. *Economic Development and Cultural Change*, 55.
- TODO, Y., ZHANG, W. & ZHOU, L.A. 2011. Intra-Industry Knowledge Spillovers from Foreign Direct Investment in Research and Development: Evidence from China's "Silicon Valley". *Review of Development Economics*, 15.
- TOMOHARA, A. & TAKII, S. 2011. Does Globalization Benefit Developing Countries? Effects of FDI on Local Wages. *Journal of Policy Modeling*, 33, 511-521.
- TOMOHARA, A. & YOKOTA, K. 2011. Foreign Direct Investment and Wage Inequality: Is Skill Upgrading the Culprit? *Applied Economics Letters*, 18, 773-781.
- TSO, A. Y. 1998. Foreign Direct Investment and China's Economic Development. *Issues & Studies*, 34.
- TUAN, C. & NG, L. F.Y. 2004. FDI and Industrial Restructuring in Post-WTO Greater PRD: Implications on Regional Growth in China. World Economy, 27, 1609-1630.
- TUAN, C. & NG, L. F.Y. 2007. The Place of FDI in China's Regional Economic Development:

Emergence of the Globalized Delta Economies. *Journal of Asian Economics -International Journal*, 18, 348-365.

- TUAN, C., NG, L. F. Y. & ZHAO, B. 2009. China's Post Economic Reform Growth: The Role of FDI and Productivity Progress. *Journal of Asian Economics - International Journal*, 20, 280-294.
- UDOMKERDMONGKOL, M. & MORRISSEY, O. 2008. Political Regime, Private Investment, and Foreign Direct Investment in Developing Countries, *Research Paper/UNU-WIDER*.
- UGUR, M. 2013. Corruption's Direct Effects on Per Capita Income Growth: A Meta-Analysis. Journal of Economic Surveys.
- UGUR, M. & DASGUPTA, N. 2011. Evidence on the Economic Growth Impacts of Corruption in Low Income Countries and Beyond: A Systematic Review. London: *EPPI-Centre*, *Social Science Research Unit, Institute of Education*, University of London.
- URATA, S., YUE, C. S., KIMURA, F. & YUEH, L. Y. 2009. Multinationals and Economic Growth in East Asia: Foreign Direct Investment, Corporate Strategies and National Development. *Economica [London]*, 76, 213-214.
- VACAFLORES, D. E. & MOGAB, J. 2012. Firm Specific Effects of FDI on Employment.
- VADLAMANNATI, K. C. & TAMAZIAN, A. 2009. Growth Effects of FDI in 80 Developing Economies: The Role of Policy Reforms and Institutional Constraints. *Journal of Economic Policy Reform*, 12, 299-322.
- VAZQUEZ, E. & VADLAMANNATI, K. C. 2009. Direct Foreign Investments in India: Potential & Performance of FDI Inflows from United States as Strategic Major Investor. *Applied Econometrics and International Development*, 9, 209-224.
- VELDE, D. W. T. & MORRISSEY, O. 2004. Foreign Direct Investment, Skills and Wage Inequality in East Asia. *Journal of the Asia Pacific Economy*, 9, 348-369.
- VITA, G. D. & KYAW, K. 2009. Growth Effects of FDI and Portfolio Investment Flows to Developing Countries: A Disaggregated Analysis by Income Levels. *Applied Economics Letters*, 16, 277-283.
- VORA-SITTHA, P. 2012. Governance and Poverty Reduction in Thailand. *Modern Economy*, 3, 487-497.
- VOYER, P. A. & BEAMISH, P. W. 2004. The Effect of Corruption on Japanese Foreign Direct Investment. *Journal of Business Ethics*, 50, 211-224.
- VU, T. B. 2008. Foreign Direct Investment and Endogenous Growth in Vietnam. *Applied Economics*, 40.
- VU, T. B., GANGNES, B. & NOY, I. 2008. Is Foreign Direct Investment Good for Growth? Evidence from Sectoral Analysis of China and Vietnam. *Journal of the Asia Pacific*

Economy, 13.

- VU, T. B. & NOY, I. 2009. Sectoral Analysis of Foreign Direct Investment and Growth in the Developed Countries. *Journal of International Financial Markets Institutions and Money*, 19, 402-414.
- WAHEED, A. 2004. Foreign Capital Inflows and Economic Growth of Developing Countries: A Critical Survey of Selected Empirical Studies. *Journal of Economic Cooperation Among Islamic Countries*, 25, 1-36.
- WANG, C. & YU, L. 2007. Do Spillover Benefits Grow with Rising Foreign Direct Investment?An Empirical Examination of the Case of China. *Applied Economics*, 39, 397-405.
- WANG, F., CHANG, C. & HE, D. 2009. An Empirical Study on the Impact of FDI on Economic Growth-the Case of Hubei Province in China.
- WANG, H. & LIU, X. 2005. Is Foreign Direct Investment Really an Effective Technology Transfer Thannel for Developing Countries?
- WANG, M. & SUNNYWONG, M. C. 2009. What Drives Economic Growth? The Case of Cross-Border M&A and Greenfield FDI Activities. *Kyklos*, 62, 316-331.
- WANG, M. & WONG, M. C. S. 2009. Foreign Direct Investment and Economic Growth: The Growth Accounting Perspective. *Economic Inquiry*, 47, 701-711.
- WANG, M. & WONG, M. C. S. 2010. FDI, Education, and Economic Growth: Quality Matters. *Atlantic Economic Journal*, 39, 103-116.
- WANG, X. 2009. Empirical Study on Economic Growth Effect of FDI in Services in China.
- WANG, X. 2012. Foreign Direct Investment and Innovation in China's E-commerce Sector. Journal of Asian Economics, 23, 288-301.
- WANG, X. & MENG, M. 2008. Analysis of the Effects of Foreign Capital Inflows on Investment in China.
- WANG, X., XU, L. C. & ZHU, T. 2011. Foreign Direct Investment under a Weak Rule of Law: Theory and Evidence from China 1. *Economics of Transition*, 20, 401-424.
- WANG, X. & YU, J. 2010. The Relationship of Foreign Direct Investment and Economic Growth of China in Post-financial Crisis Era.
- WANG, Y. 2007. On the Relationship between Overseas Direct Investment and Trade. *Chinese Economy*, 40, 56-69.
- WANG, Y. 2010. FDI and Productivity Growth: The Role of Inter Industry Linkages. Canadian Journal of Economics, 43, 1243-1273.
- WANG, Y. & YOU, J. 2012. Corruption and Firm Growth: Evidence from China. China Economic Review (1043951X), 23, 415-433.
- WANG, Z. 2010. An Empirical Research on Relationship between Foreign Direct Investment

and Technological Innovation Capability in Hubei Province of China.

- WEI, C., DENT, P. & ROBERTS, C. 2006. An Exploratory Analysis of Barriers to Investment and Market Maturity in Southeast Asian Cities. *Journal of Real Estate Portfolio Management*, 12, 49-57.
- WEI, K., YAO, S. & LIU, A. 2009. Foreign Direct Investment and Regional Inequality in China. *Review of Development Economics*, 13.
- WEI, K. C. J. & ZHANG, Y. 2008. Ownership Structure, Cash Flow, and Capital Investment: Evidence from East Asian Economies before the Financial Crisis. *Journal of Corporate Finance*, 14.
- WEI, S.J. 2000. Local Corruption and Global Capital Flows. Brookings Papers on Economic Activity, 303.
- WEI, S. J. 1996. Foreign Direct Investment in China: Sources and Consequences.
- WEI, Y. & LIU, X. 2006. Productivity Spillovers from R&D, Exports and FDI in China's Manufacturing Sector. *Journal of International Business Studies*, 37, 544-557.
- WEI, Y., LIU, X. & LUO, Q. 2002. Foreign Direct Investment in China: Determinants and Impact. *China Quarterly*, 171, 761-762.
- WEN, M. 2007. Foreign Direct Investment, Regional Market Conditions and Regional Development. *Economics of Transition*, 15, 125-151.
- WEN-JUAN, S. U. I., XIAO, L. I. U., BEI-YU, L. & ZHENG, W. 2010. Economic Growth Pattern in China from the Angle of Poverty and its Governance. *Geographical Research*, 29, 373-381.
- WERNICK, D. A. & HAAR, J. 2009. Do Governing Institutions Affect Foreign Direct Investment Inflows? New Evidence from Emerging Economies. International Journal of Economics and Business Research, 1, 3, 317 – 332.
- WHALLEY, J. & XIN, X. 2010. China's FDI and Non-FDI Economies and the Sustainability of Future High Chinese Growth. *China Economic Review*, 21, 123-135.
- WIGNARAJA, P. 2005. Pro-poor Growth and Governance in South Asia—Decentralisation and Participatory Development. *The Pakistan Development Review*, 44, 1159-1171.
- WILKIN, S. 2011. Can Bad Governance be Good for Development? Survival, 53, 61-76.
- WILLEM, T. AND SALIKE, N. 2013. Foreign Direct Investment in East Asia, RIETI Policy Discussion Series 13 – P – 003.
- WINDER, C. C. A. & LI, B. 1994. China's Economic Development and the Role of Foreign Investment. *Maandschrift Economie*, 58, 157-164.
- WOLFF, L.C. 2008. China's Private International Investment Law: One-Way Street into PRC Law? American Journal of Comparative Law, 56.

- WOMEN, U. N. D. F. T. A. O. & PROGRAM, J. I. L. 2006. FDI from Developing and Transition Economies: Implications for Development, *United Nations Publications*.
- WONG, K. N., TANG, T. C. & FAUSTEN, D. K. 2009. Foreign Direct Investment and Services Trade: Evidence from Malaysia and Singapore. *Global Economic Review*, 38.
- WONG, J., 1995. China's Economic Reform and Open-Door Policy Viewed from Southeast Asia. *ASEAN Economic Bulletin*, Vol. 11, No. 3, pp. 269 279.
- WOO, J. 2009. Productivity Growth and Technological Diffusion through Foreign Direct Investment. *Economic Inquiry*, 47, 226-248.
- WOO, J.Y. & HEO, U. 2009. Corruption and Foreign Direct Investment Attractiveness in Asia. Asian Politics and Policy, 1.
- WORLD BANK, 2013. East Asia and Pacific Data. Available online: www. Worldbank.org.
- WU, C. 2011. Economic Freedom, Economic Growth, and China. *Chinese Economy*, 44, 104-119.
- WU, X. 2001. The Impact of Foreign Direct Investment on the Relative Return to Skill. *Economics of Transition*, 9, 695-716.
- WU, X., HU, B. & WU, Z. 2005. Forecasting the Spillover Effect type of FDI in China from IS Investment Demand's Perspective.
- WU, X., LIU, X. & HUANG, Q. 2012. Impact of the Institutional Environment on the Choice of Entry Mode: Evidence from Chinese Enterprises. *China: An International Journal*, 10, 28-50.
- WU, Y. & PONFRET, R. 1999. Foreign Direct Investment and Economic Growth in China. Journal of Comparative Economics, 27, 797-798.
- WU, Y. & SONG, L. 2001. Foreign Direct Investment and Economic Growth in China. *China Journal*, 45, 178-179.
- WU, Y. R. 2000. Measuring the Performance of Foreign Direct Investment: A Case Study of China. *Economics Letters*, 66.
- XIAO, W., LIN, G.-B. & HE, J.S. 2009. FDI, Independent Innovation and Economic Growth: Evidences from the Interprovincial Experience of China.
- XIAOLAN, F. 2008. Foreign Direct Investment, Absorptive Capacity and Regional Innovation Capabilities: Evidence from China. *Oxford Development Studies*, 36, 89-110.
- XIAOWEN, T., SHUANGLIN, L. & VAI IO, L. 2004. Foreign Direct Investment and Economic Performance in Transition Economies: Evidence from China. *Post-Communist Economies*, 16, 497-510.
- XIAOYING, L. & XIAMING, L. 2005. Foreign Direct Investment and Economic Growth: An Increasingly Endogenous Relationship. World Development, 33, 393-407.

- XING, L. & CHAN, M. W. L. 2008. Relationship between Foreign Direct Investment, International Trade and Technology Progress: A Case from China.
- XU, G. & WANG, R. 2007. The Effect of Foreign Direct Investment on Domestic Capital Formation, Trade, and Economic Growth in a Transition Economy: Evidence from China. *Global Economy Journal*, 7.
- XU, X. & SHENG, Y. 2012. Productivity Spillovers from Foreign Direct Investment: Firm-Level Evidence from China. World Development, 40.
- XUAN, N. T. & XING, Y. 2008. Foreign Direct Investment and Exports The Experiences of Vietnam. *Economics of Transition*, 16.
- YABI, G. O. 2004. Does Direct Foreign Investment Really Drive Growth in Developing Countries? - Mitigated Results of an Empirical Analysis. *Canadian Journal of Development Studies-Revue Canadienne D Etudes Du Developpement*, 25.
- YACKEE, J. W. 2008. Bilateral Investment Treaties, Credible Commitment, and the Rule of (International) Law: Do BITs Promote Foreign Direct Investment? *Law & Society Review*, 42.
- YAN, L. 2007. Does Foreign Direct Investment Provide Desirable Development Finance? The Case of China. *China & World Economy*, 15.
- YANG, D. 2010. Foreign Direct Investment and Development of High-Growth Industrial Sectors in China, 1998-2006. *Chinese Economy*, 43, 93-114.
- YANG, J., XU, H., WANG, C., LAI, M. & WEI, Y. 2009. Productivity Spillovers from Foreign Direct Investment in Chinese Industries. *Journal of Chinese Economic and Business Studies*, 7, 429-446.
- YANG, L. & JIN, Y. 2009. The Impact of FDI on Self-Innovation Capability: A Comparison between Local and Foreign-Owned Firms in China.
- YANRUI, W. 2000. Measuring the Performance of Foreign Direct Investment: A Case Study of China. *Economics Letters*, 66, 143-150.
- YAO, S. 2006. On Economic Growth, FDI and Exports in China. *Applied Economics*, 38, 339-351.
- YAO, S. & WEI, K. 2007. Economic Growth in the Presence of FDI: The Perspective of Newly Industrialising Economies. *Journal of Comparative Economics*, 35, 211-234.
- YAPICI, N. 2010. Foreign Direct Investment and Spillover Effect: A Local Firm Perspective. International Journal of Business and Emerging Markets, 2, 425-442.
- YASMIN, B. 2005. Foreign Capital Inflows and Growth in Pakistan. A Simultaneous Equation Model. *South Asia Economic Journal*, 6, 207-219.
- YAXIONG, Z. & KUN, Z. 2007. Impact of Beijing Olympic-related Investments on Regional

Economic Growth of China: Interregional Input–Output Approach. *Asian Economic Journal*, 21, 261-282.

- YE, D., CHEN, Y. & YANG, W. 2011. The Determinants of Foreign Block Investment to China.
- YEUNG, H. W.C. 2000. Local Politics and Foreign Ventures in China's Transitional Economy: The Political Economy of Singaporean Investments in China. *Political Geography*, 19.
- YIN, F. 2011. Foreign Direct Investment in China's Service Industry: Effects and Determinants. *China-An International Journal*, 9.
- YIN, Z.C. & DING, R.J. 2009. Heterogeneous Human Capital, FDI Spillover and Energy Efficiency in China.
- YOUNG, S. & LAN, P. 1997. Technology Transfer to China through Foreign Direct Investment. *Regional Studies*, 31.
- YOUSAF, M. M., HUSSAIN, Z. & AHMAD, N. 2008. Economic Evaluation of Foreign Direct Investment in Pakistan. *Pakistan Economic and Social Review*, 46, 37-56.
- YOW, I. & MCKERN, B. 1990. Foreign Investment Economic Growth and Technology Transfer in the Republic of China. Stanford Business School.
- YU, K., XIN, X., GUO, P. & LIU, X. 2011. Foreign Direct Investment and China's Regional Income Inequality. *Economic Modelling*, 28.
- YU, L. 2005. The Influence of Technology Introduction by FDI to Technology Innovation in China & the Research on Suggestions.
- YU, M. & KONG, Q. 2008. The FDI and Independent Innovation of Small Enterprise of Chinese Industry Based on the Empirical Analysis of Eastern China.
- YU, Q. 1998. Capital Investment, International Trade and Economic Growth in China: Evidence in the 1980-90s. *China Economic Review*, 9.
- YUE, C. S. 1999. Trade, Foreign Direct Investment and Economic Development of Southeast Asia. *Pacific Review*, 12, 249-271.
- YUSOFF, M. B. 2011. Foreign Direct Investment, Exports, Education, and Growth in Malaysia.
- ZAMAN, K., KHAN, M. M. & AHMAD, M. 2012. The Relationship between Foreign Direct Investment and Pro Poor Growth Policies in Pakistan: The New Interface. *Economic Modelling*, 29.
- ZENG, Q. F., XU, Y. J. & CHEN, L. 2001. The Impact on the Economic Development in the Western of China by Foreign Direct Investment.
- ZENG, S. X., WAN, T. W. & TAM, V. W. Y. 2009. Towards FDI and Technology Spillover: A Case Study in China. *Transformations in Business and Economics*, 8, 50-62.
- ZHANG, J. & FU, X. 2008. FDI and Environmental Regulations in China. Journal of the Asia

Pacific Economy, 13, 3, 332 – 353.

- ZHANG, K. H. 1999. Foreign Direct Investment and Economic Growth: Evidence from Ten East Asian Economies. *Economia Internazionale*, LII, 517-536.
- ZHANG, K. H. 2001a. Does Foreign Direct Investment Promote Economic Growth? Evidence from East Asia and Latin America. *Contemporary Economic Policy*, 19.
- ZHANG, K. H. L. 2001b. How Does Foreign Direct Investment Affect Economic Growth in China? *Economics of Transition*, 9.
- ZHANG, L.Y. 2011a. Foreign Direct Investment and Urban Growth in China. China Quarterly.
- ZHANG, Q. & FELMINGHAM, B. 2002. The Role of FDI, Exports and Spillover Effects in the Regional Development of China. *Journal of Development Studies*, 38.
- ZHANG, T. 2004. Utilization of Foreign Capital and Economic Growth in China. *Chinese Economy*, 37, 62-84.
- ZHANG, X. 2006. Fiscal Decentralization and Political Centralization in China: Implications for Growth and Inequality. *Journal of Comparative Economics*, 34, 713-726.
- ZHANG, Y. 2011b. Institutions and International Investments: Evidence from China and other Emerging Markets. *Tjalling C. Koopmans Dissertation Series*.
- ZHAO, C. & DU, J. 2007. Causality between FDI and Economic Growth in China. *Chinese Economy*, 40, 68-82.
- ZHENG, Y. 2011. Credibility and Flexibility: Political Institutions, Governance, and Foreign Direct Investment. *International Interactions*, 37.
- ZHILONG, C., BERGER, Y. M., JIANYING, C., CHENGFU, L., WEIGUO, D., KANGFENG, Y., LIMING, P., DESHPANDE, A., FABRE, G. & SINGH, A. 2002. Foreign Investment and Economic Development in China. *China Report*, 38, 465-575.
- ZHONGXIU, Z. & KEVIN HONGLIN, Z. 2010. FDI and Industrial Productivity in China: Evidence from Panel Data in 2001–06. *Review of Development Economics*, 14, 656-665.
- ZHOU, B. 2010. FDI and Domestic Enterprise Innovation in Western China.
- ZHOU, J., XIAO, S.Y., CUI, S.C., FANG, G. & IEEE. 2010. Study on the Impact of Institutional Environment in Host Countries on the Outward FDI of China.
- ÉTIENNE, G. 2000. Crisis in Governance, a Review of Bangladesh's Development. *Tiers Monde*, XLI, 238.

APPENDIX

2.1 SEARCH KEY WORDS USED IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

2.2 PIOS FRAMEWORK USED IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Governance - Worldwide governance indicators OR Governance OR Voice and Accountability OR Political Stability and Absence of Violence OR Government Effectiveness OR Regulatory Quality OR Rule of Law OR Control of Corruption Inward Foreign direct investment - FDI or Foreign direct investment OR offshore investment OR cross boarder investment OR investment abroad OR overseas investment OR foreign assets OR Greenfield investment OR foreign investment OR foreign ventures OR foreign reinvestment OR foreign assets OR non-local investments OR international investment OR outside investment OR non-native investment OR remote investment OR non-domestic investment OR non-resident investment OR distant investment OR investment OR invest OR inflows OR direct investment OR investment in other countries South and East Asia & Pacific countries - Emerging economies OR East Asian economies OR South east Asian economies OR East Asia OR South Asia OR South east Asia OR Afghanistan OR Bangladesh OR Bhutan OR India OR Maldives OR Nepal OR Pakistan OR Sri Lanka OR American Samoa OR Cambodia OR China OR Fiji OR Indonesia OR Kiribati OR Korea, Dem. Rep. OR Lao PDR OR Malaysia OR Marshall Islands OR Micronesia, Fed. Sts OR Mongolia OR Myanmar OR Palau OR Papua New Guinea OR Philippines OR Samoa OR Solomon Islands OR Thailand OR Timor-Leste OR Tuvalu OR Tonga OR Vanuatu OR Vietnam OR Asean OR Developing economies OR Developing countries

PIOS framework

Population – The study should focus on South and East Asia Pacific economies or equivalent as specified in the search criteria.

Independent variable - The study should be examining the impact of measures economic governance in terms of a scale or its equivalent as specified in the search criteria.

Outcome variable - The study should be examining inward foreign direct investment or as defined in the search criteria.

Study design - Study design can be either theoretical or empirical. A study is considered to be theoretical if it is based on some theoretical model drawing verbal or mathematical conclusions analysing impact of economic governance on economic growth. A study is considered to be

empirical if it is based on regression model and draws an estimation model to estimate economic governance on economic growth.

2.3 STUDIES SATISFYING PIOS CRITERIA IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Authors (Year)	Population	Independent	Outcome	Study
		Variable	Variable	Design
1999	Т			
2000	Т			
Adeoye, A (2007)	Y	Y	Y	Y
Ahlquist, J. S. (2006)	Y	Y	Y	Y
Akisik, O. & Pfeiffer, R. (2009)	Y	N	Y	Y
Alfaro, L. & Charlton, A. (2009)	Т			
Ali, F. A., Fiess, N. & Macdonald, R (2010)	Y	Y	Y	Y
Amaro, A. & Miles, W. (2006)	Y	N	Y	Y
Anghel, B. (2004)	Y	Y	Y	Y
Anwar, S. & Cooray, A. (2012)	Y	Y	Ν	Y
Arbatli, E. (2011)	Y	Y	Y	Y
Azam, M., Khan, H., Hunjra, A. I., Ahmad,	Y	Y	Y	Y
H. M. & Chani, M. I. (2011)				
Azhar, M. (2011)	Т			
Azman-Saini, W. N. W., Baharumshah, A. Z.	Y	N	Ν	Y
& Law, S. H. (2010)				
Azémar, C. & Desbordes, R. (2010)	Y	Y	Ν	Y
Beladi, H., Marji, S. & Chakrabarti, A. (2009)	Т			
Beyer, J. (2002)	N	Y	Y	Y
Blanton, S. L. & Blanton, R. G. (2007)	Y	Y	Y	Y
Branstetter, L. G. & Feenstra, R. C. (2002)	Y	N	Ν	Y
Brooks, D. H. & Hill, H. (2004)	Т			
Brouthers, L. E., Gao, Y. & Jason Patrick	Y	N	Y	Y
Mcnicol (2008)				
Busse, M., (2004)	Y	Y	Y	Y
Busse, M., & Hefeker, C, (2005)	Y	Y	Y	Y

Busse, M., Nunnenkamp, P. & Spatareanu,	Y	Y	Y	Y
M. (2011)				
Campos, N. F. & Kinoshita, Y. (2010)	Y	N	Y	Y
Chang, Y. C., Kao, M. S., Kuo, A. & Chiu, C.	Y	Y	N	Y
F. (2012)				
Chen, YF. & Funke, M (2011)	Т			
Chen, YR., Yang, C., Hsu, SM. & Wang,	Y	Y	N	Y
Y.D (2009)				
Cheng Hsiao & Yan Shen (2003)	Y	N	Y	Y
Choi, SW. & Samy, Y (2008)	Y	Y	Y	Y
Co, C. Y., List, J. A. & Qui, L. D (2004)	Y	N	Y	Y
Cole, M. A., Elliott, R. J. R. & Jing, Z (2009)	Y	N	Y	Y
Cyrus, T. L., Iscan, T. B. & Starky, S (2006)	Y	N	Y	Y
Davis, G. D (2011)	Y	Y	Y	Y
Doces, J. A (2010)	Y	Y	Y	Y
Driffield, N., Jones, C. & Crotty, J (2012)	Y	N	Y	Y
Du, J., Lu, Y. & Tao, Z (2008)	Y	Y	N	Y
Egger, P. & Radulescu, D. M (2011)	Y	Y	N	Y
Ellingsen, T. & Wärneryd, K (1999)	Т			
Elo, K. Z (2007)	Y	Y	Y	Y
Escribano, A., Guasch, J. L., Orte, M. D. &	Y	Y	N	Y
Pena, J (2009)				
Fan, J. P. H., Morck, R., Xu, L. C. & Yeung,	Y	Y	Y	Y
B (2009)				
Feils, D. J. & Rahman, M (2011)	Ν	Y	Y	Y
Floyd, D. & Summan, S (2008)	Т			
Foster, M. J (2011)	Т			
Fung, HG., Zhang, K. H., Leung, W. K., Lo,	Т			
W. C., Chan, K. C., Cheng, L. T. W., Fung, J.				
K. W., Xu, X. E. & Pei, C (2001)				
Gani, A (2007)	Y	Y	Y	Y
Gastanaga, V. M., Nugent, J. B. &	Y	Y	Y	Y
Pashamova, B (1998)				
Ghosh, D. N (1992)	Т			

Ghosh, D. N (2005)	Т			
Globerman, S. & Shapiro, D (2002a)	Y	Y	Y	Y
Globerman, S. & Shapiro, D (2002b)	Y	Y	Y	Y
Globerman, S. & Shapiro, D (2004)	Y	Y	Y	Y
Golub, S. S (2009)	Т			
Goodspeed, T., Martinez-Vazquez, J. &	Y	Y	Y	Y
Zhang, L (2011)				
Gordon, L. A., Loeb, M. P. & Zhu, W (2012)	Y	Y	Y	Y
Guerin, S. S. & Manzocchi, S (2009)	Y	Y	Y	Y
Habib, M. & Zurawicki, L (2001)	Y	Y	Y	Y
Harms, P. & Lutz, M (2006)	Y	Ν	Y	Y
He, C., Wang, J. & Cheng, S (2011)	Y	N	Y	Y
He, C. & Zhu, Y (2010)	Y	N	Y	Y
Hsiao, C. & Shen, Y. (2003)	Y	Y	Y	Y
Hur, J., Parinduri, R. A. & Riyanto, Y. E	Y	Y	Y	Y
(2007)				
Ihara, R. & Iwahashi, R (2007)	Т			
Inoguchi, M (2009)	Y	Y	No	Y
J. Saúl Lizondo, D. J. M (1990)	Т			
Jarvis, D. S (2012)	Y	Y	Y	No
Jensen, N (2008)	Y	No	No	Y
Jensen, N. & Mcgillivray, F (2005)	Y	Y	Y	Y
Jensen, N. M (2003)	Y	Y	Y	Y
Jeong-Yeon Lee & Mansfield, E (1996)	Y	No	Y	Y
Jing, Z. & Xiaolan, F (2008)	Y	No	Y	Y
Kessing, S. G., Konrad, K. A. &	Y	No	Y	Y
Kotsogiannis, C (2007)				
Khamfula, Y (2007)	Y	Y	Y	Y
Kumar, N (2005)	Т			
Kwok, C. C. Y. & Solomon Tadesse (2006)	Y	No	No	Y
Lauridsen, L. S (2004)	Т			
Lee, J. R., Chen, W. R. & Kao, C (2003)	Y	No	No	Y
Lewer, J. J. & Terry, N (2003)	Y	No	No	Y
Li, Q (2006)	Y	Y	No	Y

Li, Q (2009)	Y	Y	No	Y
Li, Q. & Resnick, A (2003)	Y	Y	Y	Y
Luca, O. & Spatafora, N. (2012)	Y	Y	Y	Y
Maiti, D. & Mukherjee, A	Т			
Martin Schindler, M. B., Michael Hutchison	Y	No	Y	Y
Mayer, T. (2006)	Y	No	Y	Y
Mayer, W. & Mourmouras, A (2010)	Т			
Mccloud, N. & Kumbhakar, S. C (2012)	Y	No	No	Y
Medvedev, D (2012)	Y	Y	Y	Y
Menon, N. & Sanyal, P (2007)	Y	No	Y	Y
Meyer, K. E., Estrin, S., Bhaumik, S. K. &	Y	Y	No	Y
Peng, M. W (2009)				
Minquan, L., Luodan, X. & Liu, L (2004)	Y	No	No	Y
Moni, M. H (2006)	Y	No	Y	No
Morrissey, O. & Udomkerdmongkol, M	Y	No	No	Y
(2012)				
Mukherjee, A., Wang, L. F. S. & Tsai, Y	Т			
(2012)				
Mukherji, R (2008)	Т			
Nauro F. Campos, Y. K	No	Y	Y	Y
Ng, L. F. Y. & Tuan, C (2001)	Т			
Nigh, D. & Schollhammer, H (1987)	Y	No	Y	Y
Oana Luca, N. S	Y	Y	No	Y
Oh, C. H. & Oetzel, J (2011)	Y	Y	No	Y
Pajunen, K (2008)	Y	Y	Y	No
Perry, A (2000)	Т			
Rammal, H. G. & Zurbruegg, R (2006)	Y	Y	No	Y
Redek, T. & Susjan, A (2007)	Т			
Reiter, S. L. & Steensma, H. K (2010)	Y	No	No	Y
Rueda-Sabater, E. J (2000)	Т			
Samuel Chan, YS. & John Lee Wai, S	Т			
(2007)				
Shen, W (2011)	Т			
Singh, H. & Jun, K. W. 1999	Y	Y	Y	Y

Sridhar, K. S. & Reddy, A. V. (2011)	Т			
Strange, R., Filatotchev, I., Lien, YC. &	Y	No	No	Y
Piesse, J. (2009)				
Straub, S. (2008)	Y	No	Y	Y
Wernick, D.A., Haar, J. & Singh, J (2009)	Y	Y	Y	Y
Tan, Z. A (2002)	Т			
Teksoz, S. U.2004	Y	Y	Y	Y
Udomkerdmongkol, M. & Morrissey, O	Y	Y	No	Y
(2008)				
Voyer, P. A. & Beamish, P. W (2004)	Y	Y	Y	Y
Wang, X., Xu, L. C. & Zhu, T (2012)	Y	No	Y	Y
Wei, C., Dent, P. & Roberts, C (2006)	Y	Y	No	Y
Wei, K. C. J. & Zhang, Y (2008)	Y	No	Y	Y
Wei, SJ (2000)	Y	Y	Y	Y
Wolff, LC (2008)				
Woo, JY. & Heo, U (2009)	Y	Y	Y	Y
Wu, X., Liu, X. & Huang, Q (2012)	Y	Y	No	Y
Yackee, J. W (2008)	Y	Y	Y	Y
Ye, D., Chen, Y. & Yang, W (2011)				
Yeung, H. WC (2000)	Т			
Zhang, Y (2011)	Т			
Zheng, Y (2011)	Y	Y	Y	Y
Zhou, J., Xiao, SY., Cui, SC., Fang, G. &	Y	No	No	Y
Ieee (2010)				

Y= Study satisfies the criteria; N= Study does not satisfy the criteria

2.4 NUMBER OF STUDIES SATISFYING PIOS CRITERIA IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Criteria	Number of studies
	satisfying the
	criteria
Population (South and East Asia & Pacific countries)	94
Independent variable (Measures of governance)	62
Outcome variable (Inward foreign direct investments)	68

Study design – Empirical	94
Decision Select if all 4 criteria match - PIOS	
Select for next stage	40
Deselect studies	91

2.5 COMPOSITION OF PUBLISHED (69%) AND UNPUBLISHED (31%) INCLUDED IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

												■ N	lo of	fstu	idies	5											
15																											
	4	2	2	2	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
Grey literature	World	International	Applied	International	The journal of	Journal of	Polity	International	Politics and	Foreign policy	Review of	Journal of	Brookings	Journal of	World economy	Open	International	Asian politics	Law and	Journal of	Economic	International	KAKLOS	Economic inquiry	Procedia	Applied	Cato journal

2.6 DESCRIPTIVE STATISTICS OF MODERATOR VARIABLES INCLUDED IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Moderator	Definition	Mean	Standard
variable			deviation
Ptype1	=1 if the estimate is from an article published in	0.544	0.50
	journal; = 0 otherwise		
Ptype2	=1 if the estimate is from unpublished study; = 0	0.456	0.50
	otherwise		
Specific fdi	=1 if the model uses FDI data on single country;	0.020	0.14
	= 0 otherwise		
Nonspecificfdi	=1 if the model uses FDI data on more than one	0.980	0.14
	country FDI; $= 0$ otherwise		
Yearly	=1 if the model uses yearly data on FDI; = 0	0.526	0.50
	otherwise		

Nonyearly	=1 if the model uses non-yearly data on FDI; = 0	0.474	0.50
	otherwise		
Data1	=1 if the model uses panel data; = 0 otherwise	0.579	0.49
Data2	=1 if the model uses cross sectional data; = 0	0.421	0.49
	otherwise		
Fdi1	=1 if the model uses levels of FDI; = 0 otherwise	0.119	0.32
Fdi2	=1 if the model uses relative figures of FDI; = 0	0.092	0.29
	otherwise		
Fdi3	=1 if the model uses natural logarithm of FDI; =	0.788	0.41
	0 otherwise		
Country1	=1 if the estimate belongs to South Asia; = 0	0.007	0.08
	otherwise		
Country2	=1 if the estimate belongs to Mixed countries; =	0.993	0.08
	0 otherwise		
Method1	=1 if the model is estimated using OLS	0.417	0.49
	technique; = 0 otherwise		
Method2	=1 if the model is estimated using panel data	0.377	0.48
	technique; = 0 otherwise		
Method3	=1 if the model is estimated using instrumental	0.132	0.34
	variable technique; = 0 otherwise		
Method4	=1 if the model is estimated using time series	0.073	0.26
	technique; = 0 otherwise		
Method5	=1 if the model is estimated using other	0.001	0.34
	technique; = 0 otherwise		
Lauthor1	=1 if the first author of the study is American; =	0.462	0.50
	0 otherwise		
Lauthor2	=1 if the first author of the study is European; = 0	0.307	0.46
	otherwise		
Lauthor3	=1 if the first author of the study is South & East	0.047	0.21
	Asian; = 0 otherwise		
Lauthor4	=1 if the first author of the study is from other	0.184	0.39
	region; = 0 otherwise		
Subject1	=1 if the estimate is taken form a study that	0.551	0.50
	belongs to Economics and Finance discipline; $= 0$		
	otherwise		

Subject2	=1 if the estimate is taken form a study that	0.161	0.37
	belongs to Business Management and		
	Accounting discipline; $= 0$ otherwise		
Subject3	=1 if the estimate is taken form a study that	0.208	0.41
	belongs to Policy discipline; $= 0$ otherwise		
Subject4	=1 if the estimate is taken form a study that	0.069	0.25
	belongs to Development discipline; = 0 otherwise		
Subject5	=1 if the estimate is taken form a study that	0.011	0.11
	belongs to Law discipline; $= 0$ otherwise		
Dumchi1	=1 if the model includes China in the sample	0.975	0.16
	countries; = 0 otherwise		
Dumchi2	=1 if the model excludes China from the sample	0.025	0.16
	countries; = 0 otherwise		
Dumsk1	=1 if the model includes South Korea in the	0.849	0.36
	sample countries; $= 0$ otherwise		
Dumsk2	=1 if the model excludes South Korea from the	0.151	0.36
	sample countries; $= 0$ otherwise		
Form1	=1 if the model uses merger and acquisition form	0.089	0.28
	of FDI; = 0 otherwise		
Form2	=1 if the model uses aggregate FDI; = 0	0.911	0.28
	otherwise		
Flow1	=1 if the model uses stock of FDI; = 0 otherwise	0.048	0.21
Flow2	=1 if the model uses flow of FDI; = 0 otherwise	0.952	0.21
Indi1	=1 if the model includes governance as main	0.964	0.19
	independent variable; $= 0$ otherwise		
Indi2	=1 if the model includes governance as control	0.036	0.19
	variable; = 0 otherwise		
Dosurce1	=1 if model uses data on FDI from IMF database;	0.037	0.19
	= 0 otherwise		
Dsource2	=1 if model uses data on FDI from OECD	0.054	0.23
	database; = 0 otherwise		
Dsource3	=1 if model uses data on FDI from other	0.221	0.41
	databases; $= 0$ otherwise		
Dsource4	=1 if model uses data on FDI from UNCTAD	0.189	0.39
	database; = 0 otherwise		

Dsource5	=1 if model uses data on FDI from World Bank	0.499	0.50
	database; = 0 otherwise		
Idsource1	=1 if the data on governance measure in the	0.021	0.14
	model is taken from BERI database; $= 0$		
	otherwise		
Idsource2	=1 if the data on governance measure in the	0.037	0.19
	model is taken from Freedom House database; =		
	0 otherwise		
Idsource3	=1 if the data on governance measure in the	0.242	0.43
	model is taken from ICRG database; $= 0$		
	otherwise		
Idsource4	=1 if the data on governance measure in the	0.193	0.39
	model is taken from other sources; $= 0$ otherwise		
Idsource5	=1 if the data on governance measure in the	0.029	0.17
	model is taken from PRS database; $= 0$ otherwise		
Idsource6	=1 if the data on governance measure in the	0.120	0.33
	model is taken from Polity database; $= 0$		
	otherwise		
Idsource7	=1 if the data on governance measure in the	0.042	0.20
	model is taken from Transparency International		
	database; = 0 otherwise		
Idsource8	=1 if the data on governance measure in the	0.315	0.46
	model is taken from World Wide Governance		
	Indicators from World Bank database; $= 0$		
	otherwise		

2.7 SUMMARIES OF EMPIRICAL STUDIES INCLUDED IN GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Authors and	Sample	Study	Dependent	Independent	Methodolo	Findings
year	size	period	variable and	variable and	gу	
			source	source		

Castanaga	40 1000	1070	Aggragata	Various	Deeled	Duraquaraa
Gastallaga	49 1888	1970	Aggregate	various	Fooled	Dureaucrac
et al.,	develope	_	inward FDI in	institutional	cross	y – negative
(1998)	d	1995	millions of US	variables –	section	and
	countries		dollars (taken as	bureaucracy	and time	significant
			FDI to GDP	and corruption	series data	
			ratio)	Source:		Corruption
			Source:	Various		– positive
			International	sources		and
			Monetary			significant
			Fund's (IMF)			
			Balance of			
			Payments			
			Statistics			
			Yearbook			
Globerman	115	1995	US FDI	World	Cross	Law –
and Shapiro	developi	_	Source: Bureau	governance	sectional	positive and
(2002a)	ng and	1997	of Economic	indicators	data	insignificant
	develope		Analysis	Source: World		
	d		-	Bank		Voice and
	countries		(both aggregate	(Kaufman et.		accountabili
			FDI flows and	Al. (1999))		ty – positive
			industry specific			and
			(2 high			significant
			technology			e
			industries))			Political
						instability –
						nositive and
						insignificant
						msignificant
						Government
						offoctivonos
						s – positive
						and
						significant

						Regulation
						– positive
						and
						significant
						Corruption
						– positive
						and
						significant
		1007			~	~
Globerman	114	1995	Net inward FDI	World	Cross	Governance
and Shapiro	developi	-	(=inward FDI –	governance	sectional	has positive
(2002b)	ng and	1997	FDI outflows)	indicators		and
	develope		averaged 1995 –	Source: World		significant
	d		97.	Bank		effect on
	countries		Source: The			FDI
			world			
			investment			
			report,			
			UNCTAD			
			(1998) Annex B			
Hsiao and	23	1976	Total inward	Governance	Panel data	Absence of
Shen (2003)	developi	_	FDI flows as	institutions		corruption –
	ng	1997	percentage of			positive and
	countries		gross domestic			insignificant
			product (GDP)			
			(in percentage			
			values).			
			Source: World			
			Development			
			Indicator CD			
			Rom (2000)			
Anghel	80	1996	Net FDI as a	Governance	Cross	Political
(2004)	countries	-	percentage of	institutions (5	sectional	stability –
		2000	average GDP	indicators are	data	

			Source: World	used		positive and
			Bank	government		significant
				effectiveness,		
				regulatory		Government
				quality, rule of		effectivenes
				law and		s – positive
				control of		and
				corruption)		significant
				Source: World		Rule of law
				Bank		– positive
				governance		and
				indicators		significant
				(Kaufman et.		
				al. 2004)		Control of
						corruption –
						positive and
						significant
Globerman	154	1995	Merger and	Governance	Panel data	Governance
and Shapiro	countries	-	Acquisition	indicators.		– positive
(2004)		2001	inflows.			and
				Source: World		significant
			Source:	Bank,		
			UNCTAD	Kaufmann et		
				al. (2003).		
Gani (2007)	17	4	FDI as a share	Governance	Panel data	Rule of law
	countries	period	of GDP	indicators.		– positive
	from	s –	Source: World			and
	Asia and	1996,	Bank (2004)	Source: World		significant
	Latin	1998,		Bank,		
	America	2002,		Kaufmann et		Control of
		2004		al. (2003).		corruption –
						positive and
						positive and
						significant

Hur et al.,	172	1995	Merger and	Governance	Panel data	Governance
(2007)	countries	-	Acquisition	indicators.		– positive
		2002	flows			and
			Source:	Source: World		significant
			UNCTAD	Bank,		
				Kaufmann et		
				al. (2003).		
Adeoye	33	1997	Inwards FDI as	Governance	Panel data	Governance
(2009)	emerging	_	% of GDP	indicators.		– positive
	countries	2002	Source: World			and
			Bank	Source: World		significant
				Bank,		
				Kaufmann et		
				al. (2003).		
Wernick et	64	1996	Inward FDI	Overall	Panel data	Governance
al., (2009)	emerging	_	measured in	governance	OLS	– Positive
	economi	2006	millions of US	Source: World	technique	and
	es		dollars	Bank,		significant
			Source: World	Kaufmann et		
			Bank	al. (2003).		
Ali et	69	1981 -	FDI net inflows	Institutional	Panel data	Governance
al.,(2010)	countries	2005	expressed as a	quality		- Positive
			percentage of	comprising of		and
	Sectoral		GDP.	investment		significant
	analysis			profile index		
			Source: World	and law &		
			Bank, World	order		
			Development	Source: ICRG		
			Indicators.			
Muhammad	7 Asian	1996	Inward FDI	Institutional	Panel data	Governance
et al. (2011)	economi	-	Source: Central	quality	- Fixed	- Positive
	es	2007	banks of each	Source: World	effect and	and
			country	Bank,	Random	significant
				Kaufmann et	effect	effect
				al. (2003).	model	

Jadhav	5 BRICS	2000 -	Inward FDI in	Voice and	Panel data	Regulatory
(2012)	nations	2009	billion dollars	accountability		quality –
	(Brazil,		Source: World	Government		positive and
	Russia,		Bank	effectiveness		insignificant
	India,			Regulatory		Rule of law
	China			quality		– positive
	and			Rule of law		and
	South			Corruption		significant
	Africa)			Political		Democracy
				stability		– negative
						and
						significant
						Political
						stability –
						negative
						and
						insignificant
						Control of
						corruption –
						positive and
						insignificant
Luca and	103	2001 -	Private capital	World	Cross	Mixed
Spatafora	countries	2007	flows (which	governance	country	results both
(2012)			includes debt	indicators	and panel	in effect and
			and equity) as a	Source: World	data	significance
			share of nominal	Bank	analysis –	
			GDP		OLS, IV,	
			Source: Global		GMM	
			development		techniques	
			finance, World			
			Bank (2011)			
Habib and	111	1994 -	Source:	Corruption	Panel data	Corruption -
Zurawicki	countries	1998	International	Source: Private	- OLS	Negative
(2001)			Monetary Fund	risk		and
						significant

				assessment		
				company		
Wei (2001)	93	1994 -	Source: OECD	Corruption	Panel data	Corruption -
	countries	1996		Source: World	– random	Negative
				development	effects	and
				indicators	model	significant
Teksoz	102	1995 -	Net inward FDI	Corruption	Panel data	Corruption -
(2004)	countries	2000	as a percentage	Source: Global	– OLS,	Positive and
			of GDP (GDP	competitivenes	2SLS	significant
			measured in	s reports		
			current			
			international			
			dollars)			
			Source: World			
			development			
			indicators			
Voyer and	59	2000 -	Japanese FDI	Corruption	Cross	Corruption -
Beamish	countries	2001	per capita	Source: The	sectional –	Positive and
(2004)			Source: Toyo	Transparency	linear	significant
			Keizai	International	regression	in case of
				Corruption		emerging
				Index (CPI) –		economies.
				2002		Positive and
						insignificant
						in case of
						industrialise
						d
Straub and	106	1995	FDI flows as a	Corruption	Panel data	Corruption -
Edinburgh	countries	_	share of total	Source:		Negative
(2005)		1999	private capital	Corruption		and
			flows	Index from		significant
			Source: IMF's	International		
			International	Country Risk		
			Financial	Guide		

			Statistics			
			Database			
Dahlstrom	99	1996	Total annual	Corruption	Panel data	Corruption -
and Johnson	countries	-	flows of FDI	Source:	– Random	Negative
(2007)		2002	millions of US\$	Transparency	effects	and
			Source: World	International	model	significant
			development	Corruption		
			indicator (2004)	Perception		
				Index (2004)		
Khamfula	18	1994	FDI/Nominal	Corruption	Panel data	Corruption -
(2007)	countries	-	GDP	Source: Centre	- Fixed	Positive and
		2004	Source: IMF	for corruption	effects	significant
			International	research		effect
			Finance			
			Statistics			
Sadig	117	1984	FDI per capita	Corruption	Panel data	Corruption -
(2009)	countries	-	Source:	Source:	- OLS	Negative
		2004	UNCTAD	International		and
				country risk		significant
				guide (ICRG)		
Woo and	8 Non-	1984	Ratio of a	Corruption	Panel data	Corruption
Heo (2009)	OECD	-	nations share in	level		– negative
	countries	2004	world inward	Source:		and
			FDI to its share	International		significant
			in global GDP	country risk		
			Source:	guide (ICRG)		Democracy
			UNCTAD			– negative
						and
						insignificant
						(Non
						OECD
						Asian
						countries)
						Democracy
						– positive

						and
						significant
						in case of
Mathur and	29	1980 -	Net inward FDI	Corruption	Panel data	Corruption -
Singh	countries	2000	Source: IMF	perception	- Random	Positive and
(2013)	(emergin				effects	significant
	g or				GLS	
	developi					
	ng)					
Singh and	31	1970 -	RFDI = FDI	Political risk	Pooled	Political
Jun (1995)	countries	1993	flows in	index.	time series	risk -
			constant dollars		and cross	Positive
			relative to real	Source:	sectional	effect but
			GDP.	Business	analysis.	results are
				Environment		not robust
			Source: World	Risk		
			Debt tables,	Intelligence,		
			World Bank.	S.A. (BERI)		
Busse and	83	1984 -	FDI net inflows	12 category	Panel data	Government
Hefeker	developi	2003	per capita in	political risk		stability,
(2005)	ng		current US	Index and		absence of
	countries		dollars (FDI).	institutions		internal
						conflict and
			Source:	Source:		tensions,
			UNCTAD	International		democratic
			(2005).	Country Risk		rights, law
				Guide (ICRG)		and order
						have
						significant
						effect
Baek and	22	1984 -	Stock of FDI in	12 category	Panel data	Political
Qian (2011)	industrial	2008	the host country.	political risk	– Basic	stability -
	ised and			Index and	gravity	Positive and
	94			institutions	model	significant
	developi					effect in

	ng			Source:		case of all
	countries			International		and
				Country Risk		developing
				Guide (ICRG)		countries.
Zheng	135	1980 -	FDI net inflows	Democracy	Time	Democracy
(2011)	developi	2008	as a percentage		series	- Positive
	ng		of GDP.	Source:	cross	and
	countries			Henisz's	sectional	significant
			Source: World	(2000a)	data	
			Development	political		
			Indicators	constraints		
			(WDI) database.	index polcon.		
Li and	53	1982 -	FDI net inflows	Democracy –	Pooled	Democracy
Resnick	countries	1995	measured in	Polity IV	time-series	has both
(2003)			billions of	Property rights	cross	positive and
			current US	protection	section	negative
			dollars.	index.	data	effect
			Source: World	Source:		
			Bank's World	Stephen Knack		
			Development	and Philip		
			Indicators.	Keefer for the		
				IRIS centre at		
				the University		
				of Maryland.		
Ahlquist	80	1985 -	Net inward FDI.	Institutional	Unbalance	Governance
(2006)	developi	2002		quality	d panel	– positive
	ng		Source: World		time series	and
	countries		Bank			significant
Goodspeed	53	1984	Aggregate stock	Policy	Panel data	Overall
et al.,(2010)	countries	-	of FDI	variables =		governance
	for tax	2002		Infrastructure		= negative
	rates.	for tax	Source:	quality		and
		rates.	UNCTAD.			significant
L	1	1	1		1	1

			FDI stock of	Source: World		Corruption
	47	1995	destination	Bank.		= negative
	countries	_	country			and
	for the	2002		Good		insignificant
	corruptio	for	Source: OECD	governance =		and
	n index.	corrup		corruption		significant
		tion		perception		
	37	index.		index and		
	countries			government		
	for			efficiency.		
	infrastru	1996				
	cture	_		Corruption		
	index.	2002		perception		
		for		index		
		infrast				
		ructur		Source:		
		e		Transparency		
		idex.		International.		
				Government		
				efficiency		
				Source: IMD		
				Competitivene		
				ss Yearbook.		
Arbatli	46	1990 -	FDI as a	Law and order;	Panel data	Law and
(2011)	countries	2009	percentage of	Bureaucracy		order –
			FDI.	quality		negative
						and
			Source: IFS,	ICRG		insignificant
			World			
			Investment			Bureaucrac
			Report			y – negative
			Database.			

						and
						insignificant
Davis	109	1980 -	Inward FDI in	Democracy	Cross	Democracy
(2011)	states	2005	millions of US		sectional	- Negative
			dollars.	Source: Polity	time series	and
				IV		insignificant
			Source: World			
			Development			
			Indicators			
			(WDI, World			
			Bank, 2007).			
Gordon et	124	1996 -	Foreign direct	Democracy,	Panel data	All
al., (2012)	countries	2009	investment	Political		governance
			inflow data in	stability,		variables
			current US	corruption,		show mixed
			dollars.	regulation,		effects
				government		
			Source: World	effectiveness		
			development	and law		
			indicator (WDI)			
			database	Source: World		
			published by	Bank,		
			world bank.	Kaufmann et		
				al. (2003).		
Fan et al.,	61	1961 -	Per capita FDI	Rule of Law.	Panel data.	Law -
(2009)	countries	2003	in constant 2000			Positive and
			US\$ winsorized	Source:		significant
			at 5%.	International		
				Country Risk		
			Source: World	Guide.		
			Bank, World			
			Development			
			Indicators			
			database.			

Busse et al.,	82	1984 -	Absolute	Political risk	Panel data	Political
(2011)	countries	2004	bilateral inward			stability -
			FDI.	Source:		Positive and
				International		significant
			Source: Country Risk			
			UNCTAD.	Guide.	e	
Harms and	62	1989 -	Average level of	Democracy	Panel data	Democracy
Ursprung	developi	1997	per capita FDI.			- Positive
(2002)	ng and			Source:		but
	emerging		Source: World	Freedom		statistically
	market		Bank.	Bank. House (2000)		mixed
	countries					effect
Jensen	79	1990	Cross sectional -	Democracy –	Cross	Democracy
(2003)	countries	-	Average net	Polity III data	sectional	has positive
	for cross	Cross	inward FDI as a	Jagger & Gurr	data for	and
	sectional	sectio	percentage of	1996;	1999;	significant
	data.	nal.	GDP.			effect;
			Corruption, Time		others –	
	114	1970	Time-series	Rule of law,	series	insignificant
	countries	-	cross-sectional –	ross-sectional – Corruption and cross-		
	for time-	1997	Annual inward		sectional	
	series	for	FDI as a	Bureaucracy –	analysis	
	cross-	time-	percentage of	Easterly Data		
	sectional	series	GDP.	Set, Easterly		
	data.	cross-		1999		
		sectio	World Bank's			
		nal	World			
		data.	Development			
			Indicators 1999.			
Jensen &	115	1975 -	Inward FDI as a	Democracy	Cross-	Democracy
McGillivray	countries	1995	percentage of		sectional	- Positive
(2005)			GDP.	Source: time-series and		and
				Marshall and	data	significant
			Source: World	Jaggers (2000).		
			Bank's World			

		Development				
			Indicators, 1999.			
Busse	69	1972 -	Foreign direct	Democracy	Panel data	Democracy
(2004)	developi	2001	investment per			- Positive
	ng and		capita, net	Source:		and
	emerging		inflows in	Freedom		significant
	market		current US	House (2002)		effect from
	countries		dollars.	data for		1990
				political rights		onwards
			Source:	and civil		
			UNCTAD,	liberties.		
			2003.			
Blanton &	Non-	1980 -	Net inward FDI	Democracy	Time-	Democracy
Blanton	OECD	2003	as a percentage		series	- Positive
(2007)	countries		of total GDP.	of total GDP. Source: cross-		and
				Developed by	sectional	significant
			Source: World	Stohl, Gibney,	data	
			Development	Poe and Co-		
			Indicators,	researchers.		
			World Bank,			
			2005.			
Choi (2008)	Developi	20	Foreign direct	Democracy	Pooled	Democracy
	ng	years	investment as a		panel data	- Positive
	countries		ratio of GDP in	Source: Polity		and
			dollar amounts.	IV		significant
Guerin and	14	1992 -	Bilateral gross	Democracy	Panel data	Democracy
Manzocchi	OECD	2004	inward FDI			- Negative
(2009)	source		from source	Source: The		and
	countries		country to host	Freedom		significant
	and 24		country in	House Political		
	emerging		constant 2000	Rights index.		
	host		US dollars.			
	countries					

			Source: OECD			
			International			
			Direct			
			Investment			
			Database (2006			
			release).			
Doces	55	1990 -	Inward flows of	Democracy	Panel data	Democracy
(2010)	countries	1999	FDI measured in			- Positive
			millions of	Source: Polity		and
			dollars.	IV		significant
			Source: World			
			Bank.			

2.8 OVERVIEW OF MEAURES OF GOVERNANCE AND FDI META-REGRESSION ANALYSIS

Field	Search	Types of	Effect size	Number of	Countries	Aim of the
	engines	studies		studies		study
	used	included		(estimates)		
Measures	Google,	English	Partial	48 (771*)	South	Parameter
of	Web of	language	correlation		and East	estimate and
governance	Knowledge	studies –			Asia &	heterogeneity
and FDI		published			Pacific	
		and			countries	
		unpublished			as	
					defined	
					by world	
					bank +	
					South	
					Korea	

*Total number of estimates (combining all measures of governance)
2.9 PRECISION EFFECT TEST (PET)

2.7.1 SIMILLE MILTA REORESSION – I RECISION EFFECT TEST (TET)	2.9.1 \$	SIMPLE META	REGRESSION -	PRECISION	EFFECT TEST	(PET)
---	----------	-------------	---------------------	-----------	-------------	-------

	VOICE	POLITI	GOVER	REGU	LAW	COR	AGGR
	AND	CAL	NMENT	LATIO		RUPT	EGAT
	ACCOU	STABI	EFFECT	Ν		ION	Е
	NTABIL	LITY	IVENES				GOVE
	ITY		S				RNNA
							CE
PET	-0.03	0.02	0.05	0.08	0.04	-0.05	0.08
(UNW	(-1.70)	(0.35)	(0.74)	(2.44)	(1.60)	(-	(1.44)
EIGHT	$R^2 = 0.25$	$R^2 = 0.0$	$R^2 = 0.03$	$R^2 = 0.4$	$R^2=0.$	1.73)	$R^2 = 0.0$
ED)		5		6	09	$R^2=0.$	2
						12	
PET	-0.02	0.10	0.00	0.09	0.13	-0.04	0.04
(WEIG	(3.65)	(3.50)	(0.01)	(2.43)	(11.33	(-	(0.91)
HTED)	$R^2 = 0.08$	$R^2 = 0.1$	$R^2 = 0.01$	$R^2=0.5$)	1.00)	$R^2 = 0.0$
		3		5	$R^2=0.$	$R^2=0.$	2
					19	10	
Ν	149	154	36	51	42	166	62

2.9.2 MULTIPLE META REGRESSION – PRECISION EFFECT TEST (PET)

	VOICE	POLITI	GOVE	REGU	LAW	CORR	AGGR
	AND	CAL	RNME	LATIO		UPTIO	EGAT
	ACCO	STABI	NT	Ν		Ν	Е
	UNTA	LITY	EFFEC				GOVE
	BILIT		TIVEN				RNNA
	Y		ESS				CE
PET	0.12	-0.96	0.17	0.66	0.27	0.10	0.07
(WEIG	(6.03)	(-6.10)	(2.37)	(8.05)	(4.53)	(1.71)	(2.19)
HTED)	Adj.R ²						
	=0.87	=0.90	=0.14	=0.80	=0.76	=0.71	=0.42
PET	0.12	-0.96	0.17	0.66	0.27	0.10	0.07
(CLUS	(2.43)	(-4.17)	(7.95)	(6.72)	(12.40)	(1.35)	(1.51)]
TERED	$R^2 = 0.8$	$R^2 = 0.9$	$R^2=0.1$	$R^2 = 0.8$	$R^2=0.7$	$R^2=0.7$	$R^2 = 0.4$
)	8	1	9	1	9	3	5
N	149	154	36	51	42	166	62

2.10 RESULTS FOR VOICE AND ACCOUNTABLITY MEASURE

	SIMPLE META	MULTIPLE META
	REGRESSION	REGRESSION
Un weighted estimates, β0	0.02	-0.97
(Row1)	(1.64)	(-29.45)
	R ² =0.26	R ² =0.88
Estimates weighted by precision, β0	0.03	-0.97
(Row2)	(2.12)	(-294.8)
	R ² =0.05	R ² =0.89
Number of estimates	147	147

2.10.1 SIMPLE AND MULTIPLE META REGRESSION RESULTS

* Infeasible value (not between -1 and +1). This could be due to unobserved heterogeneity.

2.11 MODERATOR VARIABLES FOR VOICE AND ACCOUNTABILITY MEASURE

MODERATOR VARIABLES	VOICE AND ACCO	DUNTABILITY
Fdi1	-0.04	-0.04
	(-3.67)	(-2.49)
Lauthor2	1.09	1.09
	(11.95)	(15.71)
Lauthor3	1.14	1.14
	(1.74)	(6.81)
Subject2	1.12	1.12
	(30.35)	(24.52)
Subject3	1.05	1.05
	(30.30)	(65.84)
Dsource4	0.06	0.06
	(4.59)	(4.68)
Method1	-0.04	-0.04
	(-3.06)	(-0.82)
Number of observations	147	147
Adjusted R ² /R ²	0.88	0.89

2.12 FUNNEL PLOT AND CHRONOLOGICAL ORDER – VOICE AND ACCOUNTABILITY





3.1 FOREIGN DIRECT INVESTMENT, NET INFLOWS (% OF GDP) AND GDP PER CAPITA GROWTH (ANNUAL %) FOR SOUTH AND EAST ASIA & PACIFIC COUNTRIES FROM 1980 - 2012

S No	Country Name	Foreign direct	GDP per capita growth
		investment, net	(annual %)
		inflows (% of GDP)	Total 1980 - 2012
		Total 1980 - 2012	
1	Afghanistan	19.44419326	51.79618818
2	Bangladesh	10.73793938	83.64950207
3	Bhutan	11.77360721	185.8861903
4	India	23.14934215	137.0493466
5	Maldives	104.9397353	92.64983969
6	Nepal	3.437867088	63.44560173
7	Pakistan	31.76716453	76.13858073
8	Srilanka	35.35383478	122.0381579
9	Cambodia	102.8831484	100.3651837
10	China	88.37563769	290.4240831
11	Fiji	124.1558657	29.31332501
12	Indonesia	23.54549626	123.9951353
13	Kiribati	26.53998207	-60.62413252
14	North Korea	0	0
15	Lao PDR	71.44000787	105.59484
16	Malaysia	129.8912812	118.9182412
17	Palau	103.4352979	-13.346027
18	Papua New Guinea	92.75981917	18.48078628
19	Phillippines	39.63774466	32.57815305
20	Samoa	53.13671039	47.67167517
21	Solomon Islands	121.6002	17.4355877
22	Thailand	76.62320399	139.3261377
23	Timor Leste	25.15579257	36.32144373
24	Tuvalu	207.2438055	37.871193
25	Tonga	34.03938971	52.64713537
26	Vanuatu	251.6043381	24.27734224
27	Vietnam	136.4825187	137.8556875

28	South Korea	16.37862953	164.9093698

Source: World Bank (2013) [The above figures are calculated based on the available data on World Bank website. There are data gaps for most of the countries [for few years from 1980 – 2012. In case of North Korea, figures are unavailable on World Bank Database]

3.2 SEARCH KEYWORDS USED IN FDI AND GROWTH META-REGRESSION ANALYSIS

Keywords for inward FDI: FDI or Foreign direct investment or offshore investment or cross boarder investment or investment abroad or overseas investment or foreign assets or Greenfield investment or foreign investment or foreign ventures or foreign reinvestment or foreign assets or non-local investments or international investment or outside investment or non-native investment or remote investment or non-domestic investment or non-resident investment or distant investment or investment or inflows or direct investment or investment in other countries.

Keywords for economic growth: Economic growth or development or economic performance or investment or labour productivity or capital or innovation or labour market participation or progress or expansion or increase or improvement or advance.

Keywords for South and East Asia & Pacific countries: Emerging economies or East Asian economies or South east Asian economies or East Asia or South Asia or South east Asia or Afghanistan or Bangladesh or Bhutan or India or Maldives or Nepal or Pakistan or Sri Lanka or American Samoa or Cambodia or China or Fiji or Indonesia or Kiribati or Korea, Dem. Rep. or Lao PDR or Malaysia or Marshall Islands or Micronesia, Fed. Sts or Mongolia or Myanmar or Palau or Papua New Guinea or Philippines or Samoa or Solomon Islands or Thailand or Timor-Leste or Tuvalu or Tonga or Vanuatu or Vietnam or Asian or Developing economies or Developing countries.

3.3 PIOS FRAMEWORK USED IN FDI AND GROWTH META-REGRESSION ANALYSIS

Population – The study should focus on South and East Asia Pacific economies or equivalent as specified in the search criteria.

Independent variable - The study should be examining the impact of inward FDI or its equivalent as specified in the search criteria.

Outcome variable - The study should be examining economic growth or as defined in the search criteria.

Study design - Study design can be either theoretical or empirical. A study is considered to be theoretical if it is based on some theoretical model drawing verbal or mathematical conclusions analysing impact of economic governance on inward FDI. A study is considered to be empirical if it is based on regression model and draws an estimation model to estimate inward FDI on economic growth.

3.4 STUDIES SATISFYING PIOS CRIETRIA IN FDI AND GROWTH META-REGRESSION ANALYSIS

Author (year)	Population	Independent	Dependent	Study
		variable	variable	design
Abbott, Bentzen and Tarp (2009)	Т			
Abraham, Konings and Slootmaekers	Т			
(2008)				
Acharyya (2009)	Y	Y	Y	Y
Adam and Filippaios (2007)	Y	Ν	Ν	Y
Agatiello (2007)	Т			
Agosin and Machado (2005)	Y	Ν	N	Y
Ahlquist and Prakash (2008)	Y	Ν	N	Y
Ahmad, Alam and Butt (2003)	Y	N	N	Y
Ahmed (2010)	Y	N	Y	Y
Ahmed (2012)	Y	Y	Y	Y
Ahuja and Nabar (2012)	Y	N	Y	Y
Alfaro, Chanda, Kalemli-Ozcan and	Y	Y	Y	Y
Sayek (2004)				
Alfaro, Chanda, Kalemli-Ozcan and	Т			
Sayek (2010)				
Algucil, Cuadros and Orts (2010)	Y	Y	Y	Y
Ali and Nishat (2009)	Y	Ν	Y	Y
Ang (2009)	Y	Y	Y	Y
Anwar and Nuguyen (2010)	Y	Y	N	Y
Anwar and Nuguyen (2011)	Y	Y	N	Y

Anwar and Coorey (2012)	Y	Y	Y	Y
Arnold and Javorcik (2009)	Y	Y	N	Y
Asterious and Price (2005)	Y	N	N	Y
Athukorala and Rajapatirana (2003)	Y	Y	N	Y
Azman-Saini, Baharumshah and Law	Y	Y	Y	Y
(2010)				
Bagchi (2002)	Т			
Baharumshah and Almasaied (2009)	Y	Y	Y	Y
Baharumshah and Thanoon (2006)	Y	Y	Y	Y
Balasubramanyam (2002)	Т			
Balasubramanyam and Sapsford (2007)	Т			
Balasubramanyam, Salisu and Sapsford	Y	Y	Y	Y
(1996)				
Balasubramanyam, Salisu and Sapsford	Y	Y	N	Y
(2006)				
Banga (2003)	Y	Y	N	Y
Basu and Guariglia (2003)	Y	Y	Y	Y
Basu and Yao (2009)	Y	N	N	Y
Bayoumi and Lipworth (1997)	Т			
Bende-Nabende and Ford (1998)	Y	Y	N	Y
Beugelsdijik, Smeets and Zwinkels	Y	Y	Y	Y
(2008)				
Bhaduri (2005)	Y	N	N	Y
Bhalla (1998)	Y	N	N	Y
Blackman (1998)	Т			
Bleaney (1996)	Y	N	N	Y
Borensztein, Gregorio and Lee (1998)	Y	Y	Y	Y
Bosworth, Collins and Reinhart (1999)	Y	Y	N	Y
Brink (2003)	Т			
Braodman (2002)	Т			
Brooks, Fan and Sumulong (2003)	Т			
Brooks, Fan and Sumulong (2004)	Т			
Buckley, Clegg, Zhend, Siler and	Y	Y	N	Y
Giorgioni (2007)				

Buckley, Wang and Clegg (2007)	Y	N	N	Y
Bunyaratavej, Hahn and Doh (2008)	Y	N	N	Y
Burger and Karreman (2010)	Т			
Burhop (2004)	Т			
Burke and Ahmadi-Esfahani (2006)	Y	N	N	Y
Bussmann (2010)	Y	N	N	Y
Calvo, Leiderman and Reinhart (1996)	Т			
Carlos, Lliana and Perez (2011)	Т			
Chakraborty and Basu (2010)	Y	N	N	Y
Chakraborty and Nunnenkamp (2007)	Y	Y	N	Y
Chang (2006)	Y	Y	N	Y
Chang (2012)	Y	N	N	Y
Chang and Lu (2011)	Y	N	N	Y
Chantasasawat, Fung, Iizaka and Siu	Y	N	N	Y
(2005)				
Chari (2004)	Т			
Chaudhary and Qaisrani (2002)	Y	N	Y	Y
Chaudhuri and Mukhopadhyay (2003)	Т			
Chen, Chang and Zhang (1995)	Y	Y	N	Y
Chen, Melachroinos and Chang (2010)	Т			
Chew-Ging (2009)	Y	Y	N	Y
Das (2007)	Т			
Dees (1998)	Y	Y	N	Y
Deng, Falvey and Blake (2012)	Т			
Dhanani and Hasnain (2002)	Т			
Dhar and Roy (1996)	Т			
Diao, Rattso and Stokke (2002)	Y	N	N	Y
Dimelis and Papaioannou (2010)	Y	Y	N	Y
Doytch and Uctum (2011)	Y	Y	N	Y
Du, Li and Wu (2011)	Y	N	N	Y
Durham (2004)	Y	Y	Y	Y
Durlauf (2006)	Т			
Dutt (1997)	Y	Y	N	Y

Duttaray, Dutt and Mukhopadhyay	Т			
(2011)				
Dwibedi and Chaudhuri (2007)	Т			
Economidou, Lei and Netz (2006)	Y	Y	Y	Y
Editorial (2007)	Т			
Emilia and Chaitanya (2009)	Т			
Erdem (2012)	N	Y	Y	Y
Ericsson (2010)	N	Y	Y	Y
Estrade, Park and Ramayandi (2010)	Y	N	Ν	Y
Fan (2002)	Т			
Felipe (2008)	Т			
Filip and Daniel (2002)	Т			
Fratzscher and Bussiere (2004)	N	Y	Y	Y
Freeman (2004)	Т			
Fu (2008)	Y	Y	N	Y
Fu and Balasubramanyam (2005)	Y	Y	Ν	Y
Fu and Gong (2011)	Y	Y	N	Y
Fung, Iizaka and Tong (2004)	Т			
Gao (2005)	Т			
Garcia (2007)	Т			
Gazioglou and McCausland (2002)	Т			
Ge (2006)	Y	Y	Ν	Y
Gelan (2004)	Т			
Girma and Gong (2008)	N	N	Y	Y
Girma, Gong and Gorg (2008)	Y	Y	N	Y
Glass and Saggi (2002)	Т			
Gorg and Greenaway (2003)	Т			
Han, Liu, Kong, Tang and Kan (2011)	Т			
Havrylchyk and Poncet (2007)	Y	N	Ν	Y
Hein (1992)	Y	Y	Ν	Y
Hendricks (2000)	Т			
Hermes and Lensink (2003)	Y	Y	Y	Y
Herzer, Klasen and Nowak-Lehmann	Y	Y	Ν	Y
(2008)				

Hitam and Borhan (2012)	Y	Y	Ν	Y
Hooi and Wah (2010)	Y	Y	N	Y
Huinqun and Jinyong (2010)	Y	N	Y	Y
Ito, Dominguez, Qureshi, Shengman and	Т			
Yoshitomi (1999)				
Jensen (2006)	Т			
Jiang, Cheng and Isaac (1998)	Т			
Jiang, Yang and Wang (2011)	Ν	Y	Y	Y
Jin, Lee and Kim (2008)				
Jindra and Rojec ()	Т			
Jung (2007)	Т			
Kajiwara (1994)	Y	N	Y	Y
Kasuga (2007)	Y	N	Y	Y
Kathuria (2010)	Y	Y	Ν	Y
Kholdy (1995)	Ν	Y	Y	Y
Kim (2010)	Y	N	Ν	Y
Kim and Trumbore (2010)	Ν	Y	Y	Y
Kim and Yang (2011)	Т			
Kim and Zhang (2008)	Т			
Kimura (2012)	Y	Ν	Ν	Y
Kinoshota (2006)	Т			
Kiong and Jomo (2005)	Y	Y	Ν	Y
Klasra (2009)	Y	Y	Ν	Y
Kohpaiboon (2003)	Y	Y	Ν	Y
Konings (2000)	Ν	Y	Y	Y
Kose, Prasad and Terrones (2008)	Y	Y	Ν	Y
Kottaridi and Stengos (2010)	Y	Y	Y	Y
Kotwal (2010)	Т			
Krammer (2010)	Ν	Y	Y	Y
Laaksonen-Craig (2004)	Ν	Y	Y	Y
Lall and Narula (2004)	Т			
Lan, Kakinaka and Huang (2012)	Y	Y	N	Y
Lardy (1995)	Т			
Larty (2008)	Т			

Le and Hui (2006)	Т			
Le and Pomfret (2011)	Y	Y	Ν	Y
Le and Suruga (2005)	Y	Y	Y	Y
Lee (2005)	Т			
Lee (2009)	Y	N	N	Y
Lee, Lee and Kim (2011)	Y	Y	N	Y
Lemi (2004)	Y	Y	N	Y
Lensick and Morrissey (2006)	Y	Y	Y	Y
Li and Liu (2005)	Y	Y	Y	Y
Li and Sherali (2003)	Т			
Liang (2007)	Т			
Liefner and Wei (2011)	Т			
Lin, Chen, Du and Niu (2012)	Y	N	Y	Y
Lin, Lee and Yang (2011)	Y	Y	N	Y
Lin, Liu and Zhang (2009)	Y	Y	Ν	Y
Lindsey (1992)	Т			
Lipsey and Sjoholm (2011)	Т			
Lipsey and Sjoholm (2011a)	Т			
Liu (2002)	Y	Y	Ν	Y
Liu, Wang and Wei (2001)	Y	N	N	Y
Lombard and Lombard (2011)	Т			
Lopes, Ruddock and Ribeiro (2002)	Т			
Luo (2002)	Т			
Lv, Wen and Xiong (2009)	Y	N	N	Y
Mah (2010)	Y	Y	Ν	Y
Mah (2010a)	Y	Y	Ν	Y
Maj (2010)	Y	N	N	Y
Majeed and Ahmad (2008)	Y	N	Ν	Y
Makki and Somwaru (2004)	Y	Y	Y	Y
Maliar, Maliar and Sebastian (2008)	Т			
Manuscript (2008)	N	Y	Y	Y
Martinez-Vazquez, McNab and Everhart	Y	N	Ν	Y
(2005)				
Marwah and Tavakoli ()	Т			

Mastromarco (2008)	Т			
Mastromarco and Ghosh (2008)	Т			
Masuyama, Mitarai abd Iwasa (1999)	Т			
McCloud and Kumbhakar (2012)	Y	Ν	N	Y
Mello (2007)	Т			
Mercereau (2005)	Y	Ν	Ν	Y
Michie (1999)	N	Ν	Ν	Y
Michie (2001)	Т			
Mielniki and Goldembergb (2002)	Т			
Milberg (1990)	Т			
Mingyong, Shuijun and Qun (2006)	Y	Ν	Y	Y
Mirza abd Giroud (2003)	Т			
Mlachila and Takebe (2011)	Т			
Mody and Murshid (2005)	Y	Ν	Ν	Y
Montes (1997)	Т			
Moran ()	Т			
Moran (1999)	Т			
Moran, Graham and Blomstrom (2006)	Т			
Mukherjee and Suetrong (2007)	Т			
Mzenda and Buys (2006)	Т			
Naceur, Bakardzhieva abd Kamar (2011)	Y	Y	Ν	Y
Nair-reichert and Weinhold (2001)	Y	Y	Ν	Y
Nguyen and Amin (2001)	Т			
Oura (2008)	Y	Ν	Ν	Y
Park, Kui, Keong, Kin and Peng (2010)	Т			
Qi (2007)	Т			
Reinhart and Khan (1989)	Y	Ν	Y	Y
Saha and Vickers (2001)	Т			
Sasidharan and Khathuria (2011)	Y	Y	Ν	Y
Sebu (2006)	N	Y	Y	Y
Shamshad and Siddiqui (2009)	Т			
Sjoholm ()	Т			
Sjoholm (1998)	Y	Y	Ν	Y
Soysa and Oncal (1999)	Y	Ν	Y	Y

Sun (2007)	Т			
Tang, Metwalli and Smith (2010)	Т			
Tantatape and Komain (2009)	Y	N	N	Y
Thun (2006)	Y	Y	N	Y
Tian (2010)	Y	N	Y	Y
Tian, Lin and Lo (2004)	Y	Y	N	Y
Todo, Zhand and Zhou (2006)	Y	N	Y	Y
Todo, Zhang and Zhou (2011)	Y	N	Y	Y
Tomohara and Yokota (2011)	Y	Y	N	Y
Tuan and Ng (2004)	Y	N	N	Y
Tuan and Ng (2006)	Y	N	N	Y
Tuan, Ng and Zhao (2009)	Y	Y	Ν	Y
Vacaflores and Mogab (2012)	Y	Y	Ν	Y
Vadlamannati and Tamazian (2009)	Y	Y	Ν	Y
Vita and Kyaw (2009)	Y	Y	Y	Y
Vu (2011)	Y	Y	N	Y
Vu and Ganges (2007)	Y	Y	Ν	Y
Waheed (2004)	Т			
Wang and Wong (2009)	Y	Y	Y	Y
Wang and Wong (2010)	Y	Y	N	Y
Wang and Wong (2011)	Y	Y	Y	Y
Wei ()	Y	Y	N	Y
Wei, Yao and Liu (2009)	Y	Y	N	Y
Wen (2007)	Y	Y	N	Y
Wen, Gao-bang and Jin-Song (2009)	Y	Y	N	Y
Whalley and Xin (2006)	Т			
Wignaraja (2008)	Y	N	N	Y
Wong, Tang and Fausten (2007)	Y	Y	N	Y
Woo (2008)	Y	Y	N	Y
World Investment Report (2006)	Т			
Xiping and Ming ()	Y	Y	N	Y
Xu and Sheng ()	Y	N	Y	Y
Xuan and Xing (2006)	Y	Y	N	Y
Xue and Shu-hui (2008)	Y	Y	N	Y

Yang, Xu, Wang, Lai and Wei (2009)	Y	Y	N	Y
Yao (2006)	Y	Y	N	Y
Yao and Wei (2006)	Y	Y	Ν	Y
Yasmin (2005)	Y	Y	Ν	Y
Young and Lan (1997)	Т			
Yousaf, Hussain and Ahmad (2008)	Y	Y	Ν	Y
Yu and JingMei (2009)	Y	Y	Ν	Y
Yue (1999)	Т			
Yusoff ()	Y	Y	Ν	N
Zhang and Rogers (2009)	Y	Y	Ν	Y
Zhang (2001)	Y	Ν	Ν	Y
Zhang (2002)	Y	Y	N	Y
Zhang (2011)	Т			
Zhang and Felmingham (2002)	Y	Y	N	Y
Zhang and Zhao (2007)	Т			
Zhang, Wang and Zhu (2012)	Y	N	Y	Y
Zhao and Zhang (2010)	Y	Y	Ν	Y
Zhaoyang ()	Y	Y	Ν	Y
Zhixiong (2010)	Y	Ν	Ν	Y
Bing	N	Y	Y	Y
Feng (2011)	Y	Y	N	Y
Ping and Chen (2010)	Т			
Pomfret (1999)	Т			
Qi, Zheng, Laurenceson and Li (2009)	Y	Y	Ν	Y
Tomohara and Takki (2011)	Y	N	Y	Y
Vu and Noy (2009)	N	Y	Y	Y
Wang (2012)	Т			
Wei and Liu (2006)	Y	Y	N	Y
Wen (2006)	Y	Y	N	Y
Wu (2000)	Y	N	N	Y
Yu, Xin, Guo and Liu (2011)	Y	Y	N	Y
Yu	Y	Y	N	Y
Zaman, Khan and Ahmad (2012)	Y	Y	Ν	Y
Zeng, Wan and Tam (2009)	Y	Y	Ν	Y

Mzenda and Buys (2006)	Т			
Meyer (2003)	Т			
Lee and Chang (2009)	Y	Y	Ν	Y
Lauridsen (2004)	Т			
Lalwani (2002)	Y	Y	Y	N
Kottaridi (2005)	N	Y	Y	Y
Huang (2004)	Y	N	Y	Y
Khamfula (2007)	Y	Y	Ν	Y
Mullen and Williams (2005)	N	Y	Y	Y
Schaumburg-Muller (2003)	Т			
Velde and Morrissey (2004)	Y	Y	Ν	Y
Wang (2010)	N	Y	Y	Y
Abdou and Moshiri (2009)	Y	N	Ν	Y
Ahmad and Hamdani (2003)	Y	Y	Y	Y
Ahmad and Iman (2011)	Т			
Ahuja and Nabar (2012)	Y	N	Y	Y
Anwar and Cooray (2012)	Y	Y	Ν	Y
Anwar and Nguyen (2011)	Y	Y	Ν	Y
Anwar and Nguyen (2010)	Y	Y	Ν	Y
Athukorla and Rajapatinara (2003)	Y	Y	Ν	Y
Athukorla (2002)	Т			
Athukorla and Menon (2001)	Т			
Athukorla and Tien (2012)	Y	N	Ν	Y
Basu, Chakraborty and Reagle (2003)	Y	N	Ν	Y
Berthelemy and Demurger (2000)	Т			
Bhat and Raj (2006)	Y	Y	Ν	Y
Bhat, Sundari and Raj (2004)	Y	N	Ν	Y
Brambilla (2009)	Y	N	Y	Y
Brooks, Fan and Sumulong (2003)	Т			
Bussiere and Fratzscher (2008)	Y	Y	Ν	Y
Campos and Kinoshita (2002)	N	Y	Y	Y
Chen (1997)	Т			
Chen, Ge and Lai (2011)	Y	Y	Ν	Y
Cheung (2010)	Y	N	Y	Y

Cheung and Lin (2004)	Y	Y	N	Y
Chew-Ging, Broga and Ehsan ()	Y	Y	N	Y
Choong (2005)	Y	Y	N	Y
Chow (2006)	Y	N	N	Y
Chow and Zeng (2001)	Т			
Christerson (2000)	Y	Ν	N	Y
Chuang and Lin (1999)	Y	Y	N	Y
Cole, Elliot and Zhang (2011)	Y	N	N	Y
Colen, Maertens and Swinnen (2008)		Т		
Contractor (1995)	Т			
Danhui (2010)	Т			
Doraisami (2007)	Т			
Editorial (2007)	Т			
Emilia and Chaitanya (2009)	Т			
Fan (2002)	Т			
Freckleton, Wright and Craigwell (2011)	Y	Y	Y	Y
Fry (1996)	Y	Y	Y	Y
Fung, Zeng and Zhu (1999)	Т			
Glass and Saggi (2002)	Т			
Hale and Long (2011)	Y	Ν	Y	Y
Herzer (2012)	Y	Y	N	Y
Hoang, Wiboonchutikula and	Y	Y	Y	Y
Tubtimtong (2010)				
Homlong and Springler (2010)	Т			
Hong and Sun (2011)	Y	Y	N	Y
Hosseini (2005)	Т			
Hsiao and Shen (2003)	Y	Y	Y	Y
Hsu and Mckern (1990)	Y	Ν	Y	Y
Huang (2004)	Y	Y	N	Y
Huang, Liu and Xu (2012)	Y	Y	N	Y
Huang, Teng and Tsai (2010)	Y	Y	N	Y
Нуе (2011)	Y	Ν	Y	Y
Im (2007)	Y	Y	N	Y
Ishida ()	Т			

Islam (1994)	Т			
Ito, Yashiro, Xu, Chen and Wakasugi	Y	Y	Ν	Y
(2012)				
Jansan (1995)	Т			
Jansen (2003)	Т			
Kasibhatla, Stewart and Khojasteh	Y	Y	Ν	Y
(2008)				
Kim and Zhang (2008)	Т			
Kotrajaras, Bangorn and Tubtimtong ()	Y	Y	Ν	Y
Liang (2007)	Т			
Lim (1976)	Y	Y	Ν	Y
Liu (2003)	Y	N	Ν	Y
Luiz and Mellio (1997)	Т			
Luo (2002)	Т			
Mingxia (2009)	Y	Y	Ν	Y
Mirza abd Giroud (2003)	Т			
Mlachila and Takebe (2011)	Т			
Mutafoglu (2012)	Т			
Mytelka and Barclay (2004)	Т			
Negara and Adam (2012)	Y	Y	Ν	Y
Ng and Tuan (2006)	Y	Y	N	Y
Nunnenkamp and Stracke (2007)	Y	N	Ν	Y
Ouyang and Fu (2012)	Y	Y	Ν	Y
Prasad, Rajan and Subramanian ()	Y	N	Y	Y
Prasanna (2010)	Y	Y	Ν	Y
Qi (2007)	Т			
Qin, Cagas, Quising and He (2006)	Y	N	Y	Y
Quader (2009)	Y	Y	Y	Y
Rawski (2002)	Т			
Razin and Sadka (2003)	Т			
Reis (2001)	Т			
Reiter and Steensma (2010)	Y	Y	Ν	Y
Rizvi and Nishat (2009)	Y	Y	Ν	Y
Robertson and Teitelbawm (2011)	Y	Y	Ν	Y

Rothgeb (1995)	Y	Y	N	Y
Saadi (2011)	Y	Y	N	Y
Sadoi (2008)	Т			
Saggi ()	Т			
Sahoo and Mathiyazhagan (2003)	Y	Y	N	Y
Semyonov and Shenhavl ()	Y	N	N	Y
Spar ()	Т			
Spenser (2008)	Т			
Srivatsava (2006)	Y	Y	N	Y
Sumner (2005)	Т			
Sun (1998)	Y	Y	N	Y
Sun (2001)	Y	Y	N	Y
Sun (2010)	Y	Y	N	Y
Sun (2011)	Y	Y	N	Y
Sun and Parikh (2001)	Y	Y	N	Y
Suyanto, Blocj and Salim (2012)	Y	Y	N	Y
Suyanto, Salim and Bloch (2009)	Y	N	Y	Y
Sylwester (2005)	Y	Y	Y	Y
Takki (2011)	Y	Y	N	Y
Tang, Selvanathan and Selvanathan	Y	Y	N	Y
(2008)				
Tao (2004)	Y	Y	N	Y
Tekin (2012)	Т			
Thangavelu, Yong and Chongvilaivan	Y	Y	Y	Y
(2009)				
Thompson (2002)	Y	N	N	N
Torre (1981)	Т			
Wang (2010)	Ν	Y	Y	Y
Wang and Yu (2007)	Y	N	N	Y
Wu (2000)	Y	Y	N	Y
Wu (2001)	Т			
Yingxin (2007)	Т			
Yu, Chen and Sun (2010)	Y	Y	N	Y
Zhang (2006)	Y	Y	N	Y

Zhao and Du (2007)	Y	Y	Ν	Y
Zhao and Du (2009)	Y	Ν	Ν	Y
Zhu and Tan (2000)	Y	Y	Ν	Y

Y= Study satisfies the criteria; N= Study does not satisfy the criteria

3.5 NUMBER OF STUDIES SATISFYING PIOS CRITERIA IN FDI AND GROWTH META-REGRESSION ANALYSIS

Criteria	Number of
	studies satisfying
	the criteria
Population (South and East Asia & Pacific countries)	245
Independent variable (Inward foreign direct investments)	183
Outcome variable (Economic growth)	79
Study design – Empirical	262
Decision Select if all 4 criteria match - PIOS	
Select for next stage	32
Deselect studies	387

3.6 COMPOSITION OF PUBLISHED (81%) AND UNPUBLISHED (19%) STUDIES INCLUDED IN FDI AND GROWTH META-REGRESSION ANALYSIS



3.7 DESCRIPTIVE STATISTICS OF MODERATOR VARIABLES INCLUDED IN FDI AND GROWTH META-REGRESSION ANALYSIS

Moderator	Definition	Mean	Standard
variable			deviation
Paper 1	=1 if the estimate is from a study published in a	0.678	0.47
	journal; = 0 otherwise		
Paper 2	=1 if the estimate is from a working paper; = 0	0.317	0.47
	otherwise		
Paper 3	=1 if the estimate is from a discussion paper; = 0	0.007	0.08
	otherwise		
Single	=1 if the model uses FDI and growth data on single	0.060	0.24
	country; = 0 otherwise		
Multi	=1 if the estimate uses FDI and growth data from	0.940	0.24
	multiple countries; $= 0$ otherwise		
Year1	=1 if the estimate belongs to a model that uses yearly	0.415	0.49
	data on FDI; = 0 otherwise		
Year2	=1 if the estimate belongs to a model that uses non-	0.585	0.49
	yearly data on FDI; $= 0$ otherwise		
Obs1	=1 if the estimate belongs to a model where	0.921	0.27
	observations are reported; $= 0$ otherwise		
Obs2	=1 if the estimate belongs to a model where	0.079	0.27
	observations are not reported; $= 0$ otherwise		
Fdi1	=1 if the estimate belongs to a model that uses	0.170	0.38
	relative figures of FDI; $= 0$ otherwise		
Fdi2	=1 if the estimate belongs to a model that uses levels	0.129	0.34
	of FDI; = 0 otherwise		
Fdi3	=1 if the estimate belongs to a model that uses natural	0.356	0.48
	logarithm of FDI; $= 0$ otherwise		
Method1	=1 if the estimate belongs to a model that is estimated	0.377	0.48
	using OLS techniques; = 0 otherwise		
Method2	=1 if the estimate belongs to a model that is estimated	0.188	0.39
	using panel data techniques; $= 0$ otherwise		
Method3	=1 if the estimate belongs to a model that is estimated	0.309	0.46
	using instrumental variable techniques; $= 0$ otherwise		

Method4	=1 if the estimate belongs to a model that is estimated	0.048	0.21
	using time series techniques; $= 0$ otherwise		
Method5	=1 if the estimate belongs to a model that is estimated	0.079	0.27
	using other techniques; $= 0$ otherwise		
Growth1	=1 if the estimate belongs to a model that uses	0.212	0.41
	relative figures of growth; $= 0$ otherwise		
Growth2	=1 if the estimate belongs to a model that uses levels	0.004	0.06
	of growth; = 0 otherwise		
Growth3	=1 if the estimate belongs to a model that uses natural	0.804	0.50
	logarithm of growth; $= 0$ otherwise		
Data1	=1 if the estimate belongs to a model that is estimated	0.491	0.50
	from panel data; $= 0$ otherwise		
Data2	=1 if the estimate belongs to a model that is estimated	0.039	0.19
	from time series data; $= 0$ otherwise		
Data3	=1 if the estimate belongs to a model that is estimated	0.470	0.50
	from cross section data; $= 0$ otherwise		
Country1	=1 if the estimate belongs to East Asia; = 0 otherwise	0.023	0.15
Country2	=1 if the estimate belongs to South East Asia; = 0	0.113	0.32
	otherwise		
Country3	=1 if the estimate belongs to South Asia; = 0	0.031	0.17
	otherwise		
Country4	=1 if the estimate belongs to Mixed countries; = 0	0.834	0.37
	otherwise		
China1	=1 if the estimate belongs to a model which includes	0.492	0.50
	China in the list of sample countries; $= 0$ otherwise		
China2	=1 if the estimate belongs to a model which excludes	0.580	0.50
	China from the list of sample countries; $= 0$ otherwise		
Skorea1	=1 if the estimate belongs to a model which includes	0.594	0.49
	South Korea in the list of sample countries; $= 0$		
	otherwise		
Skorea2	=1 if the estimate belongs to a model which excludes	0.406	0.49
	South Korea from the list of sample countries; $= 0$		
	otherwise		
Fditype1	=1 if the estimate belongs to a model that has used	0.036	0.19
	Greenfield form of FDI; $= 0$ otherwise		

Editype?	-1 if the estimate belongs to a model that has used	0.036	0.19
1 ditype2	Margar and Acquisition form of EDI: – 0 otherwise	0.050	0.17
Editerra 2		0.029	0.26
Fullypes	=1 if the estimate belongs to a model that has used	0.928	0.20
	aggregate FDI; = 0 otherwise		
Lauthor1	=1 if the estimate belongs to a study where the first	0.578	0.50
	author comes from American University; $= 0$		
	otherwise		
Lauthor2	=1 if the estimate belongs to a study where the first	0.244	0.43
	author comes from European University; $= 0$		
	otherwise		
Lauthor3	=1 if the estimate belongs to a study where the first	0.067	0.25
	author comes from South East Asian University; $= 0$		
	otherwise		
Lauthor4	=1 if the estimate belongs to a study where the first	0.112	0.32
	author comes from other University; $= 0$ otherwise		
Journal1	=1 if the estimate is taken from a journal that belongs	0.891	0.31
	to Economic and Finance discipline; = 0 otherwise		
Journal2	=1 if the estimate is taken from a journal that belongs	0.019	0.14
	to Business Management and Accounting discipline;		
	= 0 otherwise		
Journal3	=1 if the estimate is taken from a journal that belongs	0.057	0.23
	to Policy discipline; = 0 otherwise		
Journal5	=1 if the estimate is taken from a journal that belongs	0.033	0.18
	to Development discipline; $= 0$ otherwise		
Omitted1	= 1 if the estimate is taken from a model that includes	0.365	0.48
	population related variable; $= 0$ otherwise		
Omitted2	= 1 if the estimate is taken from a model that includes	0.605	0.49
	domestic investment related variable; = 0 otherwise		
Omitted3	= 1 if the estimate is taken from a model that includes	0.860	0.35
	education related variable; $= 0$ otherwise		
Uni1	=1 if the estimate belongs to a study where the first	0.205	0.40
	author of the study belongs to IVY universities; $= 0$		
	otherwise		
Uni3	=1 if the first author of the study belongs to other	0.795	0.40
	universities; = 0 otherwise		

Abs10a1	=1 if the ABS 2010 ranking of the journal is 1*; = 0 otherwise	0.020	0.14
Abs10a2	=1 if the ABS 2010 ranking of the journal is 2*; = 0 otherwise	0.372	0.48
Abs10a3	=1 if the ABS 2010 ranking of the journal is 3*; = 0 otherwise	0.584	0.49
Abs10a4	=1 if the ABS 2010 ranking of the journal is 4*; = 0 otherwise	0.025	0.16

3.8 SUMMARIES OF EMPIRICAL STUDIES INCLUDED IN FDI AND GROWTH META-

REGRESSION ANALYSIS

Study and year	Time	Countrie	Dependent	Independent	Findings	Technique
	period	s	variable	variable		s
Alguacil,	1976 -	26	Real GDP	Gross fixed	Statisticall	
Cuadros and	2005	developi	per capita	capital as a	у	
Orts (2011)		ng	growth	ratio of FDI	significant	
		countries			and	
					positive	
Anwar and	1970 -	8 South			Statisticall	GMM and
Cooray (2012)	2009	Asian			У	fixed
		countries			significant	effects
					and	
					positive	
Ahmad and	1965 -	32	Real GDP	FDI	Positive	Common
Hamdani	1992	developi	in constant		and	intercept,
(2003)		ng	US \$	(International	statisticall	random
		countries	prices	Monetary	у	effects
			(Penworld	Fund (1994))	significant	and fixed
			(1995))		effect	effects
Alfaro (2003)	1981 -	47	Average	Sectoral FDI	Positive	OLS
	1999	countries	real annual	as a	but	
			per capita	percentage of	insignifica	
			growth rate	GDP	nt effect	
			(World	(OECD's		
			developme	International		
			nt	Direct		

			indicators	Investment		
			(2001))	Statistics		
				Yearbook		
				(2001)) and		
				UNCTAD's		
				World		
				Investment		
				Directory (7		
				– Volume		
				series 1992 –		
				2000)		
Alfaro,	1975 -	71	Growth	FDI inflows	FDI has	OLS
Chanda,	1995	countries	rate of real	(IMF	positive	
Kalemli-Ozcan			per capita	International	and	
and Sayek			GDP in	Financial	statisticall	
(2004)			constant	Statistics)	У	
			dollars		insignifica	
			(World		nt effect	
			Developm		on	
			ent		growth.	
			Indicators			
			(World			
			Bank,			
			2000))			
Alfaro,	1975 -	72	Average	Net FDI	Negative	OLS
Kalemli-Ozcan	1995	countries	growth rate	inflows	and	
and Sayek			of real	(IMF	statisticall	
(2009)			GDP per	International	у	
			capita	Statistics)	insignifica	
			(World		nt effect	
			Developm			
			ent			
			Indicators			
			(World			
	1	1			1	

			Bank,			
			2000))			
Azman-Saini,	1976 -	85	Per capita	FDI inflows	FDI has	Generalis
Baharumshah	2004	countries	real GDP	as percentage	no effect	ed method
and Law			(chain	of GDP	on growth	of
(2010)			weighted)	(World		moments
			(Penn	Bank)		
			World			
			Table			
			(PWT))			
Baharumshah	1982 -	8 Asian	Gross	Foreign	Positive	Dynamic
and Thanoon	2001	countries	domestic	direct	and	generalise
(2006)			product	investment	statisticall	d least
			(Developin	(Developing	у	squares
			g Asian	Asian and	significant	
			and Pacific	Pacific	effect	
			countries,	countries,		
			2003, Vol.	2003, Vol.		
			XXXI,	XXXI,		
			Oxford	Oxford		
			University	University		
			Press, New	Press, New		
			York)	York)		
Balasubramany	1970 -	46	Gross	Stock of	Positive	OLS,
am, Salisu and	1985	developi	domestic	foreign	and	generalise
Sapsford		ng	product in	capital	statisticall	d
(1996)		countries	real terms	(Various	у	instrumen
			(Summers	editions of	significant	tal
			and Heston	Transnational	effect in	variable
			(1988))	Corporations	case of EP	estimator
				in World	countries.	
				Development	Insignific	
)	ant effect	
					in case of	
					IS	

					countries	
					(both	
					positive	
					and	
					negative).	
Basu and	119	1970 -	Growth of	Net inflows	Positive	Fixed
Guariglia	developi	1999	real per	of FDI as a	and	effects
(2003)	ng		capita	percentage of	highly	and
	countrie		GDP	GDP	significant	system
	s		(World	(World		GMM
			Developm	Development		
			ent	Indicators		
			Indicators	(2000))		
			(2000))			
Beugelsdijik,	44	1983 -	GDP per	Total US FDI	Mixed	Two stage
Smeets and	countrie	2003	capita	stock as a %	effects	least
Zwinkels	s		growth (%)	of GDP	with	squares
(2008)			(World	(UNCTAD);	respect to	model
			Developm	Horizontal	developed	
			ent	and Vertical	and	
			Indicators)	FDI	developin	
					g	
					countries	
Borensztein,	69	1970 -	Average	Net inflows	Negative	Three
Gregorio and	developi	1989	annual rate	of FDI	and	stage least
Lee (1998)	ng		of per	(OECD)	statisticall	squares
	countrie		capita real		у	
	s		GDP		significant	
			growth		effect	
			over each			
			decade			
			(Summers			
			and Heston			
			(release			
		1			1	

			5.5 of June			
			1993)			
Busse and	84	1984 -	Real	FDI, net	Positive	System
Groizard	developi	2003	growth of	inflows in	and	GMM
(2008)	ng		GDP per	per cent of	statisticall	
	countrie		capita in	GDP	у	
	s		per cent	(UNCTAD	significant	
			(World	(2007))	effect	
			Bank			
			(2006b));			
			GDP per			
			capita in			
			internation			
			al US\$			
			(PPP)			
			(World			
			Bank			
			(2006b))			
Carkovic and	72	1960 -	Real per	Gross FDI	Positive	OLS,
Levine (2002)	countrie	1995	capita	inflows as a	but	GMM
	S		gross	share of	insignifica	
			domestic	GDP.	nt effect	
			product	Average		
			growth	seven year		
				period FDI		
				(world Bank		
				dataset		
				(Kreey et al.		
				1999) and		
				IMF)		
Durham (2004)	80	1979 -	Real per	FDI	Negative	OLS
	countrie	1998	capita	(OECD) and	and	
	s		GDP	(IFS)	statisticall	
			(World		У	
			Developm		insignifica	

Indicators with	
2000, The OECD	
World data,	
Bank Positive	
(2000)) and	
statically	
significant	
results	
with IFS	
data	
Economidou, 47 1970 - Rate of FDI Positive Fixed	
Lei and Netz developi 1989 GDP per (International and effects	
(2006) ng capita Monetary statisticall	
countrie growth Fund (2002) y	
s (Penn International insignifica	
World Financial nt effect	
Table 5.6 Statistics	
dataset)	
Freckleton, 42 1998 - Per capita FDI as a Positive Dynam	ic
Wright and developi 2008 GDP percentage of and OLS	
Craigwell ng and GDP statisticall	
(2012) 28 y	
develop significant	
ed effect	
countrie	
S	
Fry (1996)1972 -SixRate ofInflow ofNegativeThree	
1992 pacific growth in foreign direct and stage le	ast
basin GNP investment/G statisticall squares	
countries (constant NP (dollar y	
prices, values insignifica	
continuous converted to nt effect	
ly domestic	

			compound	currency,		
			ed)	current		
				prices)		
Hermes and	1970 -	67 less	Per capita	Gross FDI	Negative	Fixed
Lensink (2003)	1995	develope	growth rate	inflows as a	and	effects
		d	(World	percentage of	significant	and
		countries	Bank 1997	GDP	effect	random
			data	(World Bank		effects
			available	1997 data		
			on CD	available on		
			ROM)	CD ROM)		
Herzer (2012)	1970 -	44	Real GDP	FDI as a	Negative	Dynamic
	2005	developi	(World	percentage of	effect on	OLS
		ng	Developm	GDP	growth	
		countries	ent	(UNCTAD)		
			Indicators			
			(2007))			
Hsiao and Shen	1976 -	23	Real GDP	Real FDI	Positive	Vector
(2003)	1997	developi	(World	(World	and	auto
		ng	Developm	Development	statisticall	regression
		countries	ent	Indicator CD	У	
			Indicator	ROM	significant	
			CD ROM	(2000))		
			(2000))			
Kotrajaras	1990 -	15 East	GDP in	FDI in	Positive	Polled
(2010)	2009	Asian	million	million USD	and	regression
		countries	USD	(UNCTAD	statisticall	analysis,
			(UNCTAD	and IMF)	У	Fixed
			and IMF)		significant	effects
					effect in	model
					case of	
					high	
					income	
					and	
					middle	

					income	
					countries	
					only	
Kottaridi and	1970 -	25	Growth	FDI inflows	Positive	System
Stengos (2010)	2004	OECD	rate of	(UNCTAD)	and	GMM
		countries	income per		insignifica	
		and 20	capita		nt for	
		non-	(World		entire	
		OECD	Bank)		sample.	
		countries			Positive	
					and	
					significant	
					for non	
					OECD	
					countries	
					and	
					middle	
					income	
					countries.	
Le and Suruga	1970 -	105	Five year	FDI inflows	Developin	OLS
(2005)	2001	develope	moving	(World	g	
		d and	average of	development	countries	
		developi	per capita	indicators,	- positive	
		ng	GDP	2003 CD	and	
		countries	growth	ROM)	significant	
			(World		•	
			developme		Develope	
			nt		d	
			indicators,		countries	
			2003 CD		- negative	
			ROM)		and	
					significant	
Lee, Lee and	1989 -	122	Gross	Stock of FDI	Positive	IV
Kim (2011)	2008	countries	domestic	inflows	and	technique
		(22	product		significant	

		develope	and per	(OECD and	in case of	
		d)	capita	UNCATD)	all sample	
			GDP		and less	
			(World		developed	
			Bank		countries	
			dataset)		only.	
Lensick and	1975 -	87	Average	Average	Positive	OLS
Morrissey	1997	countries	real per	gross foreign	and	
(2006)		(20 are	capita	direct	statisticall	
		develope	growth rate	investment	у	
		d)	(Easterly	over GDP	significant	
			and Yu	ratio		
			(1999))	(World Bank		
				(1999))		
Li and Liu	1970 -	84	Real GDP	Ratio of FDI	Positive	Single
(2005)	1999	countries	per capita	inflows to	and	equation
		(21	growth	GDP	statisticall	and
		develope	(World	(World	у	simultane
		d and 63	Bank)	Investment	significant	ous
		developi		Directory	effect	equation
		ng		published by		model
		countries		United		
)		Nations and		
				missing data		
				from World		
				Investment		
				Report)		
Makki and	1971 -	66	Mean	FDI	Positive	Seemingl
Somwaru	2000	countries	values of	(World	and	У
(2004)			per capita	development	significant	unrelated
			growth rate	Indicators	under one	regression
			in each	published by	model and	(SUR)
			decade	World Bank	insignifica	and three
				and	nt under	

keydevelopmeMonetarymodelssquaresntFund)Fund)IndicatorsIndi
Number of the second
IndicatorsIndicatorsIndicatorspublishedpublishedIndicatorspublishedby WorldIndicatorsBank andInternationInternationInternationInternationInternationalInternationInternationFund)Fund)InternationSylwester1970 -29 lessGrowthAverage NetPositive1989developerate ofInternation
kkpublishedkk<
kkk
Sylwester1970 -29 lessGrowthAverage NetPositiveOLS and(2005)1989in tern of the sectorSulvesterInternationSulvesterSulveste
InternationInternationInternationalalInternationMonetaryMonetaryInternationFund)Fund)InternationSylwester1970 -29 lessGrowthAverage NetPositive1989developerate ofInflows ofandSUR
kalalkkMonetaryMonetaryKKKFund)Fund)KKKSylwester1970 -29 lessGrowthAverage NetPositiveOLS and(2005)1989developerate ofinflows ofandSUR
Monetary Fund)Monetary Fund)Image: Monetary Fund)Sylwester1970 -29 lessGrowthAverage NetPositiveOLS and(2005)1989developerate ofinflows ofandSUR
Sylwester1970 -29 lessGrowthAverage NetPositiveOLS and(2005)1989developerate ofinflows ofandSUR
Sylwester1970 -29 lessGrowthAverage NetPositiveOLS and(2005)1989developerate ofinflows ofandSUR
(2005) 1989 develope rate of inflows of and SUR
d Income per FDI as a statisticall
countries capita percentage of y
(Barro and GDP significant
Lee (1994) (World effect
Bank)
Thangavelu,1988 -10 SouthReal GDPFDI inflowsPositiveOLS,
Yong and2007-Eastgrowth rate(UNCTAD)andFixed
Chongvilaivan Asian (Asian statistical effects
(2009) and East developme y and
Asian nt bank significant Random
countries database) effect effects
Vita and Kyaw1985 -126GrowthNet inflowsNegativeSystem
(2009) 2002 developi rate of real of FDI as a and GMM
ng per capita percentage of significant
countries GDP based GDP effect in
on (World case of
purchasing Bank's low
power World income
parity development countries;
(World indicators positive
Bank's (2004)) and
World significant

			developme		effect in	
			nt		case of	
			indicators		lower	
			(2004))		middle	
					and upper	
					middle	
					income	
					countries	
Wang and	84	1987 -	Log	Gross FDI	Greenfiel	Fixed
Wong (2010)	countrie	2001	difference	inflows as a	d has	effects,
	S		of per	share of host	positive	random
			capita real	country's	and	effects,
			GDP	GDP	significant	instrumen
			(World	(World	; Merger	tal
			Bank's	Bank's	and	variable
			World	World	acquisitio	technique
			Developm	Development	n has	S
			ent	Indicators)	negative	
			Indicators)		and	
					significant	
Wang and	69	1970 –	Per capita	FDI inflows	Negative	SUR
Wong (2011)	countrie	1989	real GDP	as a share of	and	
	S		growth	GDP	statisticall	
			(Borenszte	(Borensztein,	У	
			in, E., De	E., De	significant	
			Gregorio,	Gregorio, J.,	. Effect is	
			J., & Lee,	& Lee, J. W.	positive	
			J. W.	(1998))	and	
			(1998))		statisticall	
					У	
					significant	
					when	
					interacted	
					with	
					schooling.	

Baharumshah	1 country	1974 -	Real GDP	FDI inflows	Positive	OLS
and	(Malaysi	2004	per capita	as a ratio of	and	
Almansaied	a)		growth	GDP	statisticall	
(2009)			rate	(International	у	
			(Internatio	Financial	significant	
			nal	Statistics	effect	
			Financial	database for		
			Statistics	International		
			database	Monetary		
			for	Fund)		
			Internation			
			al			
			Monetary			
			Fund)			
Acharyya	1 country	1980 -	GDP	Total FDI in	Positive	
(2009)	(India)	2003	growth in	Million US \$	and	
			Millions	(World	statisticall	
			US \$	development	у	
			(World	indicator	significant	
			developme	(2007))		
			nt			
			indicator			
			(2007))			
Ahmed (2012)	1 country	1999 –	Quarterly	Real FDI	Positive	OLS
	(Malaysi	2008	real GDP	inflows	and	
	a)	Quarterl	(Departme	(Department	statisticall	
		У	nt of	of Statistics	у	
			Statistics	of Malaysia)	significant	
			of			
			Malaysia)			
Ang (2009)	1 country	1970 -	Per capita	FDI inflows	Positive	IV
	(Thailan	2004	real GDP		and	method
	d)	annual	(World		statisticall	
	,		Bank's		y	
			World			
	1					

			Developm		significant	
			ent		effect	
			Statistics)			
Chen, Chang	1 country	1968 -	GNP	Lagged FDI	Positive	Multiple
and Zhang	(China)	1990		(China	and	regression
(1995)				Statistical	significant	model
				Yearbook,		
				1991)		
Choong,	1 country	1970 -	Growth	FDI to GDP	Negative	Unrestrict
Yusop and	(Malaysi	2001	rate of real	ratio	and	ed error
Soo (2005)	a)		GDP	(World	statisticall	correction
			(World	Bank's	У	model
			Bank's	World	significant	
			World	Development	effect	
			Developm	Indicator		
			ent	2003 CD		
			Indicator	ROM)		
			2003 CD			
			ROM)			
Hoang,	1 country	1995 -	Growth	FDI to GDP	Positive	Panel
Wiboonchutik	(Vietnam	2006	rate of	ratio	and	least
ula and)		GDP	(Statistical	statisticall	squares
Tubtimtong			(Statistical	Yearbook of	У	
(2010)			Yearbook	Vietnam)	significant	
			of			
			Vietnam)			
Quader (2009)	1 country	1990 -	GDP	FDI as	Positive	OLS
	(Banglad	2006	(Statistics	percentage of	and	
	esh)		departmen	GDP – 2 year	significant	
			t of the	lagged		
			central	(Statistics		
			bank of	department of		
			Banglades	the central		
			h, World	bank of		

			UNCTAD	World Bank		
)	and		
				UNCTAD)		
Yu and	1 country	1991 -	Annual	(provincial	Positive	OLS
JingMei	(China)	2007	growth	FDI/Total	and	
(2009)			rate of	FDI) as a	statisticall	
			regional	ratio of	у	
			GDP of	(Provincial	significant	
			Chinese	GDP/Total		
			provinces	GDP)		
			(Annual	(Annual		
			China	China		
			Statistical	Statistical		
			Yearbook	Yearbook		
			from 1992	from 1992 –		
			- 2008)	2008)		

3.9 OVERVIEW OF FDI AND GROWTH META-REGRESSION ANALYSIS

Field	Search	Types of	Effect size	Number of	Countries	Aim of the
	engines	studies		studies		study
	used	included		(estimates)		
Inward	Google,	English	Partial	37 (633)	South and	Parameter
FDI and	Web of	language	correlation		East Asia	estimate and
economic	Knowledge	studies –			& Pacific	heterogeneity
growth		published			countries	
		and			as defined	
		unpublished			by world	
					bank +	
					South	
					Korea	
3.10 PRECISION EFFECT TEST (PET)

	All	Estimates	Country 1	Country	Country
	estimates	with		2	3
		endogeneity			
PET	0.05	0.05	-0.13	0.66	0.17
(UNWEIGHTED)	(2.05)	(1.48)	(-3.50)	(1.45)	(3.00)
	R ² =0.03	R ² =0.04	R ² =0.63	R ² =0.26	R ² =0.23
PET	0.15	0.35	-0.27	0.03	0.35
(WEIGHTED)	(9.44)	(15.96)	(-2.33)	(0.86)	(2.31)
	R ² =0.08	$R^2=0.22$	$R^2 = 0.32$	$R^2 = 0.54$	R ² =0.02
N	624	229	17	73	23

3.10.1 SIMPLE META REGRESSION – PRECISION EFFECT TEST (PET)

3.10.2 MULTIPLE META REGRESSION – PRECISION EFFECT TEST (PET)

	All estimates	Estimates	Country 1	Country 2	Country 3
		with	(East Asia)	(South East	(South
		endogeneity		Asia)	Asia)
PET	-0.07	0.29	-0.21	0.43	0.35
(WEIG	(-2.24)	(13.04)	(-3.97)	(5.62)	(2.31)
HTED)	Adj.R ² =0.35	Adj.R ² =0.79	Adj.R ² =0.88	Adj.R ² =0.67	$Adj.R^2=0.$
					02
PET	-0.07	0.29	-0.21	0.43	0.35
(CLUS	(-2.62)	(8.32)	(-204.9)	(6.77)	(1.01)
TERE	R ² =0.86	$R^2=0.79$	R ² =0.90	$R^2=0.69$	R ² =0.04
D)					
Ν	562	193	17	73	23

3.11 SIMPLE AND MULTIPLE REGRESSION RESULTS FOR EAST ASIA AND SOUTH ASIA

Statistic	Estimates for	Estimates for	Estimates for	Estimates for	
	East Asia	East Asia	South Asia	South Asia	
	(SIMPLE)	(MULTIPLE)	(SIMPLE)	(MULTIPLE)	
	(Col. 1)	(Col. 2)	(Col. 3)	(Col. 4)	
Un	-0.01	-0.07	0.24	0.77	
weighted, $\beta 0$	(-0.76)	(-4.53)	(8.33)	(6.33)	
	$R^2=0.62$	R ² =0.88	$R^2=0.27$	R ² =0.46	
Weighted by	-0.03	-0.07	0.24	0.77	
precision, β0	(-0.75)	(-658.85)	(4.94)	(100.96)	
	R ² =0.20	R ² =0.90	R ² =0.00	R ² =0.51	
Number of	17	17	23	23	
estimates					

3.12 MODERATOR VARIABLES – EAT ASIA AND SOUTH ASIA

Moderator variable	East Asia			South Asi	a	
	WLS	CLUSTER		WLS		CLUSTER
Single				2.99		2.99
				(4.54)		(78.18)
Skorea1	0.14	0.14				
	(8.68)	(349.42)				
No. of observations	17		17		23	23
Adjusted R2	0.88		0.90		0.46	0.51

4.1 SEARCH KEYWORDS USED IN GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Keywords for economic governance

Corporate governance OR Governance or economic governance or worldwide governance indicators OR Voice and Accountability OR Political Stability and Absence of Violence OR Government Effectiveness OR Regulatory Quality OR Rule of Law OR Control of Corruption OR Bureaucracy

Keywords for economic growth

Growth or economic growth or development or economic performance or investment or labour productivity or capital or innovation or labour market participation or progress or expansion or increase or improvement or advance or spill over effects or efficiency

Keywords for South and East Asia & Pacific countries

Emerging economies OR East Asian economies OR South east Asian economies OR East Asia OR South Asia OR South east Asia OR Afghanistan OR Bangladesh OR Bhutan OR India OR Maldives OR Nepal OR Pakistan OR Sri Lanka OR American Samoa OR Cambodia OR China OR Fiji OR Indonesia OR Kiribati OR Korea, Dem. Rep. OR Lao PDR OR Malaysia OR Marshall Islands OR Micronesia, Fed. Sts OR Mongolia OR Myanmar OR Palau OR Papua New Guinea OR Philippines OR Samoa OR Solomon Islands OR Thailand OR Timor-Leste OR Tuvalu OR Tonga OR Vanuatu OR Vietnam OR Asean OR Developing economies OR Developing countries OR South Korea OR Republic of Korea

4.2 PIOS FRAMEWORK USED IN MEASURES OF GOVERNANCE GROWTH STUDY

Population – The study should focus on South and East Asia Pacific economies or equivalent as specified in the search criteria.

Independent variable - The study should be examining the impact of measures economic governance in terms of a scale or its equivalent as specified in the search criteria.

Outcome variable - The study should be examining economic growth or as defined in the search criteria.

Study design - Study design can be either theoretical or empirical. A study is considered to be theoretical if it is based on some theoretical model drawing verbal or mathematical conclusions analysing impact of economic governance on economic growth. A study is considered to be empirical if it is based on regression model and draws an estimation model to estimate economic governance on economic growth.

4.3 STUDIES SATISFYING PIOS CRITERIA IN MEASURES OF GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Author (Year)	Population	Independent	Outcome	Study
		Variable	Variable	design
Adams and Mengistu (2008)	Y	Y	Y	Y
Agarwal and Samata (2006)	Т			
Ahmad (2005)	Т			
Allen, Qian and Qian (2004)	Y	Y	Ν	Y
Andrews (2010)	Т			
Anwar and Aman (2010)	Y	Y	Ν	Y
Anwar and Cooray (2012)	Y	Y	Y	Y
Avellaneda (2002)	Т			
Avellaneda (2006)	Т			
Azid, Khaliq and Jamil (2006)	Y	Ν	Y	Y
Bardhan (2002)	Т			
Bardhan (2009)	Т			
Bebbington, Dharmanwan,	Т			
Fahmi and Guggenheim (2006)				
Bekaert, Harvey and Lundblad	Т			
(2011)				
Bhatti (2001)	Т			
Boubakria, Cossetb and	Y	Y	Ν	Y
Guedhami (2004)				
Butkiewicz and Yanikkaya	Y	Y	Y	Y
(2004)				
Butkiewicz and Yanikkaya	Y	Y	Y	Y
(2011)				
Cali and Sen (2011)	Y	Ν	Y	Y
Campos and Nugent (1999)	Y	Y	Y	Y
Chatterjee (2008)	Т			
Cheng, Haggard and Kang	Т			
(1998)				

Chhatre and Saberwal (2006)	Т			
Clarke and Yuxing (1998)	Т			
Cubbin and Stern (2006)	Y	Y	N	Y
Currie (1996)	Т			
Dar and Amirkhalkhali (2011)	Y	Y	N	N
Dash and Raja (2009)	Y	N	Y	Y
Evans and Rauch (1999)	Y	Y	Y	Y
Evrensel (2010)	Y	Y	Y	Y
Fernandez, Gonzaler and Suarez	Y	Y	Y	Y
(2010)				
Gamber and Scott (2007)	Y	Y	N	Y
Gounder (2004)	Т			
Grindle (2004)	Т			
Groenewold and Tang (2007)	Y	Y	Y	N
Gu, Humphrey ad Messner	Y	Y	N	Y
(2007)				
Haggard ()	Т			
Haggard and Moon (1990)	Т			
Haggard and Tiede (2011)	Y	Y	Y	Y
Haque (2001)	Т			
Haque (2004)	Т			
Heller, Harilal and Chaudhuri	Т			
(2007)				
Hill (1997)	Т			
Hwang (1996)	Т			
Islam (2005)	Т			
Jallian, Kirkpatrick and Parker	Y	Y	Y	Y
(2007)				
Khadka (1993)	Т			
Knight and Schaik (2001)	Т			
Lam (1996)	Т			
Levy (2005)	Т			
Lio and Liu (2008)	Y	Y	N	Y
Liu (2005)	Т			

Mehta (2010)	Т			
Meleisea ()	Т			
Mendez-Picazo, Galindo-Martin	Ν	Y	Y	Y
and Ribeiro-Soriano (2012)				
Meon and Weill (2005)	Y	Y	Ν	Y
Meso, Datta and Mbarika (2005)	Y	Y	Ν	Y
Mohammed and Strobl (2011)	Y	Y	Ν	Y
Moran (1999)	Т			
Mueller (2006)	Т			
Nguyen and Dijk (2012)	Y	N	N	Y
Nwabuzor (2005)	Т			
Oliva and Rivera-Batiz (2002)	Y	Y	Y	Y
Papageorgiou and Turnbull	Т			
(1999)				
Park (1990)	Т			
Peritt and Clarke (1998)	Т			
Pinto (2004)	Т			
Presbitero (2006)	Ν	Y	Y	Y
Quibria (2006)	Т			
Riversa-Batiz (2001)	Т			
Saravanamuttu (2000)	Т			
Sharma (2007)	Т			
Subramanian (2007)	Т			
Sullivan, Rogers and Bettcher	Т			
(2007)				
Vora-Sittha (2012)	Т			
Wang and You (2012)	Y	Y	Y	Y
Wilkin (2011)	Т			
Zhang (2005)	Т			
Awasthi (1984)	Y	Y	N	Y
Boist & Child (1990)	Y	N	Ν	Y
April (2011)	Y	Y	Ν	Y
Dixit (2009)	Y	Y	Ν	Y
Goldsmith	Т			

Oliva and Rivera-Batiz (2002)	Y	Y	Y	Y
Goldsmith (1995)	Y	Y	Y	Y
Feeny (2005)	Y	Y	Y	Y
Feeny and Mcgillivray (2010)	Y	Y	Y	Y
Alonso (2010)	Y	Y	Y	Y
Busse and Groizard (2008)	Y	Y	Y	Y
Khamfula (2007)	Y	Y	Y	Y
Mo (2001)	Y	Y	Y	Y

Y= Study satisfies the criteria; N= Study does not satisfy the criteria

4.4 NUMBER OF EMPIRICAL STUDIES SATISFYING PIOS CRITERIS IN GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Criteria	Number of
	studies satisfying
	the criteria
Population (South and East Asia & Pacific countries)	32
Independent variable (Measures of governance)	29
Outcome variable (Economic growth)	18
Study design – Empirical	32
Decision Select if all 4 criteria match - PIOS	
Select for next stage	20
Deselect studies	71

4.5 COMPOSITION OF PUBLISHED (85%) AND UNPUBLISHED (15%) STUDIES USED IN GOVERNANCE AND GROWTH STUDY



4.6 DESCRIPTIVE STATISTICS OF MODERATOR VARIABLES INCLUDED IN GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Moderator	Definition	Mean	Standard
variable			deviation
Ptype1	=1 if the estimate is taken from an article that is	0.676	0.47
	published in a journal		
Ptype2	=1 if the estimate is taken from unpublished study	0.324	0.47
	(working paper or discussion paper)		
Obsgiven	=1 if the estimate is taken from a model in which	0.956	0.20
	observations are reported		
Obstaken	=1 if the estimate is taken from a model in which	0.044	0.20
	observations are not reported		
Gov1	=1 if the estimate belongs to model which defined	0.267	0.44
	governance in terms of voice and accountability		
Gov2	=1 if the estimate belongs to model which defined	0.025	0.16
	governance in terms of political stability		
Gov3	=1 if the estimate belongs to model which defined	0.065	0.25
	governance in terms of government effectiveness		

Gov4	=1 if the estimate belongs to model which defined	0.016	0.13
	governance in terms of regulation		
Gov5	=1 if the estimate belongs to model which defined	0.087	0.09
	governance in terms of law		
Gov6	=1 if the estimate belongs to model which defined	0.142	0.35
	governance in terms of corruption		
Govall	=1 if the estimate belongs to model which defined	0.065	0.25
	governance in terms of aggregate governance		
Data1	=1 if the estimate is taken from model that has used	0.62	0.49
	yearly data on growth		
Data2	=1 if the estimate is taken from model that has used	0.38	0.49
	non-yearly data on growth		
Dtype1	=1 if the estimate is taken from a model that has	0.815	0.39
	used panel data		
Dtype2	=1 if the estimate is taken from a model that has	0.018	0.13
	used time series data		
Dtype3	=1 if the estimate is taken from a model that has	0.167	0.37
	used cross sectional data		
Country1	=1 if the estimate belongs to a model that has used	0.018	0.13
	data on FDI and measure of governance of South		
	East Asia		
Country2	=1 if the estimate belongs to a model that has used	0.047	0.21
	data on FDI and measure of governance of South		
	Asia		
Country3	=1 if the estimate belongs to a model that has used	0.051	0.22
	data on FDI and measure of governance of East		
	Asia		
Country4	=1 if the estimate belongs to a model that has used	0.884	0.32
	data on FDI and measure of governance of mixed		
	countries		
Method1	=1 if the estimate belongs to a model that is	0.471	0.50
	estimated using OLS techniques		
Method2	=1 if the estimate belongs to a model that is	0.276	0.45
	estimated using panel data techniques		

Method3	=1 if the estimate belongs to a model that is	0 147	0.36
Wiethous	estimated using instrumental variable techniques	0.147	0.50
Mathod	-1 if the estimate belongs to a model that is	0.018	0.13
Wiethou4	-1 If the estimate belongs to a model that is	0.018	0.15
26.1.15	estimated using time series techniques	0.007	0.00
Method5	=1 if the estimate belongs to a model that is	0.087	0.28
	estimated using other techniques		
Dumchi1	=1 if the estimate belongs to a model that has	0.675	0.47
	included China in the list of sample countries		
Dumchi2	=1 if the estimate belongs to a model that has	0.325	0.47
	excluded China from the list of sample countries		
Dumsk1	=1 if the estimate belongs to a model that has	0.716	0.45
	included South Korea in the list of sample countries		
Dumsk2	=1 if the estimate belongs to a model that has	0.284	0.45
	excluded South Korea from the list of sample		
	countries		
Journal1	=1 if the estimate is taken from a journal that	0.727	0.45
	belongs to Economics and Finance discipline		
Journal2	=1 if the estimate is taken from a journal that	0.018	0.13
	belongs to Business Management discipline		
Journal3	=1 if the estimate is taken from a journal that	0.027	0.16
	belongs to Policy discipline		
Journal5	=1 if the estimate is taken from a journal that	0.227	0.42
	belongs to Development discipline		
Lauthor1	=1 if the first author of the study is from American	0.331	0.47
	University		
Lauthor2	=1 if the first author of the study is from European	0.42	0.49
	University		
Lauthor3	=1 if the first author of the study is from South &	0.049	0.22
	East Asian University		
Lauthor4	=1 if the first author of the study is from other	0.2	0.40
	Universities		
Omv1	=1 if the estimate is taken from a model that has	0.48	0.50
	included population variable		
Omv2	=1 if the estimate is taken from a model that has	0.169	0.38
	included domestic investment variable		

Omv3	=1 if the estimate is taken from a model that has	0.413	0.49
	included education variable		
Govsource1	= 1 if the estimate is estimated using governance	0.298	0.46
	data from Freedom House database		
Govsource2	= 1 if the estimate is estimated using governance	0.236	0.43
	data from International Country Risk Guide		
	database		
Govsource3	= 1 if the estimate is estimated using governance	0.075	0.26
	data from mixed database		
Govsource4	= 1 if the estimate is estimated using governance	0.102	0.30
	data from other sources		
Govsource5	= 1 if the estimate is estimated using governance	0.136	0.34
	data from Polity database		
Govsource6	= 1 if the estimate is estimated using governance	0.02	0.14
	data from Transparency International database		
Govsource7	= 1 if the estimate has used governance data from	0.132	0.34
	World Governance Indicators from World Bank		
	database		
Grosource1	= 1 if the estimate has used growth data from	0.098	0.35
	International Monetary Fund database		
Grosource2	= 1 if the estimate has used governance data from	0.144	0.35
	other databases		
Grosource3	= 1 if the estimate has used governance data from	0.262	0.44
	Penn World database		
Grosource4	= 1 if the estimate has used governance data from	0.496	0.50
	World Bank database		
Rank101	=1 if the estimate is taken from a journal that is	0.044	0.21
	ranked as 1* in ABS 2010 ranking		
Rank102	=1 if the estimate is taken from a journal that is	0.296	0.46
	ranked as 2* in ABS 2010 ranking		
Rank103	=1 if the estimate is taken from a journal that is	0.610	0.49
	ranked as 3* in ABS 2010 ranking		
Rank104	=1 if the estimate is taken from a journal that is	0.051	0.22
	ranked as 4* in ABS 2010 ranking		

4.7 SUMMARIES OF EMPIRICAL STUDIES INCLUDED IN GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Study and	Time	Countries	Dependent	Independent	Findings	Techniques
year	period		variable	variable		
Adams	1991 -	82	Real GDP	Governance	Positive	Least
and	2002	developin	growth rate	(Kaufmann	and	squares
Mengistu		g	and Real	et al. (2005))	significant	dummy
(2008)		countries	GDP per		effect of	variable
			capita		governance	approach
					on growth	
			(World			
			Economic			
			Outlook			
			(2004) and			
			Global			
			Developme			
			nt Network			
			Growth			
			Database			
Anwar and	1970 -	8 South	Per capita	Democracy	Positive	Ordinary
Cooray	2009	Asian	income		and	least
(2012)		countries	(constant	Freedom	significant	squares
			2000 US\$)	House	even when	method,
			(World	Political	interacted	Fixed
			Developme	Rights Index	with money	effects,
			nt	and freedom	supply	System
			Indicators,	House Civil		GMM
			2011)	Liberties		
				Index		
				(Freedom		
				House,		
				2011)		

				Polity IV		
				Index		
				(Marshall		
				and Jaggers,		
				2010)		
Butkiewic	1970 -	29	Real GDP	Democracy	Rule of law	Seemingly
z and	1999	develope	growth rate	(Freedom	- positive	unrelated
Yanikkaya		d and 85	(World	House and	and	regression
(2004)		developin	Developme	Polity III)	significant	technique
		g	nt		effect.	(SUR)
		countries	Indicators,	Rule of law		and/or three
			1999)	(Easterly,	Corruption	stage least
				1999)	– negative	squares
			Initial GDP		and	(3SLS)
			per capita		insignifican	
			(Penn-		t.	
			World			
			Table)		Bureaucrac	
					y – positive	
					and	
					insignifican	
					t.	
					Democracy	
					- positive	
					and	
					insignifican	
					t.	
Butkiewic	Two	Over 100	Growth of	Rule of law	Positive	Seemingly
z and	sample	develope	real GDP	(Kaufmann	and	unrelated
Yanikkaya	period:	d and	per capita	et al., 2007)	significant	regression
(2011)	1970 –	developin	(World		in case of	(SUR)
	1999	g nations	Bank,		developing	technique
	1990 -		2007)		countries	
	2004					

					Developed	
					countries –	
					positive but	
					not	
					significant	
Campos	108	1982 -	Average	Democracy	Without	Ordinary
and	countries;	1995	level of real	(Freedom	interaction	least
Nugent	28 East		per capita	House)	terms	squares
(1999)	Asian		GDP		Democracy	method
	countries			Bureaucratic	– positive	
				quality	and	
				(ICRG)	significant	
					for all	
				Rule of Law	sample and	
				(ICRG)	East Asian;	
					Bureaucrac	
					y – positive	
					and	
					significant	
					for all	
					sample,	
					positive	
					and	
					insignifican	
					t – East	
					Asian;	
					Rule of	
					Law –	
					Positive	
					and	
					significant	
					for all	
					sample,	
1		1	1		1	

					positive	
					and	
					insignifican	
					t for East	
					Asian	
					With	
					interaction	
					terms	
					Only	
					democracy	
					is positive	
					and	
					significant	
Evrensel	31	1990 -	Average	Corruption	Negative	Ordinary
(2010)	developed	2000	growth rate	(ICRG)	and	least
	and 90		of real GDP		significant	squares
	developin		(Internation			method
	g		al Financial			
	countries		Statistics,			
			IMF, 2007)			
Fernandez,	84	1980 -	Growth rate	Rule of law	Positive	Ordinary
Gonzalez	countries	2004	of real per	(Heritage	and	least
and Suarez			capita GDP	Foundation	significant	squares
(2010)			(World	(Freedom))		method and
			Bank)			random
						effects
Haggard	74	2003 -	GDP per	Corruption	Both	Two stage
and Tiede	developin	2007	capita in	(Transparen	positive	least
(2011)	g and		1995	су	and	squares
	transition			International	significant	
	countries			CPI)		
				Rule of law		

				(World		
				Bank and		
				others)		
Jalilian,	117	1980 -	GDP	Regulatory	Positive	Ordinary
Kirkpatric	countries	2000	growth per	quality	and	least
k and	for cross		capita		significant	squares
Parker	section			Government	effect	method,
(2007)	regression		(World	effectiveness		fixed
	96		Bank)			effects and
	countries			(Kaufmann		random
	for panel			et al., 2005)		effects
	regression					
Oliva and	119	1970 -	Real per	Democracy	Democracy	Ordinary
Rivera-	developin	1994	capita	(Polity IV)	- positive	least
Batiz	g		annual		and	squares
(2002)	countries		growth rate	Rule of law	significant	method and
				(Kaufmann		Three stage
				et al., 1999)	Rule of law	least
					– positive	squares
					and	
					insignifican	
					t	
Goldsmith	59 less	1980 -	Average	Democracy	Both –	Ordinary
(1995)	developed	1990	annual	(Freedom	negative	least
	and		growth rate	House)	and	squares
	transitiona		of GDP		significant	method
	l countries		(IMF,	Property		
			1994)	rights index		
				(Johnson		
				and Sheehy		
				(1995))		
Feeny	1 country	1965 -	GDP	Governance	Mixed	Auto
(2005)		1999	growth	(ICRG)	effect but	Regressive
			(World		insignifican	Distributed
			Bank)		t	

						Lag
						(ARDL)
Feeny and	29 Small	1980 -	GDP per	Governance	Positive	Fixed
Mcgillivra	Island	2004	capita	(World	and	Effects and
y (2010)	Developin		growth	Bank)	insignifican	GMM
	g States		measured in		t	
			constant			
			local			
			currency			
			units			
			expressed			
			as a			
			percentage			
			(World			
			Bank			
			(2006),			
			Asian			
			Developme			
			nt Bank			
			(2006),			
			Grimes			
			(2000))			
Alonso	154	2006 -	Per capita	Governance	Positive	Two stage
(2010)	countries	2007	Income	index	and	least
			(Maddison)	(World	significant	squares
				Governance		with
				indicators)		instrumenta
						l variable
						technique
Busse and	84	1994 -	Real growth	Rule of Law	Positive	GMM
Groizard	countries	2003	of GDP per	(PRS	and	
(2008)			capita in	Group)	significant	
			per cent			

			(World			
			Bank,			
			2006)			
Khamfula	17	1994 -	Real GDP	Corruption	Positive	Ordinary
(2007)	countries	2004	(World	(Corruption	and	least
`			Bank)	perception	significant	squares
			,	Index from	0	method
				Centre for		
				Corruption		
				Research)		
Mo (2001)	40	1070	Growth rate	Corruption	Domocrocy	Ordinary
WIO (2001)	49	1970 -	of most CDD	(Transnorm	Democracy	
	countries	1985		(Transparen	- Positive	least
			111	cy	and	squares
			percentage	International	insignifican	method,
			(Barro and)	t	Two stage
			Lee)			least
				Democracy	Instability	squares
				(Freedom	- Negative	
				House)	and	
					significant	
				Political	when	
				stability	transmissio	
				(PINSTAB)	n channels	
					are not	
					included	
Mauro	67	1980 -	Per capita	Corruption	Positive	Ordinary
(1995)	countries	1983	GDP	(Business	and	least
			growth	International	significant	squares
)		method,
						Two stage
						least
						squares
						squares

Drury,	More than	1982 -	Growth of	Corruption	Corruption	Ordinary
Krieckhau	100	1997	GDP	(ICRG)	– negative	least
s and	countries		(World		and	squares
Lusztig			Bank)	Democracy	significant	method
(2006)				(Freedom	in non-	
				House and	democratic	
				Polity IV)	countries;	
					positive	
					and	
					insignifican	
					t in	
					democratic;	
					Democracy	
					(Freedom	
					house and	
					polity IV –	
					negative	
					and	
					insignifican	
					t; positive	
					and	
					insignifican	
					t (Alvarez,	
					Cheibub,	
					Limongi	
					and	
					Przeworski	
					(ACLP)	
					democracy	
					data))	
Assiotis	119	1984 -	Real GDP	Democracy	Corruption	System
and	countries	2007	per capita	(Freedom	and	GMM and
Sylwester			(Penn	House and	democracy	Fixed
(2012)			World	Polity Iv)	- positive	

			Table,		and	effects
			version 6.3)	Corruption	significant	model
			,	(ICRG and	C	
				Transparenc	Governanc	
				v	e –	
				International	negative	
)	and	
				Governance	significant	
				(World	8	
				Governance		
				Indicator)		
Ekanavake	85	1980 -	Growth of	Democracy	Negative	Ordinary
and	developin	2007	real GDP	(Freedom	and	least
Chatrna	α veropin	2007	per capita	House)	insignifican	squares
(2010)	5 countries		in constant	110030)	t	method
(2010)	countries		(2000) US		(significanc	memou
			(2000) US			
			uonars (World		e valles	
					with time)	
	0.4	100.0	Bank)		2	
Gani	84	1996 -	Real growth	Governance	Democracy	Ordinary
(2011)	countries	2005	of gross	indicators	– negative	least
			domestic	(World	and	squares
			product	Governance	significant	method
			(annual	indicators)		
			percentage)		Political	
			(World		stability –	
			Bank,		positive	
			2007)		and	
					significant	
					Governmen	
					t	
					effectivene	
					ss –	
					positive	

					and	
					significant	
					Regulatory	
					quality –	
					negative	
					and	
					insignifican	
					t	
					Rule of law	
					– negative	
					and	
					insignifican	
					t	
					Control of	
					corruption	
					– negative	
					and	
					significant	
Seldadyo,	82	1984 -	Average	Governance	Positive	Parsimonio
Nugroho	countries	2004	GDP per	(ICRG)	and	us
and Haan			capita		significant	regression
(2007)			growth			C
× ,			rates			
			(World			
			Bank)			
Command	159	1960 -	Per capita	Democracy	Positive	GMM
er and	countries	2009	GDP	(Freedom	and	
Nikoloski			growth	House and	insignifican	
(2010)			(World	Polity IV)	t	
			Penn Table)			

			Real GDP			
			growth			
			(World			
			Developme			
			nt			
			Indicators)			
Law and	8 East	1980 -	Real GDP	Institutional	Institutiona	FMOLS
Habibullah	Asian	2001	per capita	quality	l quality,	
(2006)	countries		(World	(ICRG)	Rule f aw,	
			Bank)		Bureaucrac	
					y and	
					corruption -	
					Positive	
					and	
					significant	

4.8 OVERVIEW OF MEASURES OF GOVERNANCE AND GROWTH META-REGRESSION ANALYSIS

Field	Search	Types of	Effect size	Number of	Countries	Aims of
	engines	data		studies	studied	study
	used	included		(Estimates)		
Economic	Various	English	Partial	29(554*)	South	Parameter
governance		language	correlation		and East	estimate and
and					Asia &	heterogeneity
economic		Published			Pacific	
growth		and			countries	
		unpublished			as	
					defined	
					by world	
					bank +	
					South	
					Korea	

*Total number of estimates (combining all measures of governance)

4.9 PRECISION EFFECT TEST (PET)

4.9.1 SIMPLE META REGRESSION RESULTS – PRECISION EFFECT TEST (PET)

	VOICE	POLITI	GOVER	REGU	LAW	COR	AGGR
	AND	CAL	NMENT	LATIO		RUPT	EGAT
	ACCOU	STABI	EFFECT	Ν		ION	E
	NTABIL	LITY	IVE)NE				GOVE
	ITY		SS				RNNA
							CE
PET	0	0.10	0	0.14	0	-0.07	0.42
(UNW	(-0.06)	(1.38)	(0.69)	(3.52)	(0.68)	(-	(4.13)
EIGHT						3.65)	
ED)	$R^2 = 0.00$	$R^2=0.1$	$R^2 = 0.01$	$R^2 = 0.6$	$R^2=0.$	$R^2=0.$	$R^2 = 0.3$
		4		4	01	15	3
PET	0	-0.10	0	0.15	-0.01	-0.17	0.51
(WEIG	(4.57)	(-4.64)	(-1.54)	(2.98)	(-	(-	(2.53)
HTED)	$R^2=0.13$	$R^2 = 0.6$	$R^2 = 0.06$	$R^2 = 0.5$	14.73)	2.49)	
		4		6	$R^2=0.$	$R^2=0.$	$R^2 = 0.1$
					83	08	6
Ν	147	14	36	9	48	78	36

4.9.2 MULTIPLE META REGRESSION RESULTS – PRECISION EFFECT TEST (PET)

	VOICE	POLITI	GOVE	REGU	LAW	COR	AGGR
	AND	CAL	RNME	LATIO		RUPT	EGAT
	ACCO	STABI	NT	Ν		ION	Е
	UNTA	LITY	EFFEC				GOVE
	BILIT		TIVEN				RNNA
	Y		ESS				CE
PET	0	0.70	0	0.15	0.04	0.01	0.27
(WEI	(0.98)	(7.78)	(-3.29)	(2.98)	(19.1)	(2.34)	(4.94)
GHT	Adj.R ²	Adj.R ²	Adj.R ²	Adj.R ²	Adj.R ²	Adj.R	Adj.R ²
ED)	=0.85	=0.95	=0.81	=0.50	=0.99	² =0.9	=0.96
						9	
PET	0	0.70	0	0.15	0.04	0.01	0.27
(CLU	(0.89)	(8.22)	(-	(9.59)	(175)	(4.51)	(1.69)
STER	$R^2 = 0.8$	R ² =0.9	104.64)	R ² =0.5	R ² =0.9	$R^2=0.$	R ² =0.9
ED)	6	6	R ² =0.8	6	9	99	7
			1				
Ν	147	14	36	9	48	78	36

4.10 REGRESSION RESULTS FOR POLITICAL STABILITY, GOVERNMENT EFFECTIVENESS AND REGULATION

	Political Stability	Government	Regulation
		effectiveness	
Un weighted, β0	0.16	-0.04	0.12
(Row1)	(2.41)	(-0.33)	(1.02)
	R ² =0.68	R ² =0.00	$R^2=0.50$
Weighted by	0.11	-1.21	0.05
precision, β0	(3.78)	(-9.28)	(0.41)
(Row2)	R ² =0.64	R ² =0.78	R ² =0.70
Number of	14	21	7
estimates			

4.10.1 SIMPLE META REGRESSION RESULTS

4.10.2 MULTIPLE META REGRESSION RESULTS

	Political Stability	Government	Regulation
		effectiveness	
Weighted by	-0.66	-0.50	0.55
precision, β0	(-1.48)	(-5.54)	(7.13)
(Row1)	Adj.R ² =0.95	Adj.R ² =0.99	Adj.R ² =0.97
Cluster analysis,	-0.66	-0.50	0.55
β0	(-2.48)	(-2.07)	(9.29)
(Row2)	R ² =0.96	R ² =0.99	R ² =0.98
Number of	14	20	7
estimates			

4.10.3 MODERATOR VARIABLES – POLITICAL STABILITY, GOVERNMENT EFFECTIVENESS AND REGULATION

Moderator	Political Stability		Government		Regu	lation
variable			Effectiveness			
Data1	0.58	0.58	-0.59	-0.59		
	(2.61)	(4.43)	(-6.62)	(-2.44)		
Dtype1			0.73	0.73		
			(14.46)	(7.84)		

Dsource1	0.71	0.71				
	(1.73)	(2.91)				
Journal1					-0.50	-0.50
					(-7.27)	(-10.69)
Govsource4	-0.13	-0.13				
	(-4.51)	(-10.07)				
No. of	14	14	20	20	7	7
observations						
Adj R^2/R^2	0.95	0.96	0.98	0.98	0.97	0.98

4.11 FUNNEL PLOTS AND CHRONOLOGICAL ORDER OF MEASURES OF GOVERNANCE AND GROWTH

4.11.1 FUNNEL PLOTS









4.11.2 CHRONOLOGICAL ORDER OF ESTIMATES



